Doctoral theses at NTNU, 2023:350

Hans E. Comtet

Drones in Healthcare Systems: Insights from a Multi-Level Perspective

NTNU

NTNU Norwegian University of Science and Technology Thesis for the Degree of Philosophiae Doctor Faculty of Architecture and Design Department of Design



Norwegian University of Science and Technology

Hans E. Comtet

Drones in Healthcare Systems: Insights from a Multi-Level Perspective



Thesis for the Degree of Philosophiae Doctor

Trondheim, November 2023

Norwegian University of Science and Technology Faculty of Architecture and Design Department of Design



Norwegian University of Science and Technology

NTNU

Norwegian University of Science and Technology

Thesis for the Degree of Philosophiae Doctor

Faculty of Architecture and Design Department of Design

© Hans E. Comtet

ISBN 978-82-326-7400-8 (printed ver.) ISBN 978-82-326-7399-5 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

Doctoral theses at NTNU, 2023:350

Printed by NTNU Grafisk senter

Drones in Healthcare Systems: Insights from a Multi-Level Perspective

Hans E. Comtet

CC-BY 2023/06/21

Preface

This article-based thesis is submitted to the Norwegian University of Science and Technology (NTNU) for partial fulfillment of the requirements for the degree of philosophiae doctor (PhD). The doctoral work was performed at Oslo University Hospital, at the Intervention Center.

The main supervisor from NTNU was Martina Keitsch and the co-supervisors were Ole Andreas Alsos and Mari Bjerck. Karl-Arne Johannessen was the main supervisor of Oslo University Hospital, with Erik Fosse as co-supervisor. Miriam Begnum, who had previously worked and completed her PhD at NTNU, acted as an additional supervisor.

This thesis is part of a project to "Research an implementation of Unmanned Aircraft Systems (UAS) for the fast and secure transport of biological material and blood products" in Oslo University Hospital. The Norwegian Research Council funded the work under the HELSEVEL program (Grant No. 282207/2018) [1].

Abstract

The process of integrating technological developments into existing practices is not straightforward. This thesis examines the potential usefulness of the concept of system innovation as a means of enhancing the ability of healthcare systems to integrate drone solutions into laboratory logistics by using Oslo University Hospital, Norway, as the primary study object, using its transport to different locations as the study topic.

The primary research question asks if and eventually how drones can become part of a future healthcare transport system. From a broader perspective, the thesis also investigates some principal factors of possible significance in the successful integration of new technologies into mature processes that are "locked in" with the current operations of an organization. The underlying basis of the analysis is that healthcare institutions are socio-technical systems with interrelated human, social, and organizational characteristics. The analysis in this thesis of this sociotechnical system uses the multi-level perspective, whereby the overall system is divided into distinct levels representing individual clinics, hospitals, and the overall healthcare institution. In the context of this framework and during the course of developing this thesis, three separate research papers related to the integration of drones into healthcare systems have been published in peer-reviewed journals. Based on the motivations, methodologies, and results of these three studies, the thesis describes their basis in the multi-level perspective, using the outcome of the research papers to identify factors that may possibly affect the successful implementation of new technologies in existing organizations. In addition, the thesis discusses how healthcare institutions may be considered socio-technical systems by eliciting the views of healthcare personnel towards new technologies and identifying the changes that technology, specifically drones, may contribute toward a sustainable healthcare system. In this way, the results presented in this thesis, in part, move beyond the original research question to more widely discuss how large-scale organizations can successfully realize the benefits of technological change.

The usefulness of the multi-level perspective and the concept of socio-technical systems in identifying and obtaining these benefits is described, and the implications for various stakeholders and policymakers are outlined and discussed. The overlying contribution of this thesis may be the usefulness of applying the multi-level perspective to an innovation project and the projection of a broader

vi Hans E. Comtet: Drones in Healthcare Systems: Insights from a Multi-Level Perspective

analytical framework beyond a protected space for experimentation and learning processes of drone integration. Although drones may have a limited impact on current organizational processes in the healthcare sphere, the thesis concludes that drones are perceived positively by healthcare workers, suggesting that future innovations could indeed enhance the current procedures used in such socio-technical systems.

Dedication

To Maia

Acknowledgments

In writing this section, I reflect on the wonderful PhD period that I enjoyed. I am grateful for the opportunity and support I have received during this journey.

My supervisor from Oslo University Hospital, Karl-Arne Johannessen, has been excellent in challenging me throughout the past three years. His incredible knowledge and life experience were a true inspiration. I learned a lot from him on multiple subjects and I am grateful to have worked with him while studying this PhD.

My supervisor from NTNU, Martina Keitsch, was a fantastic resource and supervisor. I always thought she knew where I was going, which gave me great security in the many phases of the research project, where I felt somewhat lost. Thank you, Martina, for your valuable advice, guidance and all the great discussions we had!

I also want to thank my co-supervisors, Ole Andreas Alsos, Erik Fosse, and Miriam Begnum. I appreciate their support during this project on various topics and in different phases. Sometimes, the only guide you need is a conversation, someone who clarifies your thoughts or shows you alternative pathways. Thank you for all that and for motivating and supporting me along the way!

I would also like to thank all Ph.D. students I met at *Bygg 20* (office building at the old Gaustad Hospital), at the Intervention Center, and NTNU. Knowing that others are in similar situations and that you can reach out is always helpful. And, of course, the cuisine at *Bygg 20* was amazing! Also, thank you to all who have read and commented on my work in different settings or have somehow participated in this work and contributed to the research results during these three years. Thank you, Ashis Jalote Parmar, for your valuable comments in the end!

One consequence of the Covid-19 restrictions was that my PhD was more isolated than otherwise. However, my neighbors and I had regular Friday afternoon aperitifs. A special thanks go to Lilleba, my 85-year-old neighbor and one of the regulars at the meet-ups, who has always read my published articles first. Thank you, Snippen, for your support!

Most importantly, on the home front, I want to thank my dear wife, Maia, for encouraging, motivating, and supporting me during this journey. I truly feel humbled and grateful for all the arrangements you made so that this PhD was possible. Therefore, I want to dedicate this thesis to you. I also want to mention our daughters, Lara and Liv. Thank you for your support, love, visits to the office and the whiteboard drawings! Who knows, maybe drones will be part of your

future lives.

Finally, a big thank you to my family and friends for believing in me and supporting me and for the love you gave me.

Oslo, June 21st, 2023. Hans Comtet

Contents

Pr	eface		iii		
Ab	Abstract				
De	Dedication vii				
Ac	Acknowledgments ix				
Co	ntent	t s	xi		
Fig	gures		XV		
Ta	bles				
Ac	ronyı	ms	xix		
Gl	ossar	у			
1	Intro	oductio	\mathfrak{m}		
	1.1	What	is a drone?		
		1.1.1	Technological adaption in healthcare 2		
	1.2	Resear	cch perspective 4		
		1.2.1	Socio-technical systems		
		1.2.2	The multi-level perspective 6		
	1.3	Resear	rch motivation 8		
	1.4	The ro	le of design		
	1.5	Aims a	and research questions 10		
	1.6	Resear	rch papers		
		1.6.1	Contributions of the research papers		
	1.7	Organ	ization of this thesis		
2	Rese	earch p	osition		
	2.1	Episte	mological awareness 15		
	2.2	Reflex	ivity		
		2.2.1	Insider limitations and challenges 19		
		2.2.2	Political considerations		
		2.2.3	Ethical considerations		
		2.2.4	Research journey		
	2.3	Analyt	ic strategy and plan		
	2.4	2.4 Research description			
		2.4.1	Dialogic validity 24		
		2.4.2	Catalytic validity		
		2.4.3	Democratic validity 25		
		2.4.4	Process validity		

		2.4.5	Outcome validity	25	
		2.4.6	Construct validity	25	
3	Data	a and n	nethodology	27	
	3.1	Resear	rch design	27	
	3.2	Metho	ds of data collection	27	
		3.2.1	Mixed-methods	27	
		3.2.2	Case study: Oslo University Hospital Trust, Oslo, Norway	29	
	3.3	Metho	ds to determine whether current healthcare systems may be		
		under	stood as socio-technical systems	31	
		3.3.1	Observation and shadowing	31	
		3.3.2	Scoping review	31	
	3.4	Metho	ds to assess the attitudes healthcare workers have toward		
		drones	S	32	
		3.4.1	Questionnaire	32	
	3.5	Metho	ds to assess changes that drones may accelerate in health-		
		care s	ystems, and how drones may become a part of the future		
		health	care transport system	32	
		3.5.1	Focus group interview	32	
4	Resu	ılts		35	
	4.1	Paper	1	35	
		4.1.1	Main findings and conclusion	36	
	4.2	Paper	2	36	
		4.2.1	Main findings and conclusion	37	
	4.3	Paper	3	37	
		4.3.1	Main findings and conclusion	38	
5	Disc	ussion		39	
	5.1	The cı	urrent healthcare system as a socio-technical system	39	
	5.2	Attituo	des of healthcare workers toward drones	41	
	5.3	Accele	erated changes of drones in healthcare systems	42	
	5.4	Naviga	ating transitions	44	
	5.5	Impro	ving work processes	46	
	5.6	Contextual understanding			
	5.7	Resear	rch validity	51	
		5.7.1	Dialogic validity	51	
		5.7.2	Catalytic validity	51	
		5.7.3	Democratic validity	51	
		5.7.4	Process validity	52	
		5.7.5	Outcome validity	52	
	_	5.7.6	Construct validity	53	
6	Contributions and implications				
	6.1	Contri	butions	55	
		6.1.1	Socio-technical reflexivity and sustainable change	56	
		6.1.2	The multi-level perspective as a tool and mindset	57	
	6.2	Lessor	is learned and limitations	58	

		6.2.1	Implications for researchers	59
		6.2.2	Implications for leaders	59
		6.2.3	Implications for facilitators of change	60
		6.2.4	Implications for designers and niche entrepreneurs	60
7 (Conc	lusion	s	63
5	7.1	State-c	of-the-art and future outlook	63
Bibl	liogr	aphy .		65
Рар	er 1			87
Рар	er 2			111
Рар	er 3			129
A	State	ements	of co-authorship	147

Figures

1.1	Three interrelated analytical dimensions (adapted from Geels [51])	5
1.2	Multi-level perspective of the system innovation model (adapted	
	from Geels [51, 53, 55])	6
2.1	Deductive, inductive, and abductive reasoning	16
2.2	Relationship between the research questions and the multi-level	
	perspective	21
2.3	Plan–act–observe–reflect cycle	24
3.1	Key elements that result from system science (adapted from Ison	
	[36])	28
3.2	Oslo University Hospital Case	30
5.1	Typology of designer (adapted from Manzini [164])	45
5.2	Typology of middle managers (adapted from Rydland [162])	45
5.3	Continuous improvement corresponding with system innovation	49
5.4	Organizational learning versus learning space	50

Tables

1.1	Relationships between research papers and research questions	12
2.1	Comparison of hard and soft traditions of system science (Ison [36], adapted from Checkland [95])	17
2.2	Ontological and epistemological choices (adapted from Auernhammer [99])	17
2.3	Validity criteria and goals of research (inspired by Herr [113] with #6 from Yin [114])	22
2.4	Overview of the three articles included in the thesis	23
4.1	General overview of the results	35

Acronyms

- AI Artificial Intelligence. 2
- BVLOS Beyond Visual Line of Sight. 2
- EASA European Aviation Safety Agency. 2
- EVLOS Extended Visual Line of Sight. 2
- EXCEL Spreadsheet Software (Microsoft). 29
- MDPI Multidisciplinary Digital Publishing Institute (Basel, Switzerland). 24, 51
- ML Machine Learining. 2
- NTNU Norwegian University of Science and Technology. iii, ix, 19
- PDSA Plan-Do-Study-Act. 48
- PhD philosophiae doctor. iii, ix, 20, 58, 64
- PubMed Database of biomedical literature. 51
- SPSS Statistical Software Suite (IBM). 29
- STRN Sustainable Transition Research Network. 64
- UAS Unmanned Aircraft Systems. iii
- UAV Unmanned Aerial Vehicle. 1, 2
- VLOS Visual Line of Sight. 2
- VTOL vertical take off and land. 2

Glossary

- **facilitator** Someone who helps to structure and process interactions, so that groups are able to function effectively and make high-quality decisions. 55, 60
- HELSEVEL Research program for health service research. iii, 19
- **multi-level perspective** The multi-level perspective (MLP). v, xv, 3, 6–9, 11, 12, 15, 18–21, 28, 36, 39, 41, 46, 48, 50, 53, 55–59, 63, 64
- **Oslo University Hospital** Consists of four main institutions located inside the area of Oslo, and several minor facilities: Rikshospitalet (The National Hospital, offering local, regional, and national services), Ullevål University Hospital (local, regional, and national services), Radiumhospitalet (a specialized cancer hospital), and Aker University Hospital (local and central hospital).. iii, v, ix, xv, 9, 19, 20, 29, 30, 36, 37, 40, 42, 48, 55, 56, 58
- **Sustainable development** Sustainable development describes social, economic, and environmental aspects of sustainability. The aspects are interconnected, however often understood and weighted differently. 60
- **transition** The contemporary discussion about transitions discusses how change occurs and how change can be managed and directed towards a sustainable future in complex systems such as healthcare. In this thesis similar to system innovation. 9
- wicked problem A wicked problem is ill-defined. In contrast to tame problems, where everybody agrees on what the problem is. 16, 50, 53, 59

Chapter 1

Introduction

Technological innovations promise an opportunity to create sustainable services in many sectors, including healthcare, which is the system of interest in this thesis. In the healthcare sector, technological developments and new research-based discoveries have opened new opportunities for patient diagnostics and treatment, as well as in the fields of communication and logistics. Changes in healthcare methodology have the potential for better and more efficient treatment, but may also harm patients and increase costs if not controlled. Healthcare systems are generally changing as patients become increasingly well-educated, with greater access to online medical sources and help, and solutions such as telemedicine are continuously improving.

An interesting upcoming solution is Unmanned Aerial Vehicle (UAV), commonly termed "drones." Since 2015, numerous studies have been performed around the world on drone transportation in medical logistics. Drones have been suggested to improve transport services of biological and other materials between doctors, laboratories, hospitals, and even transport home to patients. The estimate of transporting medical supplies and premium deliveries in 2050 is a fleet of 100,000 drones throughout Europe [2]. SARS-CoV-2 pushed the development towards solutions that do not require patients to leave their homes in multiple parts of health care, and telemedicine has become a common term in the last few years. For example, the use of digital platforms to treat patients has increased since 2020 in most developed health systems, and drones appeared to provide useful transport solutions when distancing and lock-downs during the pandemic were extensive [3, 4].

In the following sections, drones and technological adaptation in healthcare are introduced. Next, the research perspective, motivation, aims, and research questions addressed in this thesis are presented. Then the potential contributions to the field are elucidated. Finally, the organization of the remaining of this thesis is outlined.

1.1 What is a drone?

An Unmanned Aerial Vehicle (UAV) is commonly referred to as a drone, reflecting that they make a similar sound to male bees once airborne [5]. Drones describe autonomous or remotely controlled/managed aerial vehicles that fly and can carry cargo [6, 7].

Drones can range from large military-sized vehicles to commercially available micro-drones. Their flight distance, altitude, and payload capacity vary considerably [8]. Drones can be classified as fixed-wing, rotor-driven, or hybrid, combining these solutions, enabling vertical take-off and landing [9–11].

Currently, fixed-wing drones have longer flight ranges and greater payloads but require a runway or launch platform [12]. Rotary-wing drones can take off and land vertically (VTOL). This makes them easier to navigate in urban environments, but they currently offer a limited flying range, lower speed, and smaller payload capacity [13]. This may change considerably during the next 2-3 years.

The flexible design of drones means they are now used in multiple ways, including for transportation. Drones may reduce travel time compared to traditional road transport [14, 15] and can often access areas that are difficult to reach by ground transport [13].

Drone technology combines multiple technologies and areas of expertise that have been transferred from a wide variety of industrial sectors, that is, software and hardware engineering, Artificial Intelligence (AI), and Machine Learning (ML) [16]. Furthermore, drones can be integrated with mobile technology [16, 17].

The European Aviation Safety Agency (EASA) is continuously developing regulations for drone flights in Europe [18]. Regulation categories include visual line-of-sight (VLOS), beyond visual line-of-sight (BVLOS), and extended visual line-of-sight (EVLOS), which refer to visual observers keeping track of the drone [19]. The full potential of autonomous drones for transport will be realized when drones are integrated into airspace and do not require additional observers [20].

1.1.1 Technological adaption in healthcare

Humans of all age groups increasingly become drivers of patient action and autonomy [21, 22]. Based on the observed acceptance of new technologies in most populations and age groups, expectations related to technological solutions should be anticipated to surge in the future with regard to redesign and more user-friendly services than we traditionally have organized. Due to the increased use of technical solutions, the public may demand data integration and collaboration among experts. Different actors will have to work together seamlessly to achieve sustainable healthcare systems, with the aim of generating services, goods, designs, and social values that can eventually define the system itself [23].

It is well accepted that healthcare systems are complex [24–26] and that the use of automated and interconnected technologies has the potential to make dif-

ferent topics and services more integrated and therefore more robust and effective. However, the diffusion of new technologies in health services competes with mandatory regulations, quality demands, and organizational structures. Klitkou et al. [27] described the positive feedback of an established technology or structure as "lock-in." A potential result of multiple lock-in processes may be rigid institutions, and such systems may prefer incremental changes rather than radical innovations.

An aging population, a more comprehensive range of lifestyles, and a general increase in healthcare costs force most developed healthcare systems worldwide to work more efficiently. The combination of technical, social, and system components represents an immense but complex, and in part diffuse, solution space [28] in which sustainable concepts and new services can be developed.

The development described above may require research methodologies to manage complexity and promote sustainable system innovation [23]. The multilevel perspective is used to structure the research to explore drones in healthcare and to broaden the analysis from a traditional "technological-push" model towards system innovation. The multi-level perspective is described in more detail in Section 1.2, but in brief, it illustrates three analytical levels: landscape, regime, and niche. The *landscape* is the macro-level, presenting the broader political, social, and cultural processes of society that are expected to change slowly and outside the control of socio-technical regimes. The *regime* is the meso-level formed by practices, routines, and leading technologies. The *niche* describes the micro-level, providing 'protective spaces' for radical and path-breaking alternatives whose performance may not be competitive against existing technologies in the regime.

There is a broad body of literature that relates to system innovation. This literature also includes the research streams of socio-technical theory and transition management theory [29]. While the multi-level perspective explores sociotechnical systems, transition management is more focused on exploring alternative trajectories [30]. However, because both research streams have a similar basis, this thesis does not distinguish between them. A system approach acknowledges system mechanisms, such as feedback loops and information flows, that change over time in a dynamic and nonlinear way [31–33]. In general, system theory can be used to bring the technical and social parts together [34]. However, this depends on how strongly researchers or other stakeholders are bound by their disciplines, background, or culture [35].

The word "system" comes from the Greek verb *synhistanai*, meaning "to stand together" [36]. System science, a neutral term chosen by Mabry et al. [32], involves conceptualizing the solution space as a system of interrelated parts (i.e., the "big picture"). This meaning and conceptualization of a system are applied in this thesis.

Technological components can be combined in new ways and integrated with other technologies. For example, Gruson [37] wrote that intelligent processes, integrating big data and real-time data management, automation, the blockchain, the Internet of Things, and enhanced user experiences are all critical elements of the smart digital laboratory. This refers not only to the technology itself, but also to the knowledge around a particular technology. Technological innovations in one industry could potentially result in spillovers in other industries or sectors that share similar needs [38, 39]. For example, Meunier et al. [39] argued that, although military drones appear different from driverless cars, they share the same autonomy capacities, making these two types of vehicles more similar to each other than, for example, differing cars are in their car-system.

In this thesis, "healthcare" or "health care" encompasses actions that aim to promote, maintain, and restore individuals' physical and mental well-being. "Healthcare systems" describe a broader perspective that includes all actors, institutions, and resources that undertake health actions [40].

Sustainable healthcare systems refer to multiple characteristics that vary from affordability, accessibility, acceptability, and quality of health services [22]. It may also be used as an overarching term that covers change initiatives aimed at value-based healthcare [41]. In Norway, sustainable healthcare systems are intended to be economically efficient, have high patient satisfaction, sufficient staffing of health professionals, high-quality equipment, and highly respected hospital institutions [42].

The innovation process can be described as the development and implementation of new ideas. People engage in this process over time and in relation to others, often in an institutional context [43]. Furthermore, the innovation process includes the diffusion of technologies and markets [44]. From a system perspective, positive feedback loops may support the broader acceptance of an idea, whereas negative feedback loops may limit and restrict wider diffusion and institutional changes [45].

The use of drones in the healthcare context is relatively new. Although there is growing literature on its potential, drone technology, as well as knowledge of flight surveillance, take-off and landing, and loading, is still immature [46–49]. Furthermore, drones have mainly been evaluated from a technical perspective, and the evaluation of drones from a socio-technical perspective about the public acceptance of drones and those in the system affected by drones may add value. In a socio-technical approach, technology is used as a supportive tool in the system's design. This includes changes to processes, followed by changes to the role that humans play in the system. If existing services need to be redesigned, the attitudes of healthcare workers towards drones in healthcare may therefore be relevant. It is the ambition of this thesis to contribute to this knowledge.

1.2 Research perspective

This section presents a brief overview of the concept of system innovation, which forms the theoretical background for this thesis. However, a detailed review is considered outside the perspective of the thesis.

1.2.1 Socio-technical systems

Socio-technical systems describe the interrelatedness of human, social, and organizational factors. Human actors, social groups, and their relations sustain and reproduce these systems. Rules and institutional factors coordinate human actions. In addition, socio-technical systems also consider technical aspects in the design of organizational systems [50]. Figure 1.1 illustrates the three interrelated analytical dimensions. Note that, in practice, these dimensions are always interrelated.



Figure 1.1: Three interrelated analytical dimensions (adapted from Geels [51])

Geels [51] lists six interactions between these analytical dimensions:

- 1. Elements and linkages in socio-technical systems are reproduced by actors in their activities.
- 2. Actors operate in the context of institutional rules and routines.
- 3. Actors support and reproduce rules in their activities.
- 4. Socio-technical systems form a structuring context for activities and enable and constrain social groups.
- 5. Social groups share rules, but rules can also be embedded in artifacts and practices.
- 6. Technological conditions also shape rules and institutions.

1.2.2 The multi-level perspective

Socio-technical system frameworks have been developed to analyze and understand the transition processes of technology and social change [52]. The multilevel perspective has been firstly developed by Geels [51, 53, 54] and was further elaborated by other authors such as Gaziulusoy to understand how system innovation occurs [29]. The multi-level perspective framework in the thesis is illustrated in Figure 1.2.



Figure 1.2: Multi-level perspective of the system innovation model (adapted from Geels [51, 53, 55])

The structure of a socio-technical system in the multi-level perspective has three levels: micro (niche), meso (regime), and macro (landscape). These terms may be regarded as corresponding to the institutional levels of the clinic, hospital, and healthcare, based on Fuenfschilling et al. [56] who showed that the conceptualization at the institutional levels provides a suitable structure for the analysis of socio-technical systems.

The multi-level perspective conceptualizes technological change in terms of four phases [31, 57]: the introduction of technology within the existing environment; the exploration of functionalities and user preferences; application of change in daily operations; and gradual replacement of existing solutions.

Additionally, the multi-level perspective assumes that transitions have three dimensions: the speed of change, the size of the change, and the period of change [31]. Positive feedback loops (self-reinforcing) increase the rate of change, creating more of their output. Negative feedback loops (self-correcting) dampen the change or modulate its direction [58].

Raven et al. acknowledges that the three analytical levels do not describe actual entities but are helpful for categorizing and better understanding socio-technical change [59]. The levels used in this thesis emerged during discussions and are based on a common understanding.

This thesis applied the regime level to hospitals with different healthcare groups. Niches are equal clinics where new ideas are introduced and tested. The landscape level is translated into health, in general, as part of larger social trends and changes.

Heiskanen et al. [60] argue that although the importance of users has been acknowledged in transition analysis, most transition analyzes operate with data on a macro level. Therefore, the conceptual organization of clinics, hospitals, and health in this thesis was beneficial in having a user focus.

Hospitals are influenced by pressures from the general healthcare system and niche innovations on a clinical level. Change might be triggered by iterative development circles within the regime or more radically from niche developments. Regimes (i.e., hospitals) represent the central dimension, characterized by a stable configuration in the socio-technical system until a new technology emerges [61, 62]. Compared to other industries where a core technology may dominate, hospitals function through the interplay of multiple technologies [62].

The multi-level perspective divides the regime into five sub-regimes: technological and product regime; science regime; policy regime; socio-cultural regime; and user, markets, and distribution networks regime.

- **Technological and product regime** The technological and product regime can be characterized by its ability to meet functional requirements. The main functional requirements for institutional services are predictability, quality, and efficiency.
- **Science regime** The science regime is characterized by the acquisition of new knowledge and expertise.
- **Policy regime** The policy regime can be characterized by the relation, actions, attitudes, and values between employees and leaders.
- **Socio-cultural regime** The socio-cultural regime has the main requirement of establishing a learning environment and routines as an organization responds to innovations, in this case, drones.
- **User and market regime** The user and market regime is required to adjust and adopt new technologies in line with an organization and its stakeholders' needs and abilities.

Niches (i.e., clinics) are a protected space for radically new innovations; they require protection because their cost efficiencies, technical performance, and usability often need improvement [63]. Drones can be considered a niche innovation. The challenges of drone transport related to meteorological conditions, transport safety and security, and currently short flight ranges, still need to be solved. Niches provide the space for experiments and learning processes and the development of social networks that support innovation [64].

Deeper structural trends characterize the landscape (i.e., healthcare). The landscape level cannot be deliberately changed, according to multi-level perspective [65]. However, landscape changes can generate opportunities for niches [66]. The multi-level perspective refers to such opportunities as "windows of change."

Raven et al. point out that an essential contribution of the multi-level perspective is the fact that transitions need a fruitful coupling of developments at all three levels [59]: First, niches are needed to develop, nurture, and experiment with new technologies or practices without the regime's interference. Second, the regime needs to be open and adaptive to accept innovations. Third, the pressure of the landscape creates windows of opportunity for innovations to break through. In a later phase, this can contribute to an attitude that is more open to innovation in general.

Raven exemplified destabilizing landscape pressures to include socio-economic, demographic, political, and international trends, but also events like wars or environmental disasters [59]. According to Geels, the landscape may also include shared cultural beliefs, symbols, and values that are difficult to deviate from [51].

The landscape concept has been criticized, among others, for being relatively unspecific and for focusing too much on destabilizing influences on regimes. Geels suggests paying more attention to landscape developments that may help stabilize exiting regimes [66]. For example, the automobility regime is increasingly facing landscape pressures that destabilize (i.e. discussion of climate and fossil fuel use) and stabilize (i.e. preferences for speed, time savings, and private property) [67]. In terms of drone implementation, growing technological awareness of, for example, automated solutions may be seen as a landscape process that can challenge regime performance and create opportunities for niche developments.

In summary, in healthcare systems, there may be a tension between the perceived need for potentially disruptive innovation and the aims of ensuring patient safety and high-quality care with equitable access [68]. Consequently, Geels points out that a balanced assessment of transitions should address destabilizing and stabilizing landscape developments [66].

1.3 Research motivation

The motivation for this thesis is an interest in the broader system innovation that may arise from the establishment of drones in healthcare. The challenge of innovation should not only be considered from an economic point of view but also in terms of the societal consequences in the aftermath of the innovative activity [69].

System innovation is defined as a transition from one socio-technical system to another. The term transition reflects the idea of a path towards a new state [70]. In this thesis, the term transition is used interchangeably with system innovation. A transition ends when the emergent technology is embedded in the regime and thus creates a new social-technical system.

According to Geels, technology is often the entry point to the multi-level perspective and the socio-technological system at hand. This is, to some extent, due to implicit technological determinism. However, as Opoku et al. [71] argue, even if technological determinism is a mechanical mode of explanation, it does not mean that technology is uncontrollable or that technological development within a particular context will lead to automatism. Transitions are rather the outcome of co-evolutionary processes. This means that the interaction between subsystems influences, but does not determine, other systems [30]. For example, technical change co-evolves with institutional change—although they shape one another, neither determines the other.

Besides generic system and level assessments, Geels [72] requests that user practices, such as the initial development, diffusion, uptake in user practices, and social embedding, could receive more attention. This insight is very relevant for this thesis because a user focus is necessary to understand how drones are perceived today and how they could be socially embedded at the Oslo University Hospital in the present and future.

The research in this thesis is motivated by the learning of applying the multilevel perspective in an innovation project. Furthermore, the thesis hopes to contribute to practical implementations of multi-level perspective in similar novel technologies and innovation projects, as well as for the development of sustainable healthcare systems. The research is also methodological and pays attention to user practices and thereby tries to respond to Geels's request.

1.4 The role of design

Following system innovation in this thesis was less about the design of a drone and the service it provides, but to investigate which transitions take place when technologies such as drones enter the social context of a healthcare system. Researching views, relationships, and design roles for and within transitions is valuable because it enlarges possible solution spaces from a mere technological tool to new concepts and services [73].

The multi-level perspective presents itself here as a systemic and dynamic tool for the design of multi-level perspective about drone technology, which has recently also been applied in other disciplines, for example, land use policies [74] to expand the innovation potentials. Furthermore, the engagement with a broader range of users may, for example, contribute to the generation of ideas for new drone applications and services, which, in turn, may strengthen their integration into healthcare services, as, for example, Hiebert et al. claim [75].

Design approaches such as system-oriented design or transition design aim to create a new mindset for system innovation [76–78], and in parallel an awareness of designers to be responsible and to address social, economic, and ecological issues [79]. However, although system-oriented and transition designs have a common argument for the necessity of system innovation, they argue differently about design for complexity.

On the one hand, the system-oriented framework uses Gigamaps to visualize complexity and allow the team members to iteratively investigate, learn and adjust the system of interest [80, 81]. Sevaldson, who advocates system-oriented design, described Gigamaps as "information clouds" that try to grasp, embrace and mirror the complexity of problems [80]. Sevaldson recognizes that other fields are also concerned with solving complex problems; however, he believes that designers should use design practice to generate knowledge [79].

On the other hand, transition design draws on approaches from social science to understand the social roots of complex problems [77]. Moreover, Irwin et al., the representatives of transition design, argue that new ways of designing need to be informed by outside design knowledge to gain a deeper understanding of how to design for change/transitions in complex systems [77].

In *The Routledge companion to design research* [82], Daniela Sangiorgi and Kakee Scott argue:

When designers widen a project's scope or system boundaries to incorporate broader societal challenges, they must contend with questions of system-level change, even if working on smaller and more manageable manifestations of the wider scale phenomenon. In response, design approaches are undergoing re-tooling to adapt.

According to the authors, this has led to increased interest in design research, which draws on approaches from the social sciences developed over a long history of addressing systemic societal changes. Therefore, this thesis is positioned within design research that may be expanded into transition-scale projects according to Irwin et al. [83]. Design research may take several forms, such as observing what people do in various situations, asking questions, searching for information, making and testing prototypes, and speculating on fictional future visions [82]. In this thesis, the emphasis was placed on user inclusion through user-centered and human-centered design principles, such as understanding socio-technical systems and user needs and thus providing a foundation for a possible future design of the system [84].

1.5 Aims and research questions

By taking real cases and examples from current healthcare systems, the effect of implementing drones and if and how drones contribute to a more sustainable
healthcare system can be analyzed. Additionally, the consequences from a sociotechnical perspective can be evaluated.

The overall research question for this thesis is as follows: How can drones become a part of the future healthcare transport system?

Based on the background mentioned above and the overall research question, the following three subquestions are formulated:

- 1. How can current healthcare transport systems be understood as socio-technical systems?
- 2. What attitudes do healthcare workers have towards drone transport in health-care?
- 3. How can drones be implemented in a future healthcare transport system?

1.6 Research papers

This thesis is based on three research papers that have been published in international scientific journals:

- Paper 1: H. E. Comtet and K.-A. Johannessen, "A socio-analytical approach to the integration of drones into health care systems," *Information (Basel)*, vol. 13, no. 62, p. 62, 2022. DOI: 10.3390/info13020062
- Paper 2: H. E. Comtet and K.-A. Johannessen, "The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems," eng, *International Journal of Environmental Research* and Public Health, vol. 18, no. 5, pp. 1–15, 2021, ISSN: 1661-7827
- Paper 3: H. E. Comtet, M. Keitsch, and K.-A. Johannessen, "Realities of using drones to transport laboratory samples: Insights from attended routes in a mixed-methods study.," eng, *Journal of multidisciplinary healthcare*, vol. 15, pp. 1871–1885, 2022

1.6.1 Contributions of the research papers

The different contributions from the papers to the research questions are presented in Table 1.1.

The intended contributions of these papers are as follows:

(1) The multi-level perspective is analyzed to understand how it can be applied in an innovation project. By applying a broader analytical framework, the papers seek to contribute to an overall understanding of a socio-technical approach to innovation projects. Whether this may lead to increased use of the multi-level perspective in similar innovation projects will undoubtedly depend on the individual settings for a given case.

Research paper	1	2	3
Research question			
1) How can the current healthcare transport systems			_
be understood as socio-technical systems?	•		•
2) What attitudes do healthcare workers have			
towards drone transport in healthcare?		•	
3) How can drones be implemented in a future			
healthcare transport system?			•

Table 1.1: Relationships between research papers and research questions

- (2) The impact of drones on organizational processes has been scarcely studied and is unknown. The studies in this thesis reveal that although internal hospital transport times often may have delays because they are not being monitored or processed for optimization, a simple reconfiguration of the transport system could easily be obtained without complex organizational processes.
- (3) Drones in healthcare are perceived positively and, therefore, may theoretically be valuable to envision future scenarios and improve current processes. Thus, a contribution of this thesis is that technological innovations and system innovation approaches may add value and support continuous improvement by facilitating technological and sustainable change.

The main conclusions are listed below:

- 1. Socio-technical system theory and the multi-level perspective provide tools for understanding and facilitating innovation processes.
- 2. In a broader perspective, drones can support the aim of having a healthcare system with good working conditions because they can be used in cocreation sessions to shape how laboratories want to work.
- 3. Drones allow for developing scenarios that invite user participation.
- 4. Currently, drones for laboratory logistics are not economically sustainable in an urban environment, but they can advance the sustainable goal of good equipment for transporting laboratory samples.

1.7 Organization of this thesis

Chapter 2 positions this thesis. Chapter 3 describes the research design and strategy for this thesis to answer the research questions. Chapter 4 briefly presents the results of the three research papers. Chapter 5 discusses the main findings concerning the research questions. Chapter 6 provides some remarks and implications. Methodological considerations and implications for validity, ethical considerations of this thesis, and lessons learned and limitations are presented. Finally, Chapter 1: Introduction

Chapter 7 provides some conclusions and suggests future research directions.

Chapter 2

Research position

The following sections position the research in this thesis. First, it discusses methodological awareness and a pragmatic approach. Next, reflexivity issues are discussed, that is, my role as a researcher, insider limitations and challenges, and political and ethical considerations. In the end, the analytical strategy is outlined and the research is described, including its validity.

2.1 Epistemological awareness

This section elaborates on epistemological awareness and provides a design, system, and research perspective on the pragmatic approach taken in this thesis. Epistemology describes the nature and form of knowledge [88] and is concerned with how knowledge can be created, acquired, and communicated [89].

The philosophical conceptualization of knowledge at the levels of ontology, epistemology, methodology, and methods is widely accepted [89–91]. However, there is a philosophical debate about whether the different levels inform each other in a top-down alignment [91, 92], or whether the alignment goes both ways, as Morgan [90] suggested.

In a top-down alignment, the constructivist typically relies on qualitative methods, whereas the positivist uses quantitative methods. Morgan argues that our ontological worldview should not automatically influence the research we do [90]. Instead, we should decide what is essential and appropriate. In that sense, the pragmatic researcher is flexible in selecting the research design and methodology that are most appropriate to address the research question [93].

A pragmatic approach is helpful regarding the multi-level perspective because it includes hard and soft system elements (see the following sections). Geels [51] describe technologies as having a certain "hardness" and consisting of systems of human relations and institutional rules. This is methodologically connected to a pragmatic attitude.

From a design perspective

A practical method that has proven valuable in creating sustainable innovations is to move between divergent and convergent thinking [94]. Convergent thinking is linear, systematic, and analytical, whereas divergent thinking is outward, openended, creative, and flexible.

Divergent and convergent thinking can be aligned with the logic of deductive and inductive reasoning. The deductive approach is characterized as a top-down process that provides general statements. On the contrary, inductive reasoning describes an explorative bottom-up approach. Pragmatism is associated with abductive reasoning, moving between deduction and induction [93]. Figure 2.1 illustrates the three described forms of reasoning.



Figure 2.1: Deductive, inductive, and abductive reasoning

From a system perspective

Ison [36] described the hard side of systems as systematic and linear, following step-by-step thinking to name, build, explain, or discover systems and use systemic models as epistemological devices. The soft side of systems is described as systemic thinking in terms of wholes. Continuing along this line, the systemic models used in learning can lead to changes in understanding and practices. The system designer facilitates learning and change based on accommodations between different interests. Ison [36] compared positivistic and constructivist perspectives from a systems point of view and described the thinking traditions associated with hard and soft systems (Table 2.1).

From a research perspective

In research, a deductive process is typically associated with positivism, whereas an inductive process is associated with constructivism [91]. The positivistic perspective is based on the work of Simon [96], in which knowledge is logically connected [97]. Schön [98] criticized this approach because it may solve wellformed problems. However, particularly in design research, researchers and designers may often be confronted with complex or ill-defined problems, sometimes termed wicked problems. Schön suggested a constructivist paradigm that focuses

Hard system thinking traditions	Soft system thinking traditions
oriented to goal-seeking	oriented to learning
assumes the world contains systems that can be engineered	assumes the world is problematic, but can be explored by using system models
assumes system models to be models of the world	assumes system models to be intellectual constructs
(ontologies)	(epistemologies)
talks the language of 'problems' and 'solutions'	talks the language of 'issues' and 'accommodations'
Advantages	Advantages
allows the use of powerful techniques	is available to all stakeholders, including professional practitioners; keeps in touch with the human content of problem situations
Disadvantages	Disadvantages
may lose touch with aspects beyond the logic of the problem situation	does not produce the final answers; accepts that inquiry is never-ending

 Table 2.1: Comparison of hard and soft traditions of system science (Ison [36], adapted from Checkland [95])

on enabling people's creative capabilities through experiential learning. In organizational and management research, systemic thinking can be related to the ability of organizations to learn [99]. Table 2.2 summarizes the different concepts.

	Positivistic	Constructivist	Pragmatic
Approach	Top-down	Bottom-up	Flexible
Logic	Deductive (convergent)	Inductive (divergent)	Abductive (convergent and divergent)
System knowledge	Name, build, describe, or discover systems and use systemic models as epistemological devices	Used in a process of learning that can lead to changes in understanding and practices	Knowledge useful for problem solving and action
Authors	Simon [96]	Argyris and Schön [100], Schön [98]	Dewey [101, 102], James [103] and Cross [97]

 Table 2.2: Ontological and epistemological choices (adapted from Auernhammer [99])

The following section summarizes the influence of the different concepts on this thesis.

Positivistic A positivistic approach for technological and business dimensions lies, e.g., in technological determinism. The idea is that technology is separated from society and is first developed before being tested in society. Therefore, regulation may be necessary to control technology and avoid unintended consequences. For example, although drones and technology, in general, are continuously under development, many technical engineering issues still need to be solved before they can fly autonomously. In addition, logistical models, business values, and technical requirements may be presented objectively and numerically to be measured. From a socio-technical perspective, top-down processes from the landscape into the regime may drive change. In particular, public expectations of healthcare and political guidelines for the use of technology in healthcare actively influence the regime. Such processes can create opportunities for niche developments.

- **Constructivist** The social construction of technology explores the relationship between technology and society. In this process, technology and society shape each other over time. This extended perspective provides the opportunity to develop user-centered services and system innovations, as advocated by the multi-level perspective. The multi-level perspective broadens the analysis from technology and business perspectives to socio-technical systems. Although top-down processes may create tensions in some regimes, the bottom-up process tries to mediate the different regime dimensions. For this thesis, the inductive approach is used to interpret the findings and insights.
- **Pragmatic** The pragmatic focus is located at the regime level in this thesis. In this institutional context, understanding practical performance may lead to the identification of changes. Top-down and bottom-up processes drive changes in system transitions, and identifying which approach offers the best leverage for change during periods of innovation is an essential factor in the ultimate success of any transition [104]. Leverage points are mechanisms by which incremental changes produce more significant changes in complex systems [33].

2.2 Reflexivity

This section discusses my hypothesis and positioning in the research project. I would describe my ontology as being most closely linked to the position of soft system thinking. However, I support the argument of Ison [36] that practicing both soft and hard system thinking gives access to more choices. Furthermore, although I position myself in the constructivist dimension, I firmly believe that sustainable development and interdisciplinary projects need to move from the top-down approach, from ontology to methods, and towards a flexible and pragmatic approach. This is because to be able to work together in interdisciplinary teams, a flexible and pragmatic approach would make it possible to work together towards solutions beyond disciplinary theories and methodologies commonly. However, holding on to your own beliefs and methods may prevent interdisciplinary work because other approaches and methods may be criticized.

The broader design of socio-technical systems considers human, social, and organizational factors alongside technical factors [50]. Socio-technical transitions, i.e., the change from one socio-technical system to another, are multidimensional and can be studied from different angles by different disciplines. The crossover between ontologies may be particularly fruitful for multidimensional phenomena such as socio-technical transitions [105, 106].

The research setting has provided a reward for personal learning on multiple

levels. However, time limitations, coursework, and the writing process take time in Ph.D. projects, resulting in not everything being investigated and covered.

2.2.1 Insider limitations and challenges

Conducting research as an insider may often include relations that already exist. However, since I started as an outsider in this research project, the relationships actually needed to be established first. Additionally, because I lack relevant healthcare experience, the initial phase was characterized by learning and building relationships and agreeing with the supervisors on practical knowledge, which can be challenging due to many different opinions and priorities. The HELSEVEL project contained specific requirements and the choice of multi-level perspective as the theoretical framework was questioned. Although research on socio-institutional healthcare systems receives some attention [41], concepts such as the multi-level perspective and system innovation are still relatively unfamiliar.

My doctoral program is rooted in the Department of Design at Norwegian University of Science and Technology (NTNU). The research is rooted in the Intervention Center as an employee at Oslo University Hospital. Although beneficial to my development, the collaboration of both institutions was not strengthened beyond my involvement. One reason may have been that the research project was already initiated and that NTNU was not one of the initial stakeholders in the HELSEVEL research project.

2.2.2 Political considerations

Political considerations describe how organizations are continuously developed to serve their overall societal assignment best. It may need a vision, goals, and values to create a sense of community.

Institutional politics includes the hidden, behind-the-scenes interests and commitments that form part of any institution's social structures. Therefore, researchers need to be conscious of these largely invisible structures [107]. Political disputes may include power differences in the organizational hierarchy, the allocation of space and other resources, gender politics, and who collaborates with whom.

Social structures reflect normative, regulative and cultural cognitive elements [108]. Geels conceptualized the policy rules as one dimension in the socio-technical system (Chapter 2).

Institutions are fundamentally political, and the politics of an institution will undoubtedly influence researchers.

2.2.3 Ethical considerations

Ethical considerations are intended to investigate how to act, how to understand and evaluate the outcomes of actions, and how to act (morality) accordingly. Ethical issues occur in all phases of a research project [109]. However, in this thesis, data collection was the main ethical consideration. Respecting the particular characteristics of healthcare facilities and disturbing clinical processes as little as possible were essential ethical perspectives. Another issue involved building trust and conveying the extent of anticipated disruption in gaining the necessary access. To avoid disturbing the meetings between patients and providers of clinical services, the research project focused solely on clinical service units.

2.2.4 Research journey

My research journey began with a book on interaction design [110], where I first discovered my interest in socio-technical systems. After my Master's degree in Interaction Design, I worked as a UX designer. Continuing my research journey, I realized the importance of continuous learning and reflexivity to broaden my understanding of socio-technical systems and design processes. As a UX designer turned researcher, I sought to expand my knowledge beyond user interfaces and go deeper into system innovation.

During my PhD project, I had the opportunity to attend the Service Design Conference in Scotland just before the lockdown, where I was exposed to presentations on designing for sustainability without having all the answers. This further piqued my interest in exploring the implications of socio-technical systems on sustainability.

The COVID-19 pandemic, while limiting physical interactions and networking opportunities, surprisingly had minimal impact on data collection for my research. However, it inadvertently encouraged a more theoretical focus, allowing me to read more literature and study the multi-level perspective as a theoretical framework for system innovation design.

In the practical aspect of my research, conducting drone tests proved challenging. We hypothesized that drones could fly and aimed to test a logistical model for Oslo University Hospital [111]. However, logistical issues with drone manufacturers and restrictions on airspace due to emergency helicopter approaches hindered the smooth execution of the experiments [112]. Another time, a drone fell from the air, highlighting the complexities of implementing drones in real-world scenarios.

While more iterations would have been ideal, the lack of access to drones for experimentation and the need to coordinate with external vendors added constraints to the research process. As a designer in the project, my focus was on conceptualization and understanding how drones could contribute to system innovation rather than being directly involved in drone design or defining specific needs and requirements.

Throughout this work, I have appreciated the value of learning, reflexivity, and continuously broadening my understanding. Engaging in a design process when the problem is not clearly defined, the design approach may require a more exploratory and iterative mindset. Therefore, the research in this thesis tried to gather as much information as possible about the problem domain. Reading articles, talking to stakeholders, conducting interviews, and research to gain a deeper understanding of the context and needs of the laboratories.

In conclusion, my research journey has been characterized by a focus on sociotechnical systems, system innovation, and the exploration of drones as a potential innovation in the healthcare logistics domain. This has involved navigating the limitations posed by the pandemic, facing challenges in conducting practical experiments, and embracing a theoretical perspective to strengthen the conceptual foundations of the research.

2.3 Analytic strategy and plan

The analytical strategy of this thesis reflects the integration of drones into healthcare systems in a mixed-methods investigation. The analytical plan comprises three research questions. The relationship between the research questions and the multi-level perspective is illustrated in Figure 2.2.



Figure 2.2: Relationship between the research questions and the multi-level perspective

- The first research question asks how the current healthcare transport systems can be understood as a socio-technical system.
- The second research question concerns the attitudes of healthcare workers towards drone transport in healthcare.
- Finally, the third research question examines empirical effects and asks how drones may be implemented in a future healthcare transport system.

The quality of the research is evaluated through evaluation criteria. The validity criteria used in this thesis are described below:

Quantitative and qualitative approaches are validated to verify data quality, results, and the author's interpretation of data results [91]. In this thesis, the concept of validity is inspired by Herr et al. [113], who linked five criteria (dialogic, outcome, catalytic, democratic, and process) to the goals of research as indicators of good research. To this list, I add the criterion of construct validity [114]. Table 2.3 lists the validity criteria and goals, which are described more thoroughly in the next section.

#	Validity criteria	Goals of research
1	Dialogic validity	Generation of new knowledge
2	Catalytic validity	Education of both researchers and participants
3	Democratic validity	Results that are relevant to the local setting
4	Process validity	Sound and appropriate research methodology
5	Outcome validity	Achievement of valuable outcomes
6	Construct validity	Collection of data

Table 2.3: Validity criteria and goals of research (inspired by Herr [113] with #6 from Yin [114])

2.4 Research description

The previous section described the research strategy and charted six validity criteria. This section outlines the research description and applies the validity criteria. Different research strategies are chosen to contribute to a broader understanding of the integration of drones into healthcare systems. Table 2.4 provides an overview of the purpose, methodology, data, methods, analysis, and results of the three articles:

	Paper 1	Paper 2	Paper 3
Purpose	To examine the MLP and potential for spin-offs related to the organizational transitions and implementation of drones	To examine variables among employees at the three hospital trusts	To explore the current transport service and identify topics for improvement
Methodology	Qualitative content analysis	Quantitative analysis	Mixed method
Sample and data sources	Review based on 25 articles, supplemented with further publications	Random sampling for a study group of 400 subjects	Purposive sampling of carriers working in the transport department
Research methods	Scoping review, snowballing, and supplement literature	Self-administered questionnaire	Questionnaires, shadowing, interviews and multiple time measures from the transport system
Data analysis	Inductive and deductive process	Descriptive analysis	Convergent design
Conclusion	The MLP is useful for innovation and for understanding the interactions between different system levels. Ethical topics and public acceptance are placed into a socio-technological perspective	Drones are positively perceived. Working in an innovative environment, having experienced technological change, and having leadership that supports new ideas are drivers of beliefs in drones	Suggests experiments and learning from transport by cars and drones in combination

Chapter 2: Research position

23

Table 2.4: Overview of the three articles included in the thesis

24 Hans E. Comtet: Drones in Healthcare Systems: Insights from a Multi-Level Perspective

2.4.1 Dialogic validity

The dialogic validity refers to the generation of new knowledge and the articles published in this research project. Two papers have been published in MDPI journals because of the broader scope, wide readership, rapid turnaround, and relatively high impact factor (>3) of these journals (the impact factor of a journal is calculated in terms of how often other researchers cite articles in the journal). The general success of MDPI and their growth with multiple journals and the treatment that some researchers have experienced from specific journals have been widely debated. The Norwegian National Publication Committee has recently created a new level X for possibly predatory journals and publishers [115, 116]. MDPI, particularly the journal Sustainability, is under investigation and is categorized as level X while the investigation is ongoing.

2.4.2 Catalytic validity

The catalytic validity refers to the spiraling change within the iterative cycle of the plan–act–observe–reflect. Iteration is an accepted characteristic of action research. The overall course of the research for this thesis is depicted in Figure 2.3.



Figure 2.3: Plan-act-observe-reflect cycle

The research project started with extensive observations and subsequent learning. The first article is based on a scoping review. The development of a survey instrument led to the second article. Finally, knowledge of system transitions led to a mixed-method study of carriers that formed the third article.

2.4.3 Democratic validity

Democratic validity refers to the inclusion of involved parties in the investigation and its relevance in the local setting. The research project has focused on healthcare workers and applied qualitative and quantitative methods to assess relevance for the local setting. The system perspective ensured a broad perspective and understanding of multiple factors that may be important for the integration of drones into healthcare systems.

2.4.4 Process validity

Process validity refers to the extent to which ongoing learning is supported. A helpful approach to process validity is to have a hierarchy of themes to guide the production of several outputs on the same topic [117]. For example, the uppermost themes describe what motivated the researcher to become involved, followed by themes more closely related to the actual research project and organizational context.

2.4.5 Outcome validity

Outcome validity is dependent on process validity, described in the previous section. The outcome validity describes the successful outcomes of the research project and includes internal and external validity. Internal validity can be described by the degree to which the results represent the truth in the subject being studied, that is, the strength of the research methods and design. External validity describes to what extent the research outcome can be generalized.

2.4.6 Construct validity

The construct validity requires that correct operational measures are identified for the concept being studied [114]. Trochim [118] conceptualizes this using theoretical and practical dimensions. The theoretical dimension describes how beliefs and theoretical assumptions have been captured and measured, that is, operationalized by the research into the practical dimension.

The validity claims related to the findings in this thesis are discussed in Section 5.7

Chapter 3

Data and methodology

The three articles in this thesis have different data sources and approaches, but are all concerned with integrating drones into healthcare systems. This chapter first outlines the research design in the overall study, then explains the data collection methods and presents methods to determine and assess the research questions in this thesis. Finally, this chapter is summarized.

3.1 Research design

The research reported in this thesis consists of quantitative, qualitative, and mixedmethods. Methodologically, the thesis is grounded in the Ison [36] titled *Systems thinking and practice for action research*. However, as introduced in Chapter 1, the *system thinking* is based on the neutral term *system science*, which Mabry et al. [32] used to describe the conceptualization of a solution space in which the whole system and the interrelated parts can be analyzed. Figure 3.1 illustrates the basic idea of system thinking. An observer has determined the border of a system of interest in a given context. The border differentiates the system of interest from the environment [36].

3.2 Methods of data collection

As described above, in terms of the overall research design, this thesis draws on different methods and data. In this section, the choice of methods is outlined and described.

3.2.1 Mixed-methods

Maxwell [119] describes three purposes for using multiple methods. The first is triangulation, to check whether different methods support a single conclusion. The second purpose is to gain information about different aspects. The third purpose, which is applicable in this thesis, is to provide a more prosperous and better



Figure 3.1: Key elements that result from system science (adapted from Ison [36])

understanding of the bigger picture. In addition, using a combination of methods can strengthen the validity of research results.

Statistical analysis and probability calculations help to understand uncertainties and make comparisons straightforward [120]. However, complex and multiscale problems can be challenging to express using purely quantitative data [31]. The combination of methods can improve the depth of understanding [121]. For this, a mixed-method approach should take advantage of qualitative and quantitative methods as complements to each other.

Complementary methods examine different elements of a specific phenomenon [122]. This thesis comprises a quantitative analysis of a questionnaire and data on transport times. The qualitative findings address observations, shadowing, and interviews.

Interpreting qualitative data

The qualitative content analysis describes a development process in which categories emerge through incremental revisions and are enriched while working on the text [123]. First, however, the methodological concept of emergence is discussed [124]. In this thesis, the multi-level perspective is used as a conceptual framework and forms a skeleton for the data. Kelle [124] argued that, in the context of exploratory and interpretative research, such an approach might be beneficial because researchers may otherwise have long processes and generate a lot of data that needs to be analyzed. In contrast, the argument for the emergence of codes is open-mindedness.

However, researchers have always used the existing theoretical knowledge to describe and explain observed phenomena [124]. Researchers should be careful not to force data into specific categories. Furthermore, it is necessary to control whether a chosen theoretical concept leads to the exclusion or neglect of some phenomena [124].

In positivist research, data are considered to be obtained independently of the researcher [125]. However, in qualitative research, the researcher must know his role during the study. Moreover, qualitative research includes personal experience, i.e., the researcher may introduce his knowledge and create individual assessments of the findings. Therefore, qualitative research and its interpretations can never be separated from the researcher's reflexivity or personal views and characterizations. Section 2.2 elaborates on my personal beliefs that I brought with me to this thesis.

Quantitative data and statistical analysis

Quantitative data analysis is a structured process. According to Creswell et al., [91], the data is first evaluated with computer programs such as EXCEL or SPSS. The data are then explored to develop a general understanding of the database. Finally, results may be illustrated in statements, tables, or figures. The process of Creswell et al. applies to the quantitative data analysis in this thesis.

3.2.2 Case study: Oslo University Hospital Trust, Oslo, Norway

Case studies provide valuable insights in the early stages of any research project and can contribute to system design and further investigation.

Oslo University Hospital Trust used as the case in this thesis consists of four main institutions located inside the area of Oslo, and several minor facilities (see Figure 3.2): Rikshospitalet (The National Hospital, offering local, regional, and national services), Ullevål University Hospital (local, regional, and national services), Radiumhospitalet (a specialized cancer hospital), and Aker University Hospital (local and central hospital).

In 2018, Oslo University Hospital had a total patient activity of 94,000 hospitalizations, 45,000 daycare treatments, and 853,000 outpatient consultations. The hospital had 24,000 employees, and patient treatment took place at more than 40 locations within a radius of 20 km. Hence, Oslo University Hospital is a large institution on a European scale and provides services from local hospital treatment to advanced specialized services and transplantations.

With its complete range of medical services in many locations, large-scale economy, and variety of advanced solutions, Oslo University Hospital is a relevant case for the assessment of drone solutions as an extensive service for the time-critical clinical transport of biological samples across large and complex institutions.



Figure 3.2: Oslo University Hospital Case

At Oslo University Hospital, blood samples are transported across the laboratories at the four locations. Blood samples are taken in different medical subspecialty units (wards, outpatient clinics, centers) that offer various inpatient treatments, day treatments, and outpatient services.

Based on numbers from 2018, the total analysis performed at Oslo University Hospital is more than 22 million analyzes per year. The laboratory at the National Hospital has nearly twelve million analyses. The Ullevål laboratory performs more than six million analyzes. The numbers for the other hospitals are 1,5 million analyses at Radium Hospital and more than 500,000 at Aker Hospital.

The scenario considered in this thesis is an "inner-city project" aiming to ensure that biological material can be transported across the four Oslo University Hospital locations within defined time intervals. This inner-city scenario requires drones to fly over urban areas at high frequency and at all times of the day. If this project proves to be realistic, it may be relevant for other urban hospitals that could benefit from consolidating duplicated services.

The purpose of exploring the drones at Oslo University Hospital was to explore the potential for improved laboratory logistics for the transport of biological material across the four locations based on the hypothesis that transport times could be improved, and as a possible instrument to eliminate duplicated laboratory activities across these laboratories. The thesis aims to explore whether specific organizational topics would have to be considered and modified to obtain the full potential of such solutions.

To broaden our data related to the knowledge and expectations of future drone solutions in Paper 2, we took advantage of the fact that two other hospitals in our region also consider drones as part of future logistics. Both Innlandet Central Hospital and Vestre Viken Central Hospital plan for new centralized hospital structures where drone solutions may improve logistics. Both hospitals have a similar geographical environment with multiple districts and rural areas where biological and other goods are transported by ground vehicles. We got approval to extend our exploratory questionnaire in Paper 2 to these two hospitals.

Today's drones fly with a maximum speed varying from 40–50 km per hour. This represents a limitation of the time difference between cars and drones traveling in parallel. Longer driving distances, particularly when roads have to detour around or bypass natural barriers such as forests, lakes, and mountains, while drones can take shortcuts, drones may still yield some value, and this should be expected to increase in the future with more robust drones having larger load capacities and distance ranger.

3.3 Methods to determine whether current healthcare systems may be understood as socio-technical systems

3.3.1 Observation and shadowing

Ethnography is the combination of observation, interviews, and participation [126]. The purpose of the observations is to observe what people (in this thesis, healthcare workers) say and do when collecting blood samples. Additionally, this understanding can be enriched with photos, notes, and system maps to collect and group information. Finally, shadowing is a common approach to follow group members as they go about their business, the researcher asking questions as needed for clarification and interpretation [126].

Gold [127] described the different roles of observations, ranging from being a *complete participant* to a *participant-as-observer*, then to a *observer-as-participant*, and finally to a *complete observer*. My role during the observations can be described as an observer as a participant. Further, Gold points out that a researcher in this role, compared with a complete participant or a participant-as-observer, is less vulnerable to reminding oneself of the primary role of being an observer. However, contact with the informant is brief compared to the other roles, which can lead to misunderstandings. Applying this method in the present thesis may confirm or reveal gaps in the information gathered through the other data collection methods and aid in identifying questions for future research.

3.3.2 Scoping review

Scoping reviews can be used to identify research gaps and summarize research findings [128]. Similar to systematic reviews, they follow a structured process. However, the review question can have a broader "scope" compared to systematic reviews. Although literature searches may be less defined than systematic reviews, they are intended to provide an overview of a given topic [129].

3.4 Methods to assess the attitudes healthcare workers have toward drones

3.4.1 Questionnaire

According to Salkind [130], questionnaires offer a structured approach to data collection, allowing researchers to efficiently collect large amounts of information from a diverse range of participants in a standardized manner.

Lyu et al. [131] emphasize that questionnaires are often used in innovation studies because questions can be created to ask about information directly.

Questionnaires have several advantages, including their ability to collect data from a large sample size, ease of administration, and the potential for standardized responses [130]. However, there are also drawbacks to consider. For example, Salkind [130] and Dillmann et al. [132] highlight that response bias, low response rates, and the potential for misinterpretation of the questions can limit questionnaires.

Furthermore, questionnaires do not allow the responses to be followed up and are not suitable for a deeper investigation of the responses [133]. Validity of the questionnaire should be ensured through pre-tests and test trials, thus identifying potential issues and allowing further refinement before a wider distribution [134].

Workshops are considered to develop the survey instrument, and the questions are further improved, revised, and categorized to collect information on various topics [135]. The collaborative nature of workshops allows the immediate identification of flaws and biases in the survey questions, leading to iterative improvements and increased validity.

However, there are drawbacks to workshops. For example, time constraints and scheduling challenges in workshops can prevent a thorough exploration of all aspects of the survey instrument. Although workshops provide valuable opportunities for collaboration and refinement of survey instruments [136], it is important to consider efficient use of time to maximize the quality of the survey instrument.

3.5 Methods to assess changes that drones may accelerate in healthcare systems, and how drones may become a part of the future healthcare transport system

3.5.1 Focus group interview

Focus groups are increasingly used, ranging from promotion and marketing to evaluations and social research [137]. One reason might be that individual interviews are considered more cognitive, whereas focus group interviews tend to present spontaneous, expressive, and emotional views [137]. This output can be relevant for marketing research that aims to predict and control consumer behavior [137]. Another benefit may be that focus groups enable an analysis of social

interactions that lead to interview statements [138]. Another reason for the popularity of focus groups may be their flexible placement within a research design, for example, front, back, or in parallel.

Excursus: focus group interviews

The planned focus group interviews turned into two sessions with experts and a discussion. The first group included five participants (three laboratory experts, one medical doctor, and one entrepreneur). The second group consisted of 12 experts in health-related logistics. The focus group interviews aimed to generate impressions of the institution and learn how experts talk about drones.

Each session lasted approximately one hour. The author of this thesis first presented the findings from Paper 3, followed by a discussion among the group for 30 min. During the discussion, various questions were asked, covering which institutional elements are considered unnecessary/redundant in the organization of today, which institutional elements are essential, and the ability of drones to contribute to institutional change.

The analysis of the interviews and the focus group meetings followed a simple structure, as suggested by Stewart [139], avoiding the need for a detailed analysis and achieving a faster conclusion. The interviews were transcribed and uploaded to Atlas.ti content analysis software (version 22.0.2).

Four categories emerged from the review: (1) Institutional barriers, (2) Ownership, (3) Support from drones, and (4) Institutional gains. The category most relevant to this thesis that emerged from the review is the support of drones and is discussed in Section 5.3.

Chapter 4

Results

This chapter briefly presents the main findings contained in the three articles underlying this research. Table 4.1 provides a general overview of this chapter.

Research paper	1	2	3
Research question			
1) How can the current healthcare transport systems			
be understood as socio-technical systems?	•		•
2) What attitudes do healthcare workers have			
towards drone transport in healthcare?		•	
3) How can drones be implemented into a future			
healthcare transport system?			•
Sections	4.1	4.2	4.3

Table 4.1: General overview of the results

The full versions of Papers 1 to 3 are included in this thesis after the bibliography. The remainder of this chapter is based on the three peer-reviewed articles.

4.1 Paper 1

A socio-analytical approach to the integration of drones into healthcare systems

H. E. Comtet and K.-A. Johannessen, "A socio-analytical approach to the integration of drones into health care systems," *Information* (*Basel*), vol. 13, no. 62, p. 62, 2022. DOI: 10.3390/info13020062

This study examined what knowledge of socio-technical theories may support the implementation of drones into healthcare systems. The study was built on a scoping literature review of topics related to the multi-level perspective and sociotechnical studies from differing arenas, supplemented with studies collected on a broader basis using the snowball effect. The multi-level perspective can provide a framework that can be helpful in understanding system innovations related to drones in healthcare [29, 57, 140–142]. For this paper, the multi-level perspective levels have been translated into a healthcare context and used to organize the literature findings. The study assumes a bottom-up process, starting at the niche level, where drones are first implemented before they generate innovations in a broader perspective. The useful implication of this perspective is that it reveals the dynamics and linkages between the three levels. Furthermore, by reversing the multilevel perspective and starting at the niche level, this study may provide an interesting perspective for niche actors working with innovation and interested in systems innovation.

During the analysis of the documents and the development of the discussion, several additional publications were included to supplement the analysis, following a deductive identification of the relevant articles. The entire process is described in the research article included at the end of this thesis (see Paper 1).

4.1.1 Main findings and conclusion

The multi-level perspective may be useful for innovation and understanding the interaction between different levels of the system. Ethical topics and public acceptance are put into a socio-technological perspective. The articles considered in this study indicate that it is necessary to build a cross-cultural approach and establish a discourse across several stakeholders to develop a future sustainable healthcare system. Information about community needs can guide policymakers and decision-makers in appropriately integrating drones.

4.2 Paper 2

The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems

H. E. Comtet and K.-A. Johannessen, "The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems," eng, *International Journal of Environmental Research and Public Health*, vol. 18, no. 5, pp. 1–15, 2021, ISSN: 1661-7827

This study investigated the expectations and beliefs of 400 employees at Oslo University Hospital, Innlandet Hospital, and Vestre Viken Hospital. To extend the study beyond Oslo University Hospital, we included Innlandet Hospital and Vestre Viken Hospital because these two institutions are planning new hospitals and considering drone transports as part of future transport solutions. The study used a self-administered questionnaire and aimed to capture "the overall picture" with respect to the technological interest and competence of individuals, the knowledge and expectations of future drones, and indicators of local culture. The 36 questions in the questionnaire included 16 yes—no questions, 13 multiple-choice questions, and 7 Likert scale questions (rating individuals' agreement with a statement on a scale of 1—5).

The importance of organizational structures and expectations specifically related to the use of drones in hospital systems has been explored by [143] and others [75, 144–146]. These authors agree that this is an important issue that needs further investigation. The study confirmed that factors such as different attitudes by profession and gender might differ in different systems and cultures ([143]). Therefore, this study extends this understanding to a Norwegian setting. Furthermore, the study developed a novel questionnaire that can be used by other researchers to study local settings and systems.

4.2.1 Main findings and conclusion

Drones are perceived positively. Working in an innovative environment, having experienced technological change, and having leadership supporting new ideas drive beliefs in drones. Innovative leadership and culture may strengthen the acceptance and implementation of drone transport in healthcare. Furthermore, it may shape the organizational structure necessary to adapt to technological change. The potential for organizational development was also discussed in Paper 1.

4.3 Paper 3

Realities of Using Drones to Transport Laboratory Samples: Insights from Attended Routes in a Mixed-Methods Study

H. E. Comtet, M. Keitsch, and K.-A. Johannessen, "Realities of using drones to transport laboratory samples: Insights from attended routes in a mixed-methods study.," eng, *Journal of multidisciplinary health-care*, vol. 15, pp. 1871–1885, 2022

This article focused on the carriers at Oslo University Hospital who currently transport biological material by car along traditional circular routes.

The study explored how the current transport service is organized, identified areas for improvement, and investigated carriers' perceptions of how drones could be integrated into or substituted for their current services. The findings in this article were based on 16 questionnaires, nine shadowing reports, eight interviews and multiple measurements of transport times. In their review paper, Kellermann et al. [147] argue that current transport problems on the ground should not be the drivers of drone technology development. In our study, we respond to Kellermann et al. call for increased engagement of social sciences. Understanding work practices using novel technologies is emphasized in socio-technical approaches

[148]. Additionally, stakeholders interviews can be conducted in preparation for or as part of a transition process [149].

The involvement of hospital transport service providers as potential stakeholders and resources in the implementation of drone solutions contributed useful knowledge to optimize the existing transport solution. Lessons learned may be relevant to achieving a more innovative transportation system. This study argues that drones may represent a logistics solution with relatively little impact on organizational processes. However, the potential for additional organizational and innovative processes should always be considered when incorporating new technologies. In addition, existing transportation solutions should be carefully analyzed to identify potential improvements before implementing drones. The implications of this study illustrate how existing systems can be improved and how new systems can be created with a combination of old and new technologies.

4.3.1 Main findings and conclusion

New solutions and approaches may challenge the status quo, namely, established work processes and routines. The study revealed that the routes have existed almost unchanged for more than a decade with only minor adjustments. Therefore, the study suggests combining experimentation and learning from transport with cars and drones.

Chapter 5

Discussion

The following sections in this chapter summarize the findings of the three articles in relation to the research questions: (1) How can the current healthcare transport systems be understood as socio-technical systems? (2) What attitudes do healthcare workers have toward drone transport in healthcare? (3) How can drones be implemented into a future healthcare transport system? This will be followed by a broader discussion and evaluation of the overall research. Chapter 6 will discuss the overall contribution of this thesis.

5.1 The current healthcare system as a socio-technical system

Paper 1 and Paper 3 approached the healthcare system as a socio-technical system using the multi-level perspective as a conceptual framework and tool.

Paper 1 discusses the conceptual levels of the multi-level perspective from the dimensions of the niche upward. The multi-level perspective model is often used the other way around. However, assuming that drones are implemented in a "bottom-up" process, the model has been adapted accordingly.

Paper 3 hypothesizes that the current transport options can be improved through a reconfiguration pathway. New solutions and approaches may challenge the status quo, namely established work processes and routines. Existing systems may survive for long periods if traditional modes of operation lock them in. However, in the early phase, new technologies may develop without any consensus on their overall benefits. For example, it may be difficult to comprehend how a new technological tool should be integrated into existing workflows to improve existing processes.

According to the transition management theory, the micro-behaviors that emerge as individuals adapt to innovation may contribute to overall organizational behavior at the system level. For example, a transition perspective might suggest that the transportation department rebrand itself as an innovation hub for logistics, with an open culture for experiments and change (Paper 3). Indeed, in our study Paper 3, the users said, "The old structure needs to be cracked to be more innovative."

Paper 2 suggests that exploring and understanding an organization's readiness for technology-generated system changes may be crucial in implementing most types of technology and drones. A socio-technical approach might be helpful in this implementation process because it highlights the importance of using technological transformation as an opportunity to improve operations by combining organizational and technical needs [148, 150]. The ability to change and improve existing solutions by implementing technological innovations—in contrast to modifying new solutions to fit existing routines—may be a critical factor in the success of future healthcare organizations.

One possible value of recognizing healthcare systems as socio-technical systems is that new technology can be used to improve existing processes. The sociotechnical theory assumes that the implementation of new technology is followed by social change. Furthermore, from a systems perspective, positive feedback loops may be identified and may strengthen the diffusion process of technology implementations and, hypothetically, also for the implementation of drones. The broader perspective of the system can also identify challenges and negative feedback loops that present potential barriers to the integration of drones.

A systems perspective also provides a space to create sustainable service concepts, such as the transport department rebranding itself as an innovation hub for logistics. Furthermore, transitions can be pushed in the desired directions by considering different aspects of the systems perspective. Socio-technical theory recommends that technological acceptance should not be accepted by default, but often asks for thorough management and leadership to be achieved.

The implementation of drones should include an evaluation of the acceptance of drone integration because assuming such approval by default may lead to later rejection (Paper 1). Supporting the more substantial transformation of multiple services beyond logistics offers a space to create sustainable service concepts and ideas with a broad perspective. Therefore, the implementation of drones should include an evaluation of public acceptance. Public acceptance and ethical considerations as part of the system development were identified in the scoping review (Paper 1) as socio-technical issues.

The objective of Paper 1 was to assess the value of taking a socio-technical approach. However, the applied search terms did not identify any studies that used the keyword "socio-technical." Despite this, my overall interpretation of the findings is that although it is not necessary to operate drones, a socio-technical approach may help achieve a culture open to innovation and change.

In socio-technological terms, ethical considerations and public acceptance are highly relevant to the implementation of drones. The transport routes investigated at Oslo University Hospital are not complex and are not focused mainly on transport times (Paper 3). Substantial improvements should be realistic, with or without drones. Therefore, drones may represent a logistic solution with relatively little impact on organizational processes and may have little effect on other clinical services outside of the transportation service. Although diffusion across technological and medical processes may be somewhat limited, the roles of opinion leaders and change agents in the hospital system may contribute to the emergence of a consensus on various innovations [151]. The multi-level perspective approach makes it easier to find the level of intervention to implement niche technology with the most success.

Even if a socio-technical multi-level perspective combined with transitions research represents a powerful analytical force, it is not the only way to investigate changes. The multi-level perspective conceptualization was intended to broaden the understanding of implementing drones into existing systems. Several future research opportunities, such as the impact of drones on organizational processes and social dynamics in the adoption of drones, may expand the understanding of these questions and nudge transitions in the desired directions. Aspects such as the need to build a cross-cultural approach, cultivate a discourse among several stakeholders (Paper 1), and ensure optimal service design and technology acceptance (Paper 2) may strengthen the innovation process.

Drones in healthcare may have a higher public acceptance than in other domains [146]. However, the recent Australian study by Wiedemann et al. [152] found that, for example, privacy issues are still mentioned as a major concern for the general population. Additionally, emergency drones may be supported, whereas drones for transport may not. For the diffusion and acceptance of drones, the multi-level perspective in this thesis considers lead users, healthcare workers, and the public at the respective niche, regime, and landscape levels. Paper 3 and Paper 2 included healthcare workers and carriers at the regime level. For widespread adoption, public acceptance and support will be essential [147, 152, 153].

5.2 Attitudes of healthcare workers toward drones

An argument from a socio-technical perspective is that technology is never just technology, but that any implementation of new technology also entails social change [34]. Following the argumentation from Paper 2, that technology may be used as a tool to change and improve existing solutions may lead to questions about how much organizational change is desired or needed and to what extent technology can be used to generate change. Unfortunately, such questions probably do not have a general answer because they may differ between different settings.

However, an approach that has shown its usefulness in dealing with them is the multi-level perspective. The multi-level perspective may also be considered a component of transition studies that provide a holistic view of system changes, as the multi-level perspective requires the involvement of different actors and multiple interactions in a system and dynamic alteration at different systemic levels.

Systemic perspectives may facilitate exploring and understanding an organization's readiness before new technologies, such as drones, are implemented. The benefits of this exploration may be that current work practices are improved, in contrast to modifying new solutions to fit existing routines. This ability may be critical for future healthcare organizations to cope with technological change. Moreover, investigating the acceptance of this new technology and including all stakeholders are necessary measures to ensure the full potential of drones in the long run.

Opinion leaders and change agents may contribute to such an innovation journey. However, the improvement process may have to be tailored to the existing organization and culture.

Paper 2 found that the use of drones in healthcare is positively perceived in professional groups, ages, and locations and that the supporting factors for a positive perception of future technological transitions may be related to working in an innovative environment, having a leader who supports new ideas, and having previous experience with technological change. In our study, this was associated with the belief that drones will be used in future healthcare systems.

A possible interpretation of these findings is that to realize the optimal and extensive benefits of implementing drone solutions; the managerial approach should be open-minded, stimulate innovation, encourage experimentation, and support the necessary risk taking. Supported by the harvested literature, Paper 1 discusses how the creation and maintenance of links and networks between clinics, institutions and health systems may contribute to a broader diffusion of drones. Leaders should maintain connections and networks between the different dimensions of the system in such transition processes. It is generally accepted that innovative leadership and culture may possibly strengthen the acceptance and implementation of technology in a manner that shapes the organizational structure necessary to adapt to technological change.

These results confirm previous studies that have suggested that more experience leads to increased familiarity and the ability to learn about technology in general [154]. Furthermore, the experience and knowledge of early adopters/believers can positively contribute to the continued implementation of technology [155].

Although focusing on micro-level innovations may be crucial in the initial implementation phases, a systemic perspective may facilitate the integration of drones into the healthcare system by considering the entire process, encompassing social and technical aspects. This may help improve current work practices and maximize the long-term potential of drones.

5.3 Accelerated changes of drones in healthcare systems

Paper 3 advocates existing solutions being studied and improved. A socio-technical perspective may benefit such analyses because it suggests a broad evaluation of multiple interactions on different systemic levels.

At Oslo University Hospital, the ground transport that was studied was not primarily focused on transport time, but designed to maintain flexibility for laboratories that are carriers' customers. Implementing drones may require strict and timely routings with minimal deviations due to airspace and air traffic regulations. Flexible departures in the current system may violate such regulatory criteria.

From an economic point of view, the willingness to pay for drone solutions may be limited if existing ground transport and loading times are acceptable. Thus, the competitiveness of drones may be related to whether they have lower costs than the current ground transport system for comparable service times or real-time savings.

Paper 1 states that the institutional and clinical contexts in which drones may be a valuable medium for transporting biological material have not been extensively investigated. It is unknown whether the introduction of drones as a transport solution in healthcare will produce significant gains.

Furthermore, whether the implementation of drones for transport in healthcare systems will be primarily based on technical drone requirements alone or whether it will require specific organizational or system-related modifications needs further exploration [156]. For example, few drone studies have addressed the requirements for high-volume transport across larger healthcare institutions or time-critical transportation that requires extreme and regular uptime.

Optimal service design often starts by including stakeholders to shape the best environment for new solutions, thus strengthening the innovation process [157–159]. In this context, attitudes and culture among hospital employees in institutions that plan to implement drones as a future transportation option should accept this new technology.

The interview with the focus group interview with experts in relation to drones served as a means of sharing innovation knowledge [149]. The key point was to implement drones supplementary to conventional transport. This was a valuable add-on. Drones could be used for specific routes, i.e., during the night. Concerning the fact that drones can be expensive to maintain, the participants argued that traditional ground transport at night is also not cheap. Another option for drones outside of regular working hours could involve a collection site where samples could be received from a drone. Additionally, drones could be used for emergency sample analysis, which currently requires taxis.

The above discussions illustrate that drones are perceived as valuable in complementing existing ground transport. However, there is some tension in terms of challenging existing routines. Hospitals seek to improve existing routines that ensure quality through standard procedures and repeatable patterns. Routines in hospitals are at the core of many daily operations [160] and are often difficult to change. However, routines are sustained and evolve through the agency and choice of individuals [160].

Goh et al. [161] conducted an interesting study on incremental changes to work routines in healthcare systems. They found that personal innovation is critical in guiding all change processes in addition to leadership. In particular, users who engage in more exploration have a greater chance of determining which features of the system work most effectively in a given situation. Furthermore, the user experience and sense of control are higher when routine change is perceived volitionally compared to when change is imposed by technology. Technology ownership increases with the agency through innovative behaviors [58].

Based on the aforementioned, strengthening the agency of healthcare workers is an essential first step in creating ownership and increasing the control of technology-initiated change processes. In addition, it may be important to strengthen the understanding that one's own actions contribute to system innovation. Furthermore, insights from system innovation and socio-technical theories illustrate how these innovative healthcare systems can be established and nurtured [68].

5.4 Navigating transitions

Leadership is essential in innovation projects, as is the ability to think holistically and be open to new ideas. Paper 2 found that having a leader who supports new ideas is one of the drivers of the belief that drones will be used in future healthcare systems. However, professional roles during system changes are not well defined. For example, managers at the operational level may need to ensure efficiency, while managers at higher levels may promote the implementation of change and initiate change processes to ensure the legitimacy of the organization [162]. The complexity increases when multiple change contexts run simultaneously, whereby leaders may have multiple roles and be required to juggle operational and strategic goals.

The advantage of governing according to concrete and quantitative goals is the ability to compare, learn from, and compete against other institutions. This paradigm is called new public management. However, although new public management measures relevant goals and provides a helpful tool, it does not necessarily open up new ways of thinking. Instead, the focus is on improving existing solutions.

Networked governance seeks to control not through hierarchy or market incentives, but in a relationship with others and through collective learning [163]. In Norway, the networked governance model is called a trust-based governance paradigm.

Paper 3 considered the question of how much organizational change is desired or needed and to what extent technology can be used to generate change. Leadership needs to address such questions, and some leaders may be better suited to such tasks than others.

Manzini [164] created a mode map for designers that can be adapted to reflect the general relationship between problem, change, knowledge, and doing. The map is illustrated in Figure 5.1. Actors may consider themselves as problem solvers who are aware of the problem. They eventually try to learn from others how to solve the problem and are interested in change or actively working toward change as promoters of social innovation.

Rydland focused on the role of middle managers and developed a typology in which the Market Protector and Executor take a narrow approach to be change



Figure 5.1: Typology of designer (adapted from Manzini [164])

agents, focusing on current customers or their team [162] (Figure 5.2). In contrast, the Renewer and Co-creator take a broader approach and emphasize the entire organization. Rydland argues that the Executor and Co-creator both perceive a need to challenge their mindset in the event of a change, whereas the Market Protector and Renewer do not perceive a similar need.



Figure 5.2: Typology of middle managers (adapted from Rydland [162])

As Sosa et al. [165] and Nelson et al. [166] suggest that designers can play an important role in change initiatives because they bring the creative agency to create new products and services or even design new socio-technical systems [29]. However, the agency of designers may still be limited in influencing change at the societal level. According to Gaziulusoy [167], one primary reason for this limitation is that short-term requirements informed by company strategies bound designers.

How much should designers control a design process? Coops et al. raised this question in a discussion among experts working in these fields on how design may evoke sustainability and transitions and transformations [168].

Designers may not be able to solve social challenges by themselves, and transdisciplinary ways of working and competencies are needed [169]. Furthermore, although a general design process may act as an underlay for bringing different practices together, it will not necessarily be the designer who will be in charge of what is being designed. Existing research gaps and transdisciplinary challenges require continuous collaboration and learning from each other.

5.5 Improving work processes

Improving the current work processes instead of modifying technological innovations to fit existing routines has been identified in the literature as a critical success point for sustainable healthcare systems (Paper 2).

From a perspective of technological determinism, technology is responsible for the outcome of a transition, and the existing social context is sometimes seen as a barrier that must be overcome. From a social constructivism perspective, technology is understood as a tool for shaping practices. Additionally, technological tools shape human behaviors and organizations. Therefore, social constructivism suggests being proactive and creating societies by deciding how technology can support humans.

The technological substitution process characterizes transitions as radical innovation [170]. Innovation competes with the existing system and replaces the old solution. Eventually, incumbent actors are replaced by new entrants. According to Berkers et al., [170], this way of conceptualizing innovation is often found in the literature [171–173]. The focus is on technical and business dimensions.

In contrast, the multi-level perspective broadens the analysis from a technological and business perspective to a socio-technical perspective, focusing on the wider processes in system innovation [72].

According to Geels [72], reconfiguration suggests that the institution, actors, and practices change during a sustainable transition, whereby old and new elements are combined to lead to gradual system changes.

Section 1.2.2 introduced the terms positive feedback loops (self-reinforcing) and negative feedback loops (self-correcting). According to most authors, this thesis has studied, positive feedback loops reinforce the direction of change [33].

In a positive feedback loop, innovative and supportive management, together with previous experience of technological change, strengthens the belief in technological innovations, in this thesis, exemplified by drones. Leaders who develop links between the dimensions of the clinic, hospital, and healthcare system can
create additional reinforcing cycles, resulting in positive system effects. Facilitating experiments and establishing a culture that is open to change will, over time, generate an environment in which users adjust and accept technology and change. Leaders and change agents are necessary to create and maintain reinforcing cycles.

In a negative feedback cycle, a negative image of a new technology or system can discourage users from modifying their routines to accommodate the system. Lack of involvement and knowledge about the new technology may result in skepticism. Negative cycles ultimately lead to locked-in systems that are resistant to change.

Paper 3 suggests that laboratories are invited to participate in co-creation workshops to influence how the transport service should be designed. Furthermore, Paper 3 revealed that the transport department could rebrand itself as a hub of logistics. The visual observations presented in Paper 3 reveal that the transport boxes could be standardized. Therefore, they may be suitable for drone transportation and could score higher in terms of user experience.

In general, this will contribute to making the department more appealing to applicants. However, it may also mean that some aspects become redundant in the process. The optimized routing model theorized that the reorganization of routes could reduce the number of carriers and cars (Paper 3). In her keynote presentation at IST 2021 (International Sustainability Transitions Conference), Fuenfschilling argued that the "deinstitutionalization" of actors and/or processes could be uncomfortable, exhausting and emotional [174].

Optimizing services provides the opportunity to make them more efficient and streamlined. However, realizing that the technology itself does not determine this is essential. Instead, such processes can be constructed and managed so that they are not experienced as uncomfortable, exhausting, and emotional, but as a chance for improvement. The co-evolution comes from both the social and technical sides. Service design methods may bring people from different functions together and introduce them to new and exploratory learning situations [175]. For example, initial questions may ask, "Can the actors be removed, or their roles changed, to improve the user experience?"

In our studies, drones for medical transport are positively perceived in healthcare professions, ages, and locations (Paper 2). Furthermore, strengthening positive feedback loops and innovation can enable transformative change (Paper 1). In the short term, it is suggested that agency, drone experiments, and learning may facilitate incremental change (reactive change). Radical change may need to be avoided because the various lock-in mechanisms mean that drastic change has too much impact on established work processes. However, in the long term, a more radical fundamental system change may be obtained (Paper 3).

5.6 Contextual understanding

The reconfiguration path described may lead to more sustainable solutions and a more holistic context for future solutions. However, designers cannot force people to make changes they are not ready for, as Shedroff wrote in his book [176]. Instead, designers must understand their biases, preferences, and values. Furthermore, understanding the mission, culture, goals and vision of the organization makes it easier to act on the values and core meanings that enable and support them.

Healthcare systems are diverse. The impact of innovations can be difficult to predict due to the multiple dynamics of professional groups, policies, and opinions that tend to influence and be influenced by the system [25].

The extent to which institutions can recognize the value of new external information and then assimilate and apply this information is an interesting perspective that has been studied in innovation studies (i.e. [177]). Healthcare systems may need to build organizational capacity and the ability to handle change initiatives in addition to daily hospital operations.

Organizational learning can be described as the exploration of new possibilities and the exploitation of old processes. March [178] compared the two approaches and argued that exploring new possibilities is less specific, more distant in time, and more distant from daily operations. On the contrary, the feedback from exploitation is faster and more precise than that from exploration. Innovation journeys (i.e., the process from initiating to implementing an innovation) are, to a large degree, open and uncertain. They may have longer time horizons and more diffuse effects than the further development of existing ones. Therefore, the exploitation processes would appear to be more common [178].

Oslo University Hospital runs a continuous improvement network. This network is driven by change agents, who facilitate change initiatives in addition to their regular health-related work. The network organizes courses on facilitation and problem solving. Continuous improvement follows the cycles of Plan-Do-Study-Act (PDSA) [179]. In contrast, the multi-level perspective promotes system innovation. Continuous improvement compared to system innovation is illustrated in Figure 5.3.

The PDSA model is based on three essential questions: (1) What are we trying to accomplish? (2) How will we know whether a change represents an improvement? (3) What changes can we make that will result in improvement?

The answers to these questions depend on the complexity of the situations and the preferences of those doing the work [179]. In the multi-level perspective, learning is based on Kolb's experimental learning cycle [105, 180]. Cycles of thinking and action are stimulated by reflective thinking through a transformation of experience [180].

According to Matsuo et al. [181], Kolb's learning model corresponds to the Plan-Do-Study-Act (PDSA) cycle. Concrete experiencing is similar to doing, reflectively observing corresponds to studying, abstract conceptualizing resembles



Figure 5.3: Continuous improvement corresponding with system innovation

planning, and actively experimenting parallels acting. The advantage of these similarities is that the process can be more easily understood and applied to facilitate experiential learning.

The size of the solution/learning space may be different in system innovation, and hospitals may not have the luxury of moving too far from daily operations (Figure 5.4).

Inspired by LEAN processes with the goal of creating value with less work, decision grids can be applied to judge ideas and determine which are the most likely to be effective [182]. Problems that make a major improvement and are easy to implement may be favored over wicked problems with the potential for a major improvement but are difficult to implement. Large learning spaces may be more familiar in design. Öztekin et al. argued that system innovation inhabits a similar learning space, in which problem and solution strive to be discovered [183].

Langley et al. [179] distinguished between reactive versus fundamental changes. Reactive changes keep the system of interest running day-to-day at the current level of performance, whereas fundamental changes create a new performance system. However, as Langley et al. argued, it is important to make reactive changes to remove any current problems until suitable fundamental changes can be developed to eliminate them completely.

The scope of a systems innovation project can vary. For example, it can range from a participatory design process to lay the foundations for system design and establish a shared vision to move the project forward [84] to a topic for the disciplines of transportation and urban planning [147]. The scope of this project was to investigate whether and how drones can become part of a future healthcare transport system.



Figure 5.4: Organizational learning versus learning space

However, system change can be difficult because it involves a large number of goals and stakeholders. In addition, Clegg [73] claims that systems thinking is not universally agreed upon or observed in everyday practice. Furthermore, the idea that the designer should have all the answers and skills to solve wicked problem seems unrealistic, according to Thorpe et al. [184].

Shedroff suggests that each time a piece of the system is considered, there is an opportunity to transform that piece into a more integrated whole that meets sustainability goals [176]. For Clegg [73], system innovation needs to be made explicit so that it can be debated, challenged, defended, and improved.

Thorpe et al. [184] refers to DiSalvo et al. [185], Hillgren et al. [186], and Manzini et al. [187] who describe this design for social innovation as both a "process" and a "practice". A process to create networks and interactions of social actors to create and test systemic prototypes and address multiple goals and wicked problems. A practice in which designers apply their skills to the realization of devices and spaces that make it possible to continue collaborating.

Depending on scope and context, multi-level perspective and transition design can provide the tools to understand the process and practice to support system change. A designer may sometimes have a leading role as an expert in the system change, and other times may not have a leading role, but being a team member who can contribute with design knowledge [184]. Regarding self-reflection, a socially responsive designer may act to bring about societal change with the collaborators and resources at hand, settling for the best that can be achieved in a given context, according to Thorpe et al. [184]. However, from my own experience, this can be frustrating at times, but must be accepted when designing for system innovation where different stakeholders and goals need to be unified. On the other hand, the opportunity to define problems and possible solutions and to set the framework for the innovation project also provides a tremendous opportunity to be creative and work toward change with the available resources.

5.7 Research validity

In this section, the overall research is evaluated. Furthermore, the different validity claims introduced in Section 2.4 are discussed in relation to the findings.

5.7.1 Dialogic validity

I am the first author of the three articles included in this thesis and all have been peer-reviewed. All articles are indexed in PubMed. All journals are level 1 scientific journals [188]. Although two of three articles have been published in MDPI journals, I have experienced a well-organized and structured review process with the MDPI journals Information (Paper 1) and International Journal of Environmental Research and Public Health (Paper 2). Overall, this suggests that dialogic validity is high with respect to peer review and professional comments.

5.7.2 Catalytic validity

One limitation of this study is that drones have scarcely been used in practice and that the implementation of drones has mainly been discussed in theory. Additionally, analyzing the theoretical implementation of drones from a broader sociotechnical perspective has led to an analytical focal point. However, the insights of this thesis have generated specific findings that can be used to continue research and development on iterative innovation cycles. This suggests that the catalytic validity is high.

5.7.3 Democratic validity

Laboratories that analyze samples and carriers transporting samples have been included in the research investigation. Furthermore, important variables for integrating drones into healthcare systems among healthcare workers have been examined in the form of interviews and observations. The test flights also confirmed that, technically, the drones could fly efficiently between the National Hospital and Ulleval Hospital. One limitation of this thesis is that the laboratories did not participate in a co-creation workshop to design a drone-based service that would provide them with added value. This lack of involvement, for example, discussing clinical times and the need for flexibility concerning the strict time requirements from scheduled drone flights, shows limitations in terms of democratic validity. 52 Hans E. Comtet: Drones in Healthcare Systems: Insights from a Multi-Level Perspective

5.7.4 Process validity

Concerning the research project, the initial theme was to ask relevant questions about the integration of drones into healthcare systems and identify how much drones impact institutional organizations. A broader system perspective evolved during the project and allowed the organizational consequences to be explored on a broader scale. Figure 2.3 shows the themes inside the plan–act–observe–reflect cycle.

The broader system perspective ensured that ongoing learning was supported. However, with the danger of moving away from the main interest of investigating the implementation of drones for the transport of laboratory samples, it was difficult to maintain relevance and not drift into other related topics.

An exploratory approach may produce less certain results. However, through collaborative data analysis sessions and discussions, this thesis has maintained a strong degree of relevance. Combining medical and healthcare expertise, design, and system theory with innovation and socio-technical theory was enriching. Furthermore, workshop sessions with bio-engineers, laboratory employees, and healthcare professionals allowed an abductive process that facilitated the creation of new knowledge and insights [189]. This suggests that the process validity is high.

5.7.5 Outcome validity

Internal validity

This thesis shows that minor organizational changes are needed to implement drones. However, the broader discussion in this thesis on innovative mindsets and drivers should be applicable not only to the implementation of drones, but also to technological innovation in general. The systemic perspective and combination of complementary methods suggest high internal validity.

External validity

This thesis has created various outcomes. However, the generalizability of the results is limited. The findings are restricted to a Norwegian setting and may depend on local cultures and environments. Furthermore, compared to other cities, Oslo may have shorter distances between the relevant facilities. In addition, entrepreneurship research has shown that decision-making processes are rarely based on systemic analysis [190]. However, this thesis provides useful information for the ongoing development of drone services as sustainable business models in similar healthcare contexts. This suggests that there are certain limitations to the external validity.

5.7.6 Construct validity

Construct validity in this thesis concerns the multi-level perspective, particularly how it can be applied in an innovation project.

Transitions imagine future system states that are more sustainable than existing systems. Scenarios for future drone-operated societies range from integrated and connected healthcare systems [191, 192] to speculative fiction [193], exploring social and political structures. The drone scenario in this thesis is built on the objective of having a more reliable, efficient, and qualitative transport service. Paper 3 found that the current laboratory system is based on flexibility. Therefore, this thesis has generated relevant questions that need to be answered by laboratories.

The multi-level perspective attempts to solve wicked problems [194], such as climate change, population growth, or inequality, from responsible production to sustainable healthcare systems. The wickedness of sustainable healthcare lies in the difficulty of predicting the implications of technological innovations [23]. However, despite the label "wicked," such challenges are contextual and may be understood differently as underlying problems [195]. This leads to the dilemma of "solutionism," i.e., because both a problem and a solution are presented, there is a suspicion that this represents a method trying to legitimize its own existence by constructing problems.

Wang et al. [196] criticized the multi-level perspective for its lack of actors and agency. Although research has addressed this issue, Wang et al. argue that this mainly covers the repositioning of actors. Another review by Kivimaa et al. [197] concluded that insufficient analyses have addressed what intermediaries (facilitators of innovation) do in the different phases of a transition, as well as how their roles change and what happens when a new system is stable and these intermediaries become redundant. Their review concludes that the often idealistic picture of intermediaries requires a more realistic approach and useful strategies for supporting system innovation.

The approach of this thesis has been realistic because concrete boundaries were applied in terms of the technology and system of interest. Although the multilevel perspective applies a wider learning space, this thesis found that learning corresponds to the methods used in healthcare. Overall, this suggests that the construct validity is high.

Chapter 6

Contributions and implications

This chapter summarizes the main contributions of this thesis and outlines its limitations. Although the research findings of this thesis are not generalizable, many of the insights may be applicable in other contexts and settings. Therefore, I conclude with some possible implications for leaders facilitators of change, designers, policymakers, and researchers.

6.1 Contributions

This thesis aimed to elucidate the essential dynamics of innovation on multiple levels and explore how the diffusion of drones could be strengthened. Because sustainable change may arise within systems, coordinating and orchestrating change processes are leadership tasks. This thesis advocates incremental change along a reconfiguration pathway. Furthermore, it advocates for the social construction of change.

The contributions of this thesis are twofold. First, the multi-level perspective was used to understand and learn about its application to an innovation project. Second, this thesis brought together the micro-perspective of drone development in the niche dimension and a broader analytical framework and mindset. The findings of the three articles published during this thesis and the previous chapter's discussion contribute to an increased understanding of a socio-technical approach to innovation projects. Furthermore, this thesis contributes to the more practical use of the multi-level perspective in similar innovation projects. The empirical and theoretical combination of quantitative and qualitative methods with a pragmatic approach has been fruitful in the pursuit of the research questions posed for this study and thesis.

Drones may have a limited direct impact on organizational processes. However, this thesis has revealed that the current ground-based system using cars in Oslo University Hospital is not ultimately based on transport times but on the flexibility required by laboratories. Carriers are asked to wait and postpone a certain departure, although this may cause a delay. Such flexibility is valuable for the requesting site. However, there is limited knowledge of whether this reduces the overall quality of service when all transport legs are considered as a whole.

The overall research question for this thesis is as follows: How can drones become a part of the future healthcare transport system? Drones in healthcare are perceived positively and therefore are valuable for envisioning future scenarios and improving current processes. Another contribution of this thesis is to show that technological innovations and system innovation approaches can be used for continuous improvement, thus facilitating technological and sustainable change.

6.1.1 Socio-technical reflexivity and sustainable change

Operationalizing the multi-level perspective has the potential to push system innovation to the surface. The professional network at Oslo University Hospital may promote transition design to facilitate technological change. Thus, transition design could become another tool of incremental change rather than a tool for radical system change [198]. However, socio-technical thinking is not widespread (Paper 1), and considering system innovation is primarily based on individual preferences. Therefore, identifying who may have a similar mindset may be a valuable first step in innovation journeys.

Paper 2 suggests that incentives for experimentation, learning, network building, and vision building are crucial elements in the push for change [57, 155, 199]. Furthermore, learning from innovation activities and innovation processes can lead to an understanding of how certain barriers may affect innovation activity [200]. Most importantly, Paper 2 found that previous innovation experience is a driver for the belief that drones could be used in hospitals. My results confirm those of earlier studies in suggesting that more experience leads to increased familiarity and the ability to learn about technology, in general, [154]. In addition, the experience and knowledge of early adopters/believers can positively contribute to the further implementation of technology [155].

The concept of sustainability itself is often ambiguous and controversial [105, 201]. Sustainable healthcare was described in the introductory chapter as economically sustainable, in addition to patient satisfaction, having sufficient numbers of health professionals, good equipment and good hospitals. Each of these criteria may be interpreted and weighted differently [22]. Therefore, system innovation for sustainability may be held back by the absence of shared visions [105].

Research on innovation places a greater focus on concretizing challenges. In particular, Schlaile et al. [202] and Uyarra et al. [203] suggest that questions should be addressed relating to directionality (what futures do we want?), legitimacy (why those futures and defined by whom?), and responsibility (transition by and for whom?). Geels and Stirling discuss the need to invest in experiments and learning to stretch existing routines and encourage reflexivity [105, 201]. This may also include critical questions, such as whether the public wants a regular drone transport service and what applications are acceptable in this case. Although critical social services such as healthcare, public transport and postal deliveries can be assumed to be prioritized, a public debate may be necessary [204].

6.1.2 The multi-level perspective as a tool and mindset

Whereas the multi-level perspective usually uses a top-down concept, Paper 1 discusses how the multi-level perspective concept can be applied in bottom-up change processes to produce fundamental change toward more sustainable health-care systems. The concept is used here for continuous improvement projects. Various types of improvement projects are conducted in different clinical departments. Change agents may facilitate these approaches by initiating new technologies, ideas, or concepts. According to the multi-level perspective, the regime level is a guarantor for stability, characterized by different lock-in mechanisms. People may not commit to change because established technologies or processes are favored. Leaders are essential in explaining why and how these changes are being initiated [179]. The multi-level perspective stresses that a deep involvement is necessary to ensure the successful implementation of innovations.

The requirements created by innovations at the clinical level may spark responses at the regime level. Paper 1 claims that prompt up-front regulation could enhance innovations and actively drive new developments. However, a holistic perspective at the regime or landscape level may need supplementary or broader information than at the specific clinical level to ensure that stakeholders take an extended perspective. The institution may need to engage crucial interdisciplinary teams to provide feedback on consequences beyond the initial perspective.

Although Paper 1 specifically considered drones, a process that cuts across different users and levels in the system may be universal and applicable in terms of continuous and various day-to-day improvements. Leaders are essential in this process, because they create and maintain links between the different dimensions of the system.

Public acceptance and the social benefit of innovations effectively enable health policies. Policymakers have an essential role in recognizing and actively promoting niche innovations by identifying lock-ins and helping to remove them. Kivimaa et al. [197] concluded that policymakers could support change agents in maintaining a broader scope, creating space for niche activities, and simultaneously destabilizing existing regime structures. Creating an environment for trust may contribute an additional nutrient for the diffusion of innovation. Finally, by defining common goals and timely expectations, all stakeholders may work for a shared vision for a future system (Paper 1).

The multi-level perspective allows for social and technical shaping [205]. On this basis, Smirnov et al. [206] argued that a coordinated policy approach should account for social preferences but also support experimental engineering earlier rather than later. Referring to Chapter 1, this is related to bringing together the social and technical sides.

Although focusing on micro-level innovations may be crucial in the initial in-

novation phases, a systemic perspective may facilitate the integration of drones into the healthcare system by considering the entire process, encompassing social and technical aspects. This would help improve current work practices and realize the full potential of drones in the long run. Socio-technical systems do not work independently, but involve an interaction between technology and the user environment [205].

6.2 Lessons learned and limitations

This thesis was conducted at the point where innovation research intercepts design, technology, and healthcare contexts. The multi-level perspective framework was used as a holistic model to capture the complexity. System innovation can be studied by different disciplines and from multiple perspectives [105]. This adaptability makes the MLP highly flexible [207], but also challenging to find one's position as a PhD researcher. However, multi-level perspective is also very intellectually rewarding because it provides a structure to analyze, understand, and discuss the dynamics of change.

This thesis started with an open question about whether and how drones may disrupt organizational processes. Approaching this research project with a broader analytical perspective was valuable because it allowed us to investigate multiple queries and dimensions. However, drones may only have a limited impact on organizational processes [146], and behaviors may not need to change significantly to implement drones. The scientific articles in our study indicate that building a cross-cultural approach and establishing a dialogue between several stakeholders is necessary to develop a future sustainable healthcare system. It may be essential to note that one may not need the multi-level perspective to generate such a desired behavior. However, because multi-level perspective sets this behavior in the context of system innovation, it can contribute to understanding why this behavior is essential in innovation.

In general, the multi-level perspective may be more relevant in innovation projects that have a broader scope, such as *remote patient monitoring* or the *adoption of telehealth solutions*. However, because this thesis shows how the multi-level perspective can be applied, it may be valuable for other system innovation projects.

Referring to the literature in this Chapter 5, the concept of sustainability can be ambiguous and controversial. Furthermore, sustainability criteria may be interpreted and weighted differently. Although sustainability generally tries to balance economic, social, and environmental factors, the reality for many healthcare institutions may be that they are the first to withhold financial requirements. Oslo University Hospital has decided not to continue drone research because drones are currently not considered economically sustainable.

Most research studies have some degree of methodological shortcomings. Limitations are related to the theoretical and methodological choices made throughout the research process. Although the methodological choices are outlined and discussed in Chapter 3, I now emphasize the research process.

Based on the above evaluation, this thesis has validity limitations in terms of democratic and external outcome validity limitations. The research findings are not generalizable. Furthermore, regarding reliability, i.e., to what extent another researcher could have found similar answers, the interpretation of the data makes the results demonstrable and testable; however, they are not provable. Furthermore, the findings primarily apply to a Norwegian setting. Nevertheless, the general discussion may also be interesting for other structures.

This thesis has addressed contemporary questions that are hotly debated in the media and among healthcare workers. However, the purpose of this thesis was not to present all the arguments, but rather to illustrate the area of tension of which healthcare is a part. Therefore, the argument is that because healthcare is complex, the implementation of technology in healthcare systems is complex (wicked problem).

6.2.1 Implications for researchers

Systems innovation has increased interest in several disciplinary and societal fields to understand innovation systematically. However, there are many research gaps, and Geels has called for more research on the intersection between technical and social innovation and the further development of the reconfiguration pathway [72, 208], i.e., the complementary implementation of technological innovations.

6.2.2 Implications for leaders

Using socio-technical transition theories to understand innovation journeys is fruitful because it highlights the relations and interactions between multiple elements of the system. However, transitions can also happen independently. Systems have multiple feedback loops, and analyzing which relations and interactions influence specific outcomes may be challenging to comprehend.

Empirical contributions highlight that innovative leadership fosters innovation processes that support new ideas and provides space for experiments. Furthermore, this encourages the broader diffusion of drones, because each small interaction can add up to more significant system changes. These issues are highly relevant for leaders and change agents in hospitals in Norway and beyond, that is, how their interactions in the system contribute to overall system innovation.

- Maintain linkages and networks between the multi-level perspective levels parallel to clinic, hospital, and healthcare system
- Be open-minded, stimulate innovation, encourage experimentation, and support risk-taking, thereby
- providing positive effects for the belief in drones and adaptation to technological change

6.2.3 Implications for facilitators of change

System transitions are long-lasting and nonlinear processes; it is unclear who should lead them. Leaders have power, but are rarely neutral in decision-making. Internal or external change agents may be able to facilitate change projects, but roles can change and it is unclear who should be doing what in which phase. For example, when is it a good time to introduce system innovation?

Technical engineers may be occupied with solving specific problems, for example, a code that does not work and needs to be debugged. Designers may include other stakeholders and communicate with others, but do they have enough agency to establish change?

Eventually, innovation may be understood as an organizational commitment. In this form, innovation is defined not only as improvements in price and performance, but also as the creation of new opportunities, disruption of existing systems, generator of public enthusiasm, and builder of social and business coalitions that, in the long run, may support fundamental changes [209]. Recent research on facilitators in healthcare confirmed its importance for improvement work, but also identified a not yet established role that needs to be further strengthened [210–212].

- Improve clinical logistics
- Facilitate long-term processes
- Develop flair for a broader learning space

6.2.4 Implications for designers and niche entrepreneurs

Design plays a major role in the investigation, process, and results of system innovation [213]. New design disciplines, such as systemic design, design for sustainability, and transition design, create a new mindset in designers. Sustainable development implies a reflection on one's behavior, which may lead to reflective practices and responsible innovations [190].

The analytical result of this thesis is based on various findings and insights. Paper 3 suggests improvements to enhance the current transport of laboratory samples. Furthermore, Paper 1 covered the study by Cawthorne et al. [214], who used an ethical framework to design a drone to transport blood samples. Similar studies could be conducted to evaluate public acceptance and ethical considerations. For example, Keitsch [169] concluded that transdisciplinary collaboration could compromise different roles, goals, and responsibilities and that ethics practice could contribute to more transdisciplinary collaboration.

Designers are also experts in the processing of interactions by applying frameworks, methods, and tools. This knowledge and skills may be essential to support and accelerate innovation [215].

- Evaluate public acceptance and ethical considerations
- Reflexive design of products and services

• Application and demonstration of innovation processes that can be used to illustrate and implement innovation processes in society

Overall findings of this thesis reveal that leadership, interdisciplinary work, and collaborative communication are essential factors in innovation processes and that the reconfiguration pathway presents an alternative to technological substitution [216]. System innovation includes many social groups, reconfiguration processes can be difficult to steer, and results can be open and uncertain [72]. However, the reconfiguration pathway is useful for actors in different societal subsystems. Recently, it has been suggested, for example, for policymakers to initiate sociotechnical transitions [208], which indicates that actively promoting niche innovations by identifying lock-ins and helping to remove them seems to be a reasonable task for different stakeholders and professionals working toward sustainable societies.

Chapter 7

Conclusions

This final chapter presents the main conclusion and identifies possibilities for further research.

The primary research question asks if and eventually how drones can become part of a future healthcare transport system. This was carried out by answering three sub-questions:

- 1. How can current healthcare transport systems be understood as socio-technical systems?
- 2. What attitudes do healthcare workers have towards drone transport in healthcare?
- 3. How can drones be implemented in a future healthcare transport system?

The main conclusions are listed below:

- 1. Socio-technical system theory and the multi-level perspective provide tools for understanding and facilitating innovation processes.
- 2. In a broader perspective, drones can support the aim of having a healthcare system with good working conditions because they can be used in cocreation sessions to shape how laboratories want to work.
- 3. Drones allow for developing scenarios that invite user participation.
- 4. Currently, drones for laboratory logistics are not economically sustainable in an urban environment, but they can advance the sustainable goal of good equipment for transporting laboratory samples.

7.1 State-of-the-art and future outlook

Establishing and maintaining a culture of innovation is hard work. Change agents are important intermediates who work with innovation on a day-to-day basis. The important message in this thesis to healthcare leaders is to strengthen those roles and support this work.

This thesis turned out to be largely theoretical because it was based to a high degree on literature studies, partly due to the conditions and developments de-

scribed in Section 2.2.4. Therefore, subsequent research efforts should investigate a more practical implementation, such as modeling sustainability transitions [217], designing routines with narrative networks [161, 218], or designing transition pathways [213]. According to Raghunatha et al. [219] especially early research within the theme of adoption is an important subject to address many of the uncertainties in the implementation of drones and system transitions.

The original intention for this thesis was to include a focus group interview. However, due to time limitations, only expert interviews were conducted. Future research efforts should validate the findings of this thesis by conducting focus group interviews. The study by Krigsholm et al. [74] may help plan such a study.

A recently published bibliometric analysis by Wang et al. shows that research efforts on the multi-level perspective are flourishing [196]. This is also the case for research on system innovation in healthcare. Johansen et al. explored transitions in Dutch healthcare [41], a recent study in Finland examined factors that enable innovation activities [25], and Marjanovic et al. [68] examined system innovation based on data from England.

Chapter 1 introduced system science and the holistic conceptualization of a solution space. Although this perspective can be beneficial, the focus on systems and structures may neglect what drives human actions and ignore that people have choices regarding their focus. For this reason, it is important to combine the multi-level perspective with other theories more human-centered. A promising approach combines the multi-level perspective with social approaches to plan and evaluate sustainable transitions [208, 220]. Future studies on drones in health-care systems should attempt to understand the factors that encourage or prevent such an implementation, thereby understanding the resilience of human-centered healthcare systems and the changes triggered by technological innovations.

The multi-level perspective transferred to design may be a canvas for understanding innovation over time and how transitions could be harnessed to address complex problems [77, 221]. For example, insights from past projects could inform future visions and interventions in the present. Additionally, the multi-level perspective can be applied to analyze complex systems to discover, visualize, and design sustainable solutions.

System innovation does not provide ready-to-use practical guidelines. Therefore, it may not be directly applicable to practitioners involved in innovation projects who do not have an academic background, as Raven et al. points out [59].

An opportunity provided by this PhD project was to learn about the emerging research fields of systems innovation and sustainable transitions. For example, Sustainable Transition Research Network (STRN) was founded in 2009 and has grown from 200 members in 2010 to 1700 members in 2018 [222]. Although design networks and interactions with other disciplines have contributed to these research areas, the contributions of research and practice need to increase [83, 167].

Bibliography

- [1] The Research Council of Norway, Research an implementation of unmanned aircraft systems for a fast and secure transport of biological material and blood products, Accessed: 18.10.2023, 2018. [Online]. Available: https: //prosjektbanken.forskningsradet.no/en/project/FORISS/282207? Kilde=FORISS&distribution=Ar&chart=bar&calcType=projects& Sprak=no&sortBy=date&sortOrder=desc&resultCount=30&offset= 0&Prosjektnummer=282207.
- Single European Sky A. T. M. Research 3 Joint Undertaking, "European drones outlook study : Unlocking the value for europe," Tech. Rep., 2017.
 [Online]. Available: https://data.europa.eu/doi/10.2829/085259.
- [3] M. Kunovjanek and C. Wankmüller, "Containing the covid-19 pandemic with drones - feasibility of a drone enabled back-up transport system," *Transp Policy (Oxf)*, vol. 106, pp. 141–152, 2021. DOI: 10.1016/j.tranp ol.2021.03.015.
- [4] Y. Shen, D. Guo, F. Long, L. A. Mateos, H. Ding, Z. Xiu, R. B. Hellman, A. King, S. Chen, C. Zhang, and H. Tan, "Robots under covid-19 pandemic: A comprehensive survey," *IEEE Access*, vol. 9, pp. 1590–1615, 2021. DOI: 10.1109/ACCESS.2020.3045792.
- [5] G. Dekoulis, Ed., *Drones*, Rijeka, 2018. DOI: 10.5772/intechopen.70910.
 [Online]. Available: https://doi.org/10.5772/intechopen.70910.
- [6] S. Khisa and S. Moh, "Medium access control protocols for the internet of things based on unmanned aerial vehicles: A comparative survey," *SENSORS-BASEL*, vol. 20, no. 19, pp. 1–40, 2020, ISSN: 1424-8220. DOI: 10.3390/s20195586.
- M. Poljak and A. Šterbenc, "Use of drones in clinical microbiology and infectious diseases: Current status, challenges and barriers," *CLIN MICRO-BIOL INFEC*, vol. 26, no. 4, pp. 425–430, 2020, ISSN: 1198-743X. DOI: 10.1016/j.cmi.2019.09.014.
- [8] B. Rao, A. G. Gopi, and R. Maione, "The societal impact of commercial drones," *TECHNOL SOC*, vol. 45, pp. 83–90, 2016, ISSN: 0160-791X. DOI: 10.1016/j.techsoc.2016.02.009.

- [9] M. Nentwich and D. M. Hórvath, "Delivery drones from a technology assessment perspective," Institute for Technology Assessment Vienna (ITA). http://epub. oeaw. ac. at/ita/ita-projektberichte/2018-01. pdf, 2018.
- [10] J. C. Rosser, V. Vignesh, B. A. Terwilliger, and B. C. Parker, "Surgical and medical applications of drones: A comprehensive review," *JSLS-J SOC LA-PAROEND*, vol. 22, no. 3, e2018.00018, 2018, ISSN: 1086-8089. DOI: 10. 4293/JSLS.2018.00018.
- [11] R. F. Greaves, S. Bernardini, M. Ferrari, P. Fortina, B. Gouget, D. Gruson, T. Lang, T. P. Loh, H. A. Morris, J. Y. Park, M. Roessler, P. Yin, and L. J. Kricka, "Key questions about the future of laboratory medicine in the next decade of the 21st century: A report from the ifcc-emerging technologies division," *CLIN CHIM ACTA*, vol. 495, pp. 570–589, 2019, ISSN: 0009-8981. DOI: 10.1016/j.cca.2019.05.021.
- [12] J. Braun, S. D. Gertz, A. Furer, T. Bader, H. Frenkel, J. Chen, E. Glassberg, and D. Nachman, "The promising future of drones in prehospital medical care and its application to battlefield medicine," *The journal of trauma and acute care surgery*, vol. 87, no. 1, S28–S34, 2019, ISSN: 2163-0755. DOI: 10.1097/TA.00000000002221.
- [13] S. Truog, L. Maxim, C. Matemba, C. Blauvelt, H. Ngwira, A. Makaya, S. Moreira, E. Lawrence, G. Ailstock, A. Weitz, M. West, and O. Defawe, "Insights before flights: How community perceptions can make or break medical drone deliveries," *DRONES-BASEL*, vol. 4, no. 3, pp. 1–14, 2020, ISSN: 2504-446X. DOI: 10.3390/drones4030051.
- [14] S. J. Kim, G. J. Lim, J. Cho, and M. J. Côté, "Drone-aided healthcare services for patients with chronic diseases in rural areas," *J INTELL ROBOT SYST*, vol. 88, no. 1, pp. 163–180, 2017, ISSN: 0921-0296. DOI: 10.1007/s10846-017-0548-z.
- [15] F. J. Mateen, K. H. B. Leung, A. C. Vogel, A. F. Cissé, and T. C. Y. Chan, "A drone delivery network for antiepileptic drugs: A framework and modelling case study in a low-income country," *T ROY SOC TROP MED H*, vol. 114, no. 4, pp. 308–314, 2020, ISSN: 0035-9203. DOI: 10.1093/ trstmh/trz131.
- [16] C. Kyrkou, S. Timotheou, P. Kolios, T. Theocharides, and C. Panayiotou, "Drones: Augmenting our quality of life," *MPOT*, vol. 38, no. 1, pp. 30– 36, 2019, ISSN: 0278-6648. DOI: 10.1109/MPOT.2018.2850386.
- J. E. Scott and C. H. Scott, "Drone delivery models for medical emergencies," in ser. Healthcare Delivery in the Information Age, Cham: Cham: Springer International Publishing, 2019, pp. 69–85. DOI: 10.1007/978-3-030-17347-0_3.
- [18] M. Balasingam, "Drones in medicine-the rise of the machines," International journal of clinical practice, vol. 71, no. 9, e12989-e12989, 2017, ISSN: 1368-5031.

- [19] I. Quintanilla García, N. Vera Vélez, P. Alcaraz Martínez, J. Vidal Ull, and B. Fernández Gallo, "A quickly deployed and uas-based logistics network for delivery of critical medical goods during healthcare system stress periods: A real use case in valencia (spain)," *Drones*, vol. 5, no. 1, pp. 13–13, Feb. 2021. DOI: 10.3390/drones5010013. [Online]. Available: https://www.mdpi.com/2504-446X/5/1/13.
- [20] K. A. Johannessen, A conceptual approach to time savings and cost competitiveness assessments for drone transport of biologic samples with unmanned aerial systems (drones), 2022. DOI: 10.3390/drones6030062.
- [21] E. Wartella, V. Rideout, H. Montague, L. Beaudoin-Ryan, and A. Lauricella, "Teens, health and technology: A national survey," *Media and communication (Lisboa)*, vol. 4, no. 3, pp. 13–23, 2016, ISSN: 2183-2439. DOI: 10.17645/mac.v4i3.515.
- [22] J. E. W. Broerse and J. Grin, *Toward sustainable transitions in healthcare systems*, New York, 2017.
- [23] A. Pereno and D. Eriksson, "A multi-stakeholder perspective on sustainable healthcare: From 2030 onwards," *Futures*, vol. 122, 2020. DOI: 10. 1016/j.futures.2020.102605.
- [24] C. M. Lopes, A. J. Scavarda, G. L. R. Vaccaro, C. R. Pohlmann, and A. L. Korzenowski, "Perspective of business models and innovation for sustainability transition in hospitals," *Sustainability (Basel, Switzerland)*, vol. 11, no. 1, pp. 5–5, 2019. DOI: 10.3390/su11010005.
- [25] P. Hyrkäs, N. Lunkka, J. Reponen, and M. Suhonen, "Factors enabling innovation activities: Combining perspectives of personnel and management of a public university hospital," *International Journal of Innovation and Learning*, vol. 31, no. 1, pp. 70–90, Dec. 2021, ISSN: 1471-8197. DOI: 10.1504/IJIL.2022.119637. [Online]. Available: https://www. inderscienceonline.com/doi/abs/10.1504/IJIL.2022.119637.
- [26] F. Matthias, "Fit for the future? a new approach in the debate about what makes healthcare systems really sustainable," *Sustainability*, vol. 7, no. 1, pp. 294–312, 2014. DOI: 10.3390/su7010294.
- [27] A. Klitkou, S. Bolwig, T. Hansen, and N. Wessberg, "The role of lock-in mechanisms in transition processes: The case of energy for road transport," *Environmental Innovation and Societal Transitions*, vol. 16, pp. 22– 37, 2015. DOI: 10.1016/j.eist.2015.07.005.
- [28] L. J. Leifer and M. Steinert, "Dancing with ambiguity: Causality behavior, design thinking, and triple-loop-learning," *Information, knowledge, systems management*, vol. 10, no. 1-4, pp. 151–173, 2011, ISSN: 1389-1995. DOI: 10.3233/IKS-2012-0191.
- [29] A. İ. Gaziulusoy, "System Innovation for Sustainability," Ph.D. dissertation, 2010.

- [30] R. Kemp, D. Loorbach, and J. Rotmans, "Transition management as a model for managing processes of co-evolution towards sustainable development," *International journal of sustainable development and world ecol*ogy, vol. 14, no. 1, pp. 78–91, 2007. DOI: 10.1080/13504500709469709.
- [31] J. Rotmans, R. Kemp, and M. van Asselt, "More Evolution than Revolution : Transition Management in Public Policy," *Foresight (London): the journal* of insurance and risk management, 2001.
- [32] P. L. Mabry, D. H. Olster, G. D. Morgan, and D. B. Abrams, "Interdisciplinarity and systems science to improve population health. a view from the nih office of behavioral and social sciences research," *Am J Prev Med*, vol. 35, no. 2, S211–S224, 2008. DOI: 10.1016/j.amepre.2008.05.018.
- [33] D. H. Meadows, *Thinking in systems : a primer*, D. Wright, Ed. White River Junction, Vermont: Chelsea Green Publishing, 2008.
- [34] G. Ropohl, "Philosophy of socio-technical systems," Society for philosophy & technology quarterly electronic journal, vol. 4, no. 3, pp. 186–194, 1999, ISSN: 1091-8264. DOI: 10.5840/techne19994311.
- [35] L. G. Tornatzky, *The processes of technological innovation* (Issues in organization and management series), M. Fleischer and A. K. Chakrabarti, Eds. Lexington, Mass: Lexington Books, 1990.
- [36] R. Ison, "Systems thinking and practice for action research," *The Sage handbook of action research participative inquiry and practice*, vol. 2, pp. 139–158, 2008.
- [37] D. Gruson, "New solutions for the sample transport and results delivery: A digital lab," *Electronic Journal of the International Federation of Clinical Chemistry and Laboratory Medicine*, vol. 29, no. 3, pp. 210–214, 2018. [Online]. Available: https://pubmed.ncbi.nlm.nih.gov/30479606/.
- [38] L. E. Ferreras, Itrans: Proactive its based on drone technology to solve urban transportation challenge, Cham, 2017. DOI: 10.1007/978-3-319-51602-8_19.
- [39] F.-X. Meunier and R. Bellais, "Technical systems and cross-sector knowledge diffusion: An illustration with drones," *Technology Analysis & Strategic Management*, vol. 31, no. 4, pp. 433–446, Apr. 2019. DOI: 10.1080/ 09537325.2018.1518522. [Online]. Available: https://www.tandfonlin e.com/doi/full/10.1080/09537325.2018.1518522.
- [40] C. J. L. Murray and D. Evans, Health Systems Performance Assessment: Debates, Methods and Empiricism. Albany: Albany: World Health Organization, 2003, pp. xxxii–xxxii.
- [41] F. Johansen, D. Loorbach, and A. Stoopendaal, "Exploring a transition in dutch healthcare," *J Health Organ Manag*, vol. 32, no. 7, pp. 875–890, 2018. DOI: 10.1108/JH0M-07-2018-0185.

- [42] South-Eastern Norway Regional Health Authority, Regional utviklingsplan 2035, Accessed: 09.03.2022, Nov. 2021. [Online]. Available: https:// helse-sorost.no/regional-utviklingsplan-2035%5C#nye-arbeidsfo rmer-og-bedre-bruk-av-teknologi.
- [43] A. H. Van de Ven, "Central problems in the management of innovation," *Management science*, Management Science, vol. 32, no. 5, pp. 590–607, 1986, ISSN: 0025-1909. DOI: 10.1287/mnsc.32.5.590.
- [44] S. L. Vargo, M. A. Akaka, and H. Wieland, "Rethinking the process of diffusion in innovation: A service-ecosystems and institutional perspective," *Journal of business research*, vol. 116, pp. 526–534, 2020. DOI: 10.1016/ j.jbusres.2020.01.038.
- [45] J. A. Colyvas and S. Jonsson, "Ubiquity and legitimacy: Disentangling diffusion and institutionalization," *Sociol theor*, vol. 29, no. 1, pp. 27–53, 2011, ISSN: 1467-9558. DOI: 10.1111/j.1467-9558.2010.01386.x.
- [46] R. Maity, R. Mishra, and P. K. Pattnaik, "A review of flying robot applications in healthcare," in *Smart Healthcare Analytics: State of the Art*, P. K. Pattnaik, A. Vaidya, S. Mohanty, S. Mohanty, and A. Hol, Eds., Singapore: Springer Singapore, 2022, pp. 103–111. DOI: 10.1007/978-981-16-5304-9_8. [Online]. Available: https://doi.org/10.1007/978-981-16-5304-9_8.
- [47] F. Stephan, N. Reinsperger, M. Grünthal, D. Paulicke, and P. Jahn, "Human drone interaction in delivery of medical supplies: A scoping review of experimental studies," *PLOS ONE*, vol. 17, no. 4, e0267664, Apr. 2022. DOI: 10.1371/journal.pone.0267664. [Online]. Available: https://doi.org/10.1371/journal.pone.0267664.
- [48] A. Smith, J. E. Dickinson, G. Marsden, T. Cherrett, A. Oakey, and M. Grote, "Public acceptance of the use of drones for logistics: The state of play and moving towards more informed debate," *Technology in Society*, vol. 68, p. 101 883, 2022, ISSN: 0160-791X. DOI: 10.1016/j.techsoc.2022. 101883. [Online]. Available: https://www.sciencedirect.com/science /article/pii/S0160791X22000240.
- [49] P. Mora and C. A. S. Araujo, "Delivering blood components through drones: A lean approach to the blood supply chain," *Supply Chain Forum: An International Journal*, vol. 23, no. 2, pp. 113–123, Apr. 2022, ISSN: 1625-8312.
 DOI: 10.1080/16258312.2021.1984167. [Online]. Available: https://doi.org/10.1080/16258312.2021.1984167.
- [50] G. Baxter and I. Sommerville, "Socio-technical systems: From design methods to systems engineering," *Interacting with computers*, vol. 23, no. 1, pp. 4–17, 2011. DOI: 10.1016/j.intcom.2010.07.003.
- [51] F. W. Geels, Technological transitions and system innovations: a co-evolutionary and socio-technical analysis. Edward Elgar Publishing, 2005, ISBN: 1845-42459X.

- [52] J. Markard, R. Raven, and B. Truffer, "Sustainability transitions: An emerging field of research and its prospects," *Res policy*, vol. 41, no. 6, pp. 955– 967, 2012, ISSN: 0048-7333. DOI: 10.1016/j.respol.2012.02.013.
- [53] F. W. Geels, "Technological transitions as evolutionary reconfiguration processes: A multi-level perspective and a case-study," *Research Policy*, vol. 31, no. 8, pp. 1257–1274, 2002. DOI: 10.1016/S0048-7333(02)00062-8. [Online]. Available: http://www.sciencedirect.com/science/article /pii/S0048733302000628.
- [54] F. W. Geels, "Processes and patterns in transitions and system innovations: Refining the co-evolutionary multi-level perspective," *TECHNOL FORE-CAST SOC*, vol. 72, no. 6, pp. 681–696, 2005, ISSN: 0040-1625. DOI: 10.1016/j.techfore.2004.08.014.
- [55] F. W. Geels and J. Schot, "Typology of sociotechnical transition pathways," *Research Policy*, vol. 36, no. 3, pp. 399–417, 2007, ISSN: 0048-7333. DOI: 10.1016/j.respol.2007.01.003.
- [56] L. Fuenfschilling and B. Truffer, "The structuration of socio-technical regimesconceptual foundations from institutional theory," *Research Policy*, vol. 43, no. 4, pp. 772–791, 2014. DOI: 10.1016/j.respol.2013.10.010. [Online]. Available: http://www.sciencedirect.com/science/article/ pii/S004873313001893.
- [57] F. W. Geels, "Socio-technical transitions to sustainability: A review of criticisms and elaborations of the multi-level perspective," *Current Opinion in Environmental Sustainability*, vol. 39, pp. 187–201, 2019. DOI: 10.1016/j.cosust.2019.06.009.
- [58] J. Braithwaite, K. Churruca, J. C. Long, L. A. Ellis, and J. Herkes, "When complexity science meets implementation science: A theoretical and empirical analysis of systems change," *BMC Medicine*, vol. 16, no. 1, pp. 63– 63, Apr. 2018. DOI: 10.1186/s12916 - 018 - 1057 - z. [Online]. Available: https://bmcmedicine.biomedcentral.com/articles/10.1186/ s12916-018-1057-z.
- [59] R. Raven, S. Bosch, van den, and R. Weterings, "Transitions and strategic niche management : Towards a competence kit for practitioners," *International Journal of Technology Management*, vol. 51, no. 1, pp. 57–74, 2010, ISSN: 0267-5730. DOI: 10.1504/IJTM.2010.033128.
- [60] E. Heiskanen, S. Hyysalo, M. Jalas, J. Juntunen, and R. Lovio, "User involvement and radical innovation: the case of heat pumps in Finland," in *The highways and byways to radical innovation : design perspectives*, P. Rind Christensen and S. Junginger, Eds. Denmark: DESIGNSKOLEN KOLDING, 2014, pp. 171–196, ISBN: 978-87-90775-74-2.

- [61] D. Loorbach, N. Frantzeskaki, and F. Avelino, "Sustainability Transitions Research: Transforming Science and Practice for Societal Change," *Annual review of environment and resources*, vol. 42, no. 1, pp. 599–626, 2017, ISSN: 1543-5938. DOI: 10.1146/annurev-environ-102014-021340.
- [62] F. W. Geels, "Major system change through stepwise reconfiguration: A multi-level analysis of the transformation of American factory production (1850–1930)," *Technology in Society*, vol. 28, no. 4, pp. 445–476, Nov. 2006, ISSN: 0160791X. DOI: 10.1016/j.techsoc.2006.09.006. [Online]. Available: https://linkinghub.elsevier.com/retrieve/pii/ S0160791X06000376.
- [63] J. Leväsluoto, H. Määttä, K. Hyytinen, J. Kohl, and M. Toivonen, "Challenges of customer oriented health care service models: Perspectives from the renewal of a primary health care and integrated care system," in *Proceedings of 24th Annual RESER Conference 2014*, 24th Annual RESER Conference, RESER 2014 : Services and New Societal Challenges: Innovation for Sustainable Growth and Welfare, RESER 2014 ; Conference date: 11-09-2014 Through 13-09-2014, Finland: VTT Technical Research Centre of Finland, 2014, pp. 903–914.
- [64] F. W. Geels and R. Kemp, "Dynamics in socio-technical systems: Typology of change processes and contrasting case studies," *Technology in Society*, vol. 29, no. 4, pp. 441–455, Nov. 2007, ISSN: 0160791X. DOI: 10.1016/ j.techsoc.2007.08.009.
- [65] F. Geels, "Multi-level perspective on system innovation: Relevance for industrial transformation," in Feb. 2006, pp. 163–186. DOI: 10.1007/1-4020-4418-6_9.
- [66] F. W. Geels, "The multi-level perspective on sustainability transitions: Responses to seven criticisms," *Environmental Innovation and Societal Transitions*, vol. 1, no. 1, pp. 24–40, 2011, ISSN: 2210-4224. DOI: 10.1016/j. eist.2011.02.002.
- [67] F. Geels, "A socio-technical analysis of low-carbon transitions: Introducing the multi-level perspective into transport studies," *Journal of Transport Geography*, vol. 24, pp. 471–482, Sep. 2012. DOI: 10.1016/j.jtrangeo. 2012.01.021.
- [68] S. Marjanovic, M. Altenhofer, L. Hocking, J. Chataway, and T. Ling, "Innovating for improved healthcare: Sociotechnical and innovation systems perspectives and lessons from the nhs," *Science & public policy*, vol. 47, no. 2, pp. 283–297, 2020. DOI: 10.1093/scipol/scaa005.
- [69] A. Smith, J.-P. Voß, and J. Grin, "Innovation studies and sustainability transitions: The allure of the multi-level perspective and its challenges," *Research Policy*, vol. 39, no. 4, pp. 435–448, 2010. DOI: 10.1016/j.resp ol.2010.01.023. [Online]. Available: http://www.sciencedirect.com/ science/article/pii/S0048733310000375.

- [70] R. Kemp and J. Rotmans, *The Management of the Co-evolution of Technical, Environmental and Social Systems*. 2005, pp. 33–55.
- [71] H. N. Opoku and M. M. Keitsch, "An objective approach to sustainability? theory of science and political implications of industrial ecology," EN, *Ecologie & politique*, vol. 32, no. 1, pp. 141–152, 2006, ISSN: 9782849500844. DOI: 10.3917/ecopo.032.0141. [Online]. Available: https://www.cair n-int.info/journal-ecologie-et-politique-sciences-culturessocietes-2006-1-page-141.htm.
- [72] F. W. Geels, A. McMeekin, J. Mylan, and D. Southerton, "A critical appraisal of sustainable consumption and production research: The reformist, revolutionary and reconfiguration positions," *Global environmental change*, vol. 34, pp. 1–12, 2015. DOI: 10.1016/j.gloenvcha.2015.04.013.
- [73] C. W. Clegg, "Sociotechnical principles for system design," *Applied ergonomics*, vol. 31, no. 5, pp. 463–477, 2000, ISSN: 0003-6870. DOI: 10. 1016/S0003-6870(00)00009-0.
- [74] P. Krigsholm, K. Riekkinen, and P. Ståhle, "Pathways for a future cadastral system: A socio-technical approach," *Land use policy*, vol. 94, p. 104504, 2020, ISSN: 0264-8377. DOI: 10.1016/j.landusepol.2020.104504.
- [75] B. Hiebert, E. Nouvet, V. Jeyabalan, and L. Donelle, "The application of drones in healthcare and health-related services in north america: A scoping review," *Drones (Basel)*, vol. 4, no. 3, p. 30, 2020, ISSN: 2504-446X. DOI: 10.3390/drones4030030.
- [76] P. H. Jones, *Design for care : innovating healthcare experience*. New York, N.Y: Rosenfeld, 2013.
- [77] T. Irwin, *The Emerging Transition Design Approach*. London: Bloomsbury Visual Arts, 2019.
- [78] A. Pereno and S. Barbero, "Systemic design for territorial enhancement: An overview on design tools supporting socio-technical system innovation," *Strategic Design Research Journal*, vol. 13, no. 2, pp. 113–136, 2020.
 DOI: 10.4013/SDRJ.2020.132.02.
- [79] B. Sevaldson and B. Sevaldson, *Designing complexity : the methodology and practice of systems oriented design*. Champaign, IL: Common Ground Research Networks, 2021.
- [80] B. Sevaldson, "Giga-mapping: Visualisation for complexity and systems thinking in design," *Nordes*, no. 4, 2011, ISSN: 1604-9705.
- [81] P. Jones, "Systemic design: Design for complex, social, and sociotechnical systems," in *Handbook of Systems Sciences*, G. S. Metcalf, K. Kijima, and H. Deguchi, Eds., Singapore: Springer Singapore, 2021, pp. 787–811. DOI: 10.1007/978-981-15-0720-5_60. [Online]. Available: https://doi.org/10.1007/978-981-15-0720-5_60.

- [82] P. A. Rodgers and J. Yee, *The Routledge companion to design research* (Routledge companions). London: Routledge, 2015.
- [83] T. Irwin, C. Tonkinwise, and G. Kossoff, "Transition design: An educational framework for advancing the study and design of sustainable transitions," *Cuadernos del Centro de Estudios en Diseño y Comunicación. En*sayos, no. 105, pp. 31–72, 2022, ISSN: 1853-3523. DOI: 10.18682/cdc. vi105.4188.
- [84] A. Agrawal, S. Abraham, B. Burger, C. Christine, L. Fraser, J. Hoeksema, S. Hwang, E. Travnik, S. Kumar, W. Scheirer, J. Cleland-Huang, M. Vierhauser, R. Bauer, and S. Cox, "The next generation of human-drone partnerships: Co-designing an emergency response system," ser. CHI '20, Ithaca: Ithaca: ACM, 2020, pp. 1–13. DOI: 10.1145/3313831.3376825.
- [85] H. E. Comtet and K.-A. Johannessen, "A socio-analytical approach to the integration of drones into health care systems," *Information (Basel)*, vol. 13, no. 62, p. 62, 2022. DOI: 10.3390/info13020062.
- [86] H. E. Comtet and K.-A. Johannessen, "The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems," eng, *International Journal of Environmental Research and Public Health*, vol. 18, no. 5, pp. 1–15, 2021, ISSN: 1661-7827.
- [87] H. E. Comtet, M. Keitsch, and K.-A. Johannessen, "Realities of using drones to transport laboratory samples: Insights from attended routes in a mixedmethods study.," eng, *Journal of multidisciplinary healthcare*, vol. 15, pp. 1871– 1885, 2022.
- [88] L. Cohen, K. Morrison, and L. Manion, *Research methods in education*, Eighth edition. New York: Routledge, 2018.
- [89] J. Scotland, "Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms," *English language teaching (Toronto)*, vol. 5, no. 9, pp. 9–16, 2012, ISSN: 1916-4750. DOI: 10.5539/elt.v5n9p9.
- [90] D. L. Morgan, "Paradigms lost and pragmatism regained: Methodological implications of combining qualitative and quantitative methods," *J MIX METHOD RES*, vol. 1, no. 1, pp. 48–76, 2007, ISSN: 1558-6898. DOI: 10. 1177/2345678906292462.
- [91] J. W. Creswell and V. L. Plano Clark, *Designing and conducting mixed methods research*, International student edition. Los Angeles: Sage, 2018.
- [92] M. Crotty, *The foundations of social research : meaning and perspective in the research process*. London: Sage, 1998.
- [93] V. Kaushik and C. A. Walsh, *Pragmatism as a research paradigm and its implications for social work research*, 2019. DOI: 10.3390/socsci8090255.

- [94] V. K. Vaishnavi and W. Kuechler, Design science research methods and patterns: innovating information and communication technology, Second edition. Boca Raton, FL: Boca Raton, FL: CRC Press, 2015.
- [95] P. Checkland, "From optimizing to learning: A development of systems thinking for the 1990s," *The Journal of the Operational Research Society*, vol. 36, no. 9, pp. 757–767, 1985, ISSN: 0160-5682. DOI: 10.1057/jors. 1985.141.
- [96] H. A. Simon, *The sciences of the artificial*, 3rd ed. Cambridge, Mass: MIT Press, 1996.
- [97] N. Cross, Designerly ways of knowing, London, 2006.
- [98] D. A. Schön, *The reflective practitioner : how professionals think in action*. Aldershot: Avebury, 1991.
- [99] J. Auernhammer, "Design research in innovation management: A pragmatic and human-centered approach," *R&d manage*, vol. 50, no. 3, pp. 412– 428, 2020, ISSN: 0033-6807. DOI: 10.1111/radm.12409.
- [100] C. Argyris and D. A. Schön, Organizational learning II : theory, method, and practice (Addison-Wesley series on organizational development). Reading, Mass: Addison-Wesley, 1996.
- [101] J. Dewey, *The quest for certainty : a study of the relation of knowledge and action*. London: George Allen & Unwin, 1930.
- [102] J. Dewey, *Art as experience* (A Perigee book). New York: Berkley Publishing Group, 2005.
- [103] W. James, The meaning of truth : a sequel to "Pragmatism". New York, 1909.
- [104] B. Elzen, B. van Mierlo, and C. Leeuwis, "Anchoring of innovations: Assessing dutch efforts to harvest energy from glasshouses," *Environmental Innovation and Societal Transitions*, vol. 5, pp. 1–18, 2012. DOI: 10.1016/j.eist.2012.10.006. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S2210422412000536.
- [105] F. W. Geels, "Ontologies, socio-technical transitions (to sustainability), and the multi-level perspective," *RES POLICY*, Research Policy, vol. 39, no. 4, pp. 495–510, 2010, ISSN: 0048-7333. DOI: 10.1016/j.respol.2010.01. 022.
- [106] P. Morlacchi and B. R. Martin, "Emerging challenges for science, technology and innovation policy research: A reflexive overview," *Research Policy*, vol. 38, no. 4, pp. 571–582, 2009, ISSN: 0048-7333. DOI: 10.1016/j.re spol.2009.01.021. [Online]. Available: https://www.sciencedirect. com/science/article/pii/S0048733309000146.
- [107] J. Vink, "In/visible Conceptualizing Service Ecosystem Design," Ph.D. dissertation, 2019.

- [108] W. R. Scott, *Institutions and organizations : ideas, interests, and identities*, 4th ed. Thousand Oaks, Calif: Sage, 2014, ISBN: 978-1-4522-4222-4.
- [109] A. Broom, "Ethical issues in social research," Complementary Therapies in Medicine, vol. 14, no. 2, pp. 151–156, 2006, ISSN: 0965-2299. DOI: 10.1016/j.ctim.2005.11.002. [Online]. Available: https://www. sciencedirect.com/science/article/pii/S0965229905001445.
- [110] D. Benyon, *Designing interactive systems : a comprehensive guide to HCI, UX and interaction design*, 3rd ed. Harlow: Pearson, 2014.
- [111] K.-A. Johannessen, H. Comtet, and E. Fosse, "A drone logistic model for transporting the complete analytic volume of a large-scale university laboratory," eng, *International Journal of Environmental Research and Public Health*, vol. 18, no. 9, p. 4580, 2021, ISSN: 1661-7827.
- [112] H. Lundgaard, Blodig droneferd mellom rikshospitalet og ullevål, https: //www.aftenposten.no/oslo/i/PRnGw6/blodig-droneferd-mellom-ri kshospitalet-og-ullevaal, Accessed: 06.06.2023, Mar. 2021. [Online]. Available: https://www.aftenposten.no/oslo/i/PRnGw6/blodigdroneferd-mellom-rikshospitalet-og-ullevaal.
- [113] K. Herr and G. L. Anderson, *The action research dissertation : A guide for students and faculty*, London, 2005.
- [114] R. K. Yin and R. K. Yin, *Case study research and applications : design and methods*, Sixth edition. Los Angeles, California: SAGE, 2018.
- [115] Norwegian register for scientific journals, series and publishers, Sustainability, Accessed: 07.04.2022, 2022. [Online]. Available: https://kan alregister.hkdir.no/publiseringskanaler/KanalTidsskriftInfo. action?id=475332&request_locale=en.
- [116] C. Petrou, Guest post mdpi's remarkable growth, Accessed: 07.04.2022, Aug. 2020. [Online]. Available: https://scholarlykitchen.sspnet. org/2020/08/10/guest-post-mdpis-remarkable-growth/.
- [117] N. F. Kock Jr, "Myths in organisational action research: Reflections on a study of computer-supported process redesign groups," *Organizações & Sociedade*, vol. 4, no. 9, pp. 65–91, 1997, ISSN: 1984-9230. DOI: 10.1590/ S1984-92301997000300003.
- [118] W. M. K. Trochim, K. Arora, and J. P. Donnelly, *Research methods : the essential knowledge base*, 2nd ed. Boston: Cengage Learning, 2016.
- [119] J. A. Maxwell, Qualitative research design : an interactive approach (Applied social research methods series), 2nd ed. Thousand Oaks, Calif: Sage Publications, 2005, vol. vol. 41, ISBN: 0761926070.
- [120] O. O. Aalen and A. Frigessi, *Statistiske metoder i medisin og helsefag*, 2. utg. Oslo: Gyldendal akademisk, 2018, ISBN: 978-82-05-50764-7.

- [121] S. Vaughn, J. S. Schumm, and J. Sinagub, Focus group interviews in education and psychology. Thousand Oaks u.a: Thousand Oaks u.a: SAGE Publ, 1996, ISBN: 0803958927.
- [122] A. Tashakkori and C. Teddlie, *Sage handbook of mixed methods in social & behavioral research*, 2nd ed. Thousand Oaks, Calif: SAGE, 2010.
- P. Mayring, "Qualitative content analysis: Demarcation, varieties, developments," *Forum, qualitative social research*, vol. 20, no. 3, 2019. DOI: 10.17169/fqs-20.3.3343.
- [124] U. Kelle, ""emergence" vs. "forcing" of empirical data? a crucial problem of "grounded theory" reconsidered (reprinted from fqs-forum qualitative sozialforschung, vol 6)," *HIST SOC RES*, vol. 19, no. supp 19, pp. 133– 156, 2007, ISSN: 0172-6404.
- [125] D. J. Greenwood and M. Levin, Introduction to action research : social research for social change, 2nd ed. Thousand Oaks, Calif: Sage Publications, 2007.
- [126] J. Lazar, Research methods in human-computer interaction, 2nd ed., J. H. Feng and H. Hochheiser, Eds. Cambridge, MA: Morgan Kaufmann, 2017, ISBN: 978-0-12-805390-4.
- [127] R. L. Gold, "Roles in sociological field observations," *Social forces*, vol. 36, no. 3, pp. 217–223, 1958, ISSN: 0037-7732. DOI: 10.2307/2573808.
- [128] R. Armstrong, B. J. Hall, J. Doyle, and E. Waters, "scoping the scope' of a cochrane review," *J PUBLIC HEALTH-UK*, vol. 33, no. 1, pp. 147–150, 2011, ISSN: 1741-3842. DOI: 10.1093/pubmed/fdr015.
- [129] Z. Munn, M. D. J. Peters, C. Stern, C. Tufanaru, A. McArthur, and E. Aromataris, "Systematic review or scoping review? guidance for authors when choosing between a systematic or scoping review approach," *BMC Med Res Methodol*, vol. 18, no. 1, pp. 143–143, 2018. DOI: 10.1186/s12874-018-0611-x.
- [130] N. J. Salkind, *Encyclopedia of research design*, London, 2010.
- [131] K. L. Wenjing Lyu Neil C. Thompson, "Sources of innovation revisited: Configurations of firm usage of internal and external search and the implications for innovation and market value," 2023.
- [132] D. A. Dillman, J. D. Smyth, and L. M. Christian, "Internet, phone, mail and mixed-mode surveys: The tailored design method," *Revista española de investigaciones sociológicas*, vol. 1, no. 154, pp. 161–165, 2016, ISSN: 0210-5233.
- [133] J. Lazar, *Research methods in human-computer interaction*, J. H. Feng and H. Hochheiser, Eds. Chichester: John Wiley, 2010, ISBN: 9780470723371.
- [134] F. J. Fowler, *Survey research methods* (Applied social research methods series), 4th ed. Los Angeles: Sage, 2009, vol. 1.

- [135] R. Tourangeau, L. J. Rips, and K. Rasinski, *The psychology of survey response*. Cambridge: Cambridge University Press, 2000.
- [136] N. M. Bradburn, S. Sudman, and B. Wansink, Asking questions : the definitive guide to questionnaire design - for market research, political polls, and social and health questionnaires, Revised edition. San Francisco, Calif: Jossey-Bass, 2004.
- [137] S. Brinkmann, InterViews : learning the craft of qualitative research interviewing, 3rd ed., S. Kvale, Ed. Thousand Oaks, Calif: Sage, 2015, ISBN: 978-1-4522-7572-7.
- [138] R. Barbour, *Doing focus groups* (The Sage qualitative research kit), U. Flick, Ed. London: SAGE, 2007, ISBN: 9780761949787.
- [139] D. W. Stewart, Focus groups : theory and practice (Applied social research methods series), 3rd ed., P. N. Shamdasani, Ed. Los Angeles, Calif: SAGE, 2015, vol. 20, ISBN: 9781452270982.
- K. Haula and E. Agbozo, "A systematic review on unmanned aerial vehicles in sub-saharan africa: A socio-technical perspective," *Technology in Society*, vol. 63, p. 101 357, Nov. 2020. DOI: 10.1016/j.techsoc.2020. 101357.
- [141] F. Geels and R. Raven, "Non-linearity and expectations in niche-development trajectories: Ups and downs in dutch biogas development (1973-2003)," *Technology Analysis & Strategic Management*, vol. 18, no. 3-4, pp. 375–392, Jul. 2006, ISSN: 0953-7325. DOI: 10.1080/09537320600777143. [On-line]. Available: https://doi.org/10.1080/09537320600777143.
- [142] D. Ó Tuama, "Ripples through the city: Understanding the processes set in motion through embedding a public bike sharing scheme in a city," *Research in Transportation Business and Management*, vol. 15, pp. 15–27, 2015. DOI: 10.1016/j.rtbm.2015.03.002.
- [143] M. Krey and R. Seiler, "Usage and acceptance of drone technology in healthcare: exploring patients and physicians perspective," University of Hawai'i at Manoa, 2019, pp. 4135–4144.
- [144] A. M. Knoblauch, S. de la Rosa, J. Sherman, C. Blauvelt, C. Matemba, L. Maxim, O. D. Defawe, A. Gueye, J. Robertson, J. McKinney, J. Brew, E. Paz, P. M. Small, M. Tanner, N. Rakotosamimanana, and S. G. Lapierre, "Bi-directional drones to strengthen healthcare provision: Experiences and lessons from madagascar, malawi and senegal," *BMJ Global Health*, vol. 4, no. 4, e001541, Jul. 2019. DOI: 10.1136/bmjgh-2019-001541.
- K. Laksham, "Unmanned aerial vehicle (drones) in public health: A SWOT analysis," *Journal of Family Medicine and Primary Care*, vol. 8, no. 2, p. 342, 2019. DOI: 10.4103/jfmpc.jfmpc 413 18.

- [146] F. U. Mion, Flying drones to exchange lab samples service innovation by the Swiss multisite hospital EOC. 2019. DOI: 10.1007/978-3-030-00749-2_26.
- [147] R. Kellermann, T. Biehle, and L. Fischer, "Drones for parcel and passenger transportation: A literature review," *Transportation Research Interdisciplinary Perspectives*, vol. 4, p. 100 088, Mar. 2020. DOI: 10.1016/j.trip. 2019.100088.
- M. Berg, "Patient care information systems and health care work: A sociotechnical approach," *International Journal of Medical Informatics*, vol. 55, no. 2, pp. 87–101, Aug. 1999, ISSN: 13865056. DOI: 10.1016/S1386 -5056(99)00011-8.
- [149] C. Clausen and W. Gunn, "Staging participatory innovation as transition design," in *Staging Collaborative Design and Innovation*, Edward Elgar Publishing, Dec. 2020. DOI: 10.4337/9781839103438.00017.
- [150] L. R. Burns, E. H. Bradley, and B. J. Weiner, Shortell and Kaluzny's health care management : organization, design, and behavior, Seventh ed. Boston, MA: Cengage, 2020.
- [151] E. M. Rogers, *Diffusion of innovations*, 5th ed., E. M. Rogers, Ed. New York: Free Press, 2003.
- [152] M. Wiedemann, A. Vij, and R. Banerjee, "Validating the benefits of increased drone uptake for australia: Geographic, demographic and social insights," 2023.
- [153] R. van Brakel, O. Kudina, C. Fonio, and K. Boersma, "Bridging values: Finding a balance between privacy and control. the case of corona apps in belgium and the netherlands," *J Contingencies and Crisis Management*, vol. 30, no. 1, pp. 50–58, Mar. 2022, ISSN: 0966-0879. DOI: 10.1111/ 1468 - 5973.12395. [Online]. Available: https://doi.org/10.1111/ 1468-5973.12395.
- J. W. Alba and J. W. Hutchinson, "Dimensions of consumer expertise," *The Journal of consumer research*, vol. 13, no. 4, pp. 411–454, 1987. DOI: 10.1086/209080.
- [155] D. Rosenbloom, H. Berton, and J. Meadowcroft, "Framing the sun: A discursive approach to understanding multi-dimensional interactions within socio-technical transitions through the case of solar electricity in ontario, canada," *Research Policy*, vol. 45, no. 6, pp. 1275–1290, 2016. DOI: 10. 1016/j.respol.2016.03.012.
- [156] M. Eichleay, E. Evens, K. Stankevitz, and C. Parker, "Using the unmanned aerial vehicle delivery decision tool to consider transporting medical supplies via drone," *Glob health-sci prac*, vol. 7, no. 4, pp. 500–506, 2019, ISSN: 2169-575X. DOI: 10.9745/GHSP-D-19-00119.

- [157] G. Seyfang and A. Smith, "Grassroots innovations for sustainable development: Towards a new research and policy agenda," *Environmental Politics*, vol. 16, no. 4, pp. 584–603, 2007, ISSN: 0964-4016. DOI: 10.1080/ 09644010701419121.
- [158] A. Smith, "Translating sustainabilities between green niches and sociotechnical regimes," *Technology Analysis & Strategic Management*, vol. 19, no. 4, pp. 427–450, 2007, ISSN: 0953-7325. DOI: 10.1080/0\-9\-5\-3\-7\-3\-2\-0\-7\-0\-1\-4\-0\-3\-3\-3\-4.
- [159] F. Boons, A. McMeekin, and P. Edward Elgar, *Handbook of sustainable in*novation. Cheltenham, UK: Edward Elgar Publishing, 2019.
- [160] T. Greenhalgh, "Role of routines in collaborative work in healthcare organisations," *BMJ*, vol. 337, no. 7681, pp. 196–1271, 2008. DOI: 10.1136/ bmj.a2448.
- [161] J. M. Goh, G. (Gao, and R. Agarwal, "Evolving work routines: Adaptive routinization of information technology in healthcare," *Information Systems Research*, vol. 22, no. 3, pp. 565–585, Jun. 2011. DOI: 10.1287/ isre.1110.0365. [Online]. Available: https://pubsonline.informs. org/doi/abs/10.1287/isre.1110.0365.
- [162] M. Rydland, Middle managers' role as change agents variations in approaches and their contributions to the progress of change, 2018.
- [163] G. Stoker, "Public value management: A new narrative for networked governance?" American review of public administration, vol. 36, no. 1, pp. 41– 57, 2006. DOI: 10.1177/0275074005282583.
- [164] E. Manzini, Design, when everybody designs : An introduction to design for social innovation. R. Coad, K. Friedman, and E. Stolterman, Eds., Cambridge, 2015.
- [165] R. Sosa and J. S. Gero, "Social structures that promote change in a complex world: The complementary roles of strangers and acquaintances in innovation," *FUTURES*, vol. 40, no. 6, pp. 577–585, 2008, ISSN: 0016-3287. DOI: 10.1016/j.futures.2007.11.006.
- [166] H. G. Nelson and E. Stolterman, *The Design Way: Intentional Change in an Unpredictable World*, Second Edition. MIT Press, 2012. DOI: 10.7551/ mitpress/9188.001.0001.
- [167] Ï. Gaziulusoy and E. E. Öztekin, "Design for sustainability transitions: Origins, attitudes and future directions," *Sustainability (Basel, Switzer-land)*, vol. 11, no. 13, p. 3601, 2019, ISSN: 2071-1050. DOI: 10.3390/ su11133601.

- [168] F. Coops, C. Hummels, K. Dorst, B. Goldstein, D. Loorbach, and İ. Gaziulusoy, "Designing for transitions and transformations," en, D. Lockton, S. Lenzi, P. Hekkert, A. Oak, J. Sádaba, and P. Lloyd, Eds., eds.), DRS2022: Bilbao, 27 June - 3 July, Bilbao, Spain, 2022. DOI: 10.21606/drs.2022. 897.
- [169] M. M. Keitsch, "Transdisciplinary collaboration and ethics," in *Transdisciplinarity For Sustainability : Aligning Diverse Practices*, M. M. Keitsch and W. J. V. Vermeulen, Eds., London: Routledge, 2021, ISBN: 9780367189075.
- [170] E. Berkers and F. W. Geels, "System innovation through stepwise reconfiguration: The case of technological transitions in dutch greenhouse horticulture (1930-1980)," *Technology analysis & strategic management*, vol. 23, no. 3, pp. 227–247, 2011. DOI: 10.1080/09537325.2011.550392.
- [171] C. M. Christensen, *The innovator's dilemma : when new technologies cause great firms to fail* (The Management of innovation and change series). Boston, Mass: Harvard Business Review Press, 2016.
- [172] p. Anderson and m. L. tushman m., "Technological discontinuities and dominant designs: A cyclical model of technological change," *Administrative science quarterly*, vol. 35, no. 4, pp. 604–633, 1990. DOI: 10.2307/ 2393511.
- [173] J. A. Schumpeter, *Capitalism, socialism and democracy*. London: Routledge, 1994.
- [174] L. Fuenfschilling, Keynote international sustainability transitions conference (ist) 2021, https://www.youtube.com/watch?v=-cPDmmfrmzU, Accessed: 10.05.2022, 2021. [Online]. Available: https://www.youtube. com/watch?v=-cPDmmfrmzU.
- [175] A. Schaminée, Designing with in public organizations : building bridges between public sector innovators and designers. Amsterdam: BIS publishers, 2018.
- [176] N. Shedroff, *Design is the problem : the future of design must be sustainable.* New York: Roselfeld Media, 2009.
- [177] W. M. Cohen and D. A. Levinthal, "Absorptive capacity: A new perspective on learning and innovation," *Administrative science quarterly*, vol. 35, no. 1, pp. 128–152, 1990, ISSN: 0001-8392. DOI: 10.2307/2393553.
- [178] J. G. March, "Exploration and exploitation in organizational learning," Organization Science, vol. 2, no. 1, pp. 71–87, Aug. 1991. [Online]. Available: http://www.jstor.org/stable/2634940.
- [179] G. J. Langley, R. D. Moen, K. M. Nolan, T. W. Nolan, C. L. Norman, and L. P. Provost, *The improvement guide: a practical approach to enhancing organizational performance, second edition*, 2nd ed. San Francisco, CA: San Francisco, CA: Jossey-Bass, 2009.

- [180] D. Kolb, *Experiential Learning : Experience as the Source of Learning and Development, Second Edition,* 2nd edition. PH Professional Business, 2014.
- [181] M. Matsuo and J. Nakahara, "The effects of the pdca cycle and ojt on workplace learning," *INT J HUM RESOUR MAN*, vol. 24, no. 1, pp. 195– 207, 2013, ISSN: 0958-5192. DOI: 10.1080/09585192.2012.674961.
- [182] I. Bens, Facilitating with ease! : core skills for facilitators, team leaders and members, managers, consultants, and trainers (Jossey-Bass business & management series), 3rd ed. San Francisco: Jossey-Bass, 2012.
- E. E. Öztekin and A. İ. Gaziulusoy, "Designing transitions bottom-up: The agency of design in formation and proliferation of niche practices," *The Design journal*, vol. 22, no. 1, pp. 1659–1674, 2019, ISSN: 1460-6925. DOI: 10.1080/14606925.2019.1594999.
- [184] A. Thorpe and L. Gamman, "Design with society: Why socially responsive design is good enough," *CoDesign*, vol. 7, no. 3-4, pp. 217–230, 2011, ISSN: 1571-0882. DOI: 10.1080/15710882.2011.630477.
- [185] C. DiSalvo, T. Lodato, L. Fries, B. Schechter, and T. Barnwell, "The collective articulation of issues as design practice," *CoDesign*, vol. 7, no. 3-4, pp. 185–197, 2011, ISSN: 1571-0882. DOI: 10.1080/15710882.2011.630475.
- [186] P.A. Hillgren, A. Seravalli, and A. Emilson, "Prototyping and infrastructuring in design for social innovation," *CoDesign*, vol. 7, no. 3-4, pp. 169– 183, 2011, ISSN: 1745-3755. DOI: 10.1080/15710882.2011.630474.
- [187] E. Manzini and F. Rizzo, "Small projects/large changes: Participatory design as an open participated process," *CoDesign*, vol. 7, no. 3-4, pp. 199– 215, 2011, ISSN: 1571-0882. DOI: 10.1080/15710882.2011.630472.
- [188] Norwegian register for scientific journals, series and publishers, *Mdpi*, Accessed: 10.05.2022, May 2021. [Online]. Available: https://kanalregister.hkdir.no/publiseringskanaler/KanalForlagInfo.action?id=26778&request locale=en.
- J. Kolko, Exposing the Magic of Design: A Practitioner's Guide to the Methods and Theory of Synthesis (Human Technology Interaction Series). Cary: Cary: Oxford University Press, 2010. DOI: 10.1093/acprof:oso/9780199744336. 001.0001.
- [190] E. M. Oftedal, L. Foss, and T. Iakovleva, "Responsible for responsibility? a study of digital e-health startups," *Sustainability (Basel, Switzerland)*, vol. 11, no. 19, pp. 5433–5433, 2019. DOI: 10.3390/su1195433.
- [191] O. Bongomin, A. Yemane, B. Kembabazi, C. Malanda, M. Chikonkolo Mwape, N. Sheron Mpofu, and D. Tigalana, "Industry 4.0 disruption and its neologisms in major industrial sectors: A state of the art," *Journal of Engineering*, vol. 2020, pp. 1–45, Oct. 2020. DOI: 10.1155/2020/8090521.

- [192] A. Awad, S. J. Trenfield, T. D. Pollard, J. J. Ong, M. Elbadawi, L. E. Mc-Coubrey, A. Goyanes, S. Gaisford, and A. W. Basit, "Connected healthcare: Improving patient care using digital health technologies," *Advanced drug delivery reviews*, vol. 178, pp. 113 958–113 958, 2021. DOI: 10.1016/j.addr.2021.113958.
- [193] S. Sodero and R. Rackham, "Blood drones: Using utopia as method to imagine future vital mobilities," *Mobilities*, vol. 15, no. 1, pp. 11–24, 2019, ISSN: 1745-011X,1745-0101. DOI: 10.1080/17450101.2019.1673034.
- [194] H. W. J. Rittel and M. M. Webber, "Dilemmas in a general theory of planning," *Policy sciences*, vol. 4, no. 2, pp. 155–169, 1973, ISSN: 0032-2687.
 DOI: 10.1007/BF01405730.
- [195] I. Wanzenböck and K. Frenken, "The subsidiarity principle: Turning challengeoriented innovation policy on its head," Tech. Rep. 1806, Jan. 2018. [Online]. Available: https://EconPapers.repec.org/RePEc:egu:wpaper: 1806.
- [196] C. Wang, T. Lv, R. Cai, J. Xu, and L. Wang, "Bibliometric analysis of multilevel perspective on sustainability transition research," *Sustainability (Basel, Switzerland)*, vol. 14, no. 7, p. 4145, 2022. DOI: 10.3390/su14074145.
- P. Kivimaa, W. Boon, S. Hyysalo, and L. Klerkx, "Towards a typology of intermediaries in sustainability transitions: A systematic review and a research agenda," *Research policy*, vol. 48, no. 4, pp. 1062–1075, 2019, ISSN: 0048-7333. DOI: 10.1016/j.respol.2018.10.006.
- [198] A. İ. Gaziulusoy, "Postcards from "the edge": Toward futures of design for sustainability transitions," *Cuadernos del Centro de Estudios de Diseño y Comunicación*, no. 73, pp. 66–83, 2019, ISSN: 1853-3523. DOI: 10.18682/ cdc.vi73.1038.
- [199] C. Roberts and F. W. Geels, "Public storylines in the british transition from rail to road transport (1896-2000): Discursive struggles in the multi-level perspective," *Science as Culture*, vol. 27, no. 4, pp. 513–542, 2018. DOI: 10.1080/09505431.2018.1519532.
- [200] M. A. Demircioglu and D. B. Audretsch, "Conditions for innovation in public sector organizations," *Research policy*, vol. 46, no. 9, pp. 1681–1691, 2017. DOI: 10.1016/j.respol.2017.08.004.
- [201] A. Stirling, "Deliberate futures: Precaution and progress in social choice of sustainable technology," *SUSTAIN DEV*, Sustainable Development, vol. 15, no. 5, pp. 286–295, 2007, ISSN: 0968-0802. DOI: 10.1002/sd.347.
- [202] M. P. Schlaile, S. Urmetzer, V. Blok, A. D. Andersen, J. Timmermans, M. Mueller, J. Fagerberg, and A. Pyka, "Innovation systems for transformations towards sustainability? taking the normative dimension seriously," *Sustainability-basel*, vol. 9, no. 12, p. 2253, 2017, ISSN: 2071-1050. DOI: 10.3390/su9122253.
- [203] E. Uyarra, B. Ribeiro, and L. Dale-Clough, "Exploring the normative turn in regional innovation policy: Responsibility and the quest for public value," *null*, vol. 27, no. 12, pp. 2359–2375, Dec. 2019, ISSN: 0965-4313. DOI: 10.1080/09654313.2019.1609425. [Online]. Available: https://doi.org/10.1080/09654313.2019.1609425.
- [204] P. Maclaine Pont, R. van Est, and J. Deuten, *Shaping socio-technical inno-vation through policy: Essay commissioned by the department of knowledge, innovation and strategy of the dutch ministry of infrastructure and the environment, 2016.*
- [205] F. W. Geels, "From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory," *Research Policy*, vol. 33, no. 6, pp. 897–920, 2004, ISSN: 0048-7333. DOI: https://doi.org/10.1016/j.respol.2004.01.015. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S0048733304000496.
- [206] D. S. Smirnov and K. W. Willoughby, "Rethinking the dynamics of innovation, science, and technology: The curious case of stirling engines and stirling refrigerators," *Energy res soc sci*, vol. 79, p. 102 159, 2021, ISSN: 2214-6296. DOI: 10.1016/j.erss.2021.102159.
- [207] S. Sorrell, "Explaining sociotechnical transitions: A critical realist perspective," *Research Policy*, vol. 47, no. 7, pp. 1267–1282, Sep. 2018. DOI: 10. 1016/J.RESPOL.2018.04.008.
- [208] M. Keller, M. Noorkõiv, and T. Vihalemm, "Systems and practices: Reviewing intervention points for transformative socio-technical change," *Energy Research & Social Science*, vol. 88, p. 102608, 2022, ISSN: 2214-6296. DOI: 10.1016/j.erss.2022.102608. [Online]. Available: https: //www.sciencedirect.com/science/article/pii/S2214629622001128.
- [209] F. W. Geels, B. K. Sovacool, T. Schwanen, and S. Sorrell, "Sociotechnical transitions for deep decarbonization," *Science*, vol. 357, no. 6357, pp. 1242–1244, 2017. DOI: 10.1126/science.aao3760.
- [210] G. Harvey and E. Lynch, "Enabling continuous quality improvement in practice: The role and contribution of facilitation," *Frontiers in Public Health*, vol. 5, 2017, ISSN: 2296-2565. [Online]. Available: https://www.fronti ersin.org/article/10.3389/fpubh.2017.00027.
- [211] A. M. Nguyen, A. Cuthel, D. K. Padgett, P. Niles, E. Rogers, H. Pham-Singer, D. Ferran, S. A. Kaplan, C. Berry, and D. Shelley, "How practice facilitation strategies differ by practice context," *Journal of General Internal Medicine*, vol. 35, no. 3, pp. 824–831, 2020, ISSN: 1525-1497. DOI: 10.1007/s 11606-019-05350-7. [Online]. Available: https://doi.org/10.1007/ s11606-019-05350-7.

- [212] S. Rushton, A. A. Lewinski, S. Hwang, L. L. Zullig, K. A. Ball Ricks, K. Ramos, A. Gordon, B. Ear, L. A. Ballengee, M. V. Brahmajothi, T. Moore, D. V. Blalock, J. W. Williams Jr, S. E. Cantrell, J. M. Gierisch, and K. M. Goldstein, "Barriers and facilitators to the implementation and adoption of improvement coaching: A qualitative evidence synthesis," *J Clin Nurs*, vol. n/a, no. n/a, Apr. 2022, ISSN: 0962-1067. DOI: 10.1111/jocn.16247. [Online]. Available: https://doi.org/10.1111/jocn.16247.
- [213] A. İ. Gaziulusoy and C. Ryan, "Roles of design in sustainability transitions projects: A case study of visions and pathways 2040 project from australia," *Journal of cleaner production*, vol. 162, pp. 1297–1307, 2017. DOI: 10.1016/j.jclepro.2017.06.122.
- [214] D. Cawthorne and A. R. Wynsberghe, "From healthdrone to frugaldrone: Value-sensitive design of a blood sample transportation drone," 2019, pp. 1–7. DOI: 10.1109/ISTAS48451.2019.8938019.
- [215] OECD, Fostering Innovation in the Public Sector. 2017, pp. 252–252. DOI: 10.1787/9789264270879 - en. [Online]. Available: https://www.oecdilibrary.org/content/publication/9789264270879 - en.
- [216] A. Genus and A.-M. Coles, "Rethinking the multi-level perspective of technological transitions," *Research Policy*, vol. 37, no. 9, pp. 1436–1445, 2008, ISSN: 0048-7333. DOI: https://doi.org/10.1016/j.respol.2008.05.006. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S0048733308001169.
- [217] C. Peter and M. Swilling, "Linking complexity and sustainability theories: Implications for modeling sustainability transitions," *Sustainability*, vol. 6, no. 3, pp. 1594–1622, 2014, ISSN: 2071-1050. DOI: 10.3390/su6031594.
- [218] B. T. Pentland and M. S. Feldman, "Designing routines: On the folly of designing artifacts, while hoping for patterns of action," *Information and organization*, vol. 18, no. 4, pp. 235–250, 2008. DOI: 10.1016/j.infoan dorg.2008.08.001.
- [219] A. Raghunatha, P. Thollander, and S. Barthel, "Addressing the emergence of drones - a policy development framework for regional drone transportation systems," *Transportation research interdisciplinary perspectives*, vol. 18, p. 100 795, 2023, ISSN: 2590-1982. DOI: 10.1016/j.trip.2023. 100795.
- [220] S. Laakso, R. Aro, E. Heiskanen, and M. Kaljonen, "Reconfigurations in sustainability transitions: A systematic and critical review," *Sustainability : science, practice, & policy*, vol. 17, no. 1, pp. 15–31, 2021, ISSN: 1548-7733. DOI: 10.1080/15487733.2020.1836921.
- [221] J. Grin, J. Rotmans, and J. Schot, "Transitions to sustainable development," in *Transitions to Sustainable Development: New Directions in the Study of Long Term Transformative Change*. Routledge, May 26, 2010. DOI: 10.4324/9780203856598.

[222] J. Köhler, F. W. Geels, F. Kern, J. Markard, E. Onsongo, A. Wieczorek, F. Alkemade, F. Avelino, A. Bergek, F. Boons, L. Fünfschilling, D. Hess, G. Holtz, S. Hyysalo, K. Jenkins, P. Kivimaa, M. Martiskainen, A. McMeekin, M. S. Mühlemeier, B. Nykvist, B. Pel, R. Raven, H. Rohracher, B. Sandén, J. Schot, B. Sovacool, B. Turnheim, D. Welch, and P. Wells, "An agenda for sustainability transitions research: State of the art and future directions," *Environmental Innovation and Societal Transitions*, vol. 31, pp. 1–32, 2019, ISSN: 2210-4224. DOI: 10.1016/j.eist.2019.01.004. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2210422418303332.

Paper 1

H. E. Comtet and K.-A. Johannessen, "A socio-analytical approach to the integration of drones into health care systems," *Information* (*Basel*), vol. 13, no. 62, p. 62, 2022. DOI: 10.3390/info13020062



MDP

Article A Socio-Analytical Approach to the Integration of Drones into Health Care Systems

Hans E. Comtet ^{1,2,*} and Karl-Arne Johannessen ¹

- ¹ The Intervention Centre, Oslo University Hospital (OUS), 0372 Oslo, Norway; karajoha@online.no
- ² Department of Design, Norwegian University of Science and Technology (NTNU), 7491 Trondheim, Norway
- * Correspondence: hanscomtet@gmail.com

Abstract: The integration of drones into health care as a supplement to existing logistics methods may generate a need for cooperation and involvement across multiple resource areas. It is currently not well understood whether such integrations would merely represent a technical implementation or if they would cause more significant changes to laboratory services. By choosing socio-technical theory as the theoretical lens, this paper intends to harvest knowledge from the literature on various organizational concepts and examine possible synergies between such theories to determine optimal strategies for introducing the use of drones in a health care context. Our particular interest is to examine whether the insights generated from the multi-level perspective (MLP) may have the potential to create dynamic spin-offs related to the organizational transitions associated with the implementation of drones in health services. We built our study on a scoping literature review of topics associated with the MLP and socio-technical studies from differing arenas, supplemented with studies harvested on a broader basis. The scoping review is based on 25 articles that were selected for analysis. As a way of organizing the literature, the niche, regime, and landscape levels of the MLP are translated to the corresponding health care-related terms, i.e., clinic, institution, and health care system. Furthermore, subcategories emerged inductively during the process of analysis. The MLP provides essential knowledge regarding the context for innovation and how the interaction between the different levels can accelerate the diffusion of innovations. Several authors have put both ethical topics and public acceptance into a socio-technological perspective. Although a socio-technical approach is not needed to operate drones, it may help in the long run to invest in a culture that is open to innovation and change.

Keywords: drones; socio-technical theory; scoping review; multi-level perspective; health care

1. Introduction

As health care costs surge and the need for resources seems to exceed any realistic prospect of supplying them, new technological solutions are being pursued to save costs and reduce the need for specialized resources [1–5]. Among the strategies for reducing the cost curve that several health care systems are considering is improved transport and logistics using drones for multiple purposes.

There are multiple reports on drones in health care, ranging from search and rescue following natural disasters, drug and vaccine delivery in rural districts, the provision of care technology in emergency situations and the transportation of blood samples and organs have been studied [6–15]. Transport of biological samples across laboratories and institutions and of blood samples from remote locations to central laboratories has gained special interest [6,16,17] and promises faster and improved laboratory services by providing service to rural districts and enabling savings.

Drones may be a complete substitute for ground transport in areas where roads are non-existent for large parts of the year. Alternatively, drones may also be relevant as supplements to existing logistics methods, where they can be integrated into existing ground transport systems as an extension to provide last-mile or on-demand services to



Citation: Comtet, H.E.; Johannessen, K.-A. A Socio-Analytical Approach to the Integration of Drones into Health Care Systems. *Information* **2022**, *13*, 62. https://doi.org/10.3390/info13020062

Academic Editor: Lorenzo Mucchi

Received: 10 December 2021 Accepted: 23 January 2022 Published: 26 January 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). meet time-critical demands [18–21]. Such integrations may generate a need for cooperation and involvement across multiple resource areas, i.e., across medical, logistical, and transport workforces. How such processes should best be developed and implemented has not yet been studied extensively.

Another interesting topic is whether drones can be implemented in ways that have a broad impact on service and organization models. Creating support for the more substantial transformation of multiple services far beyond logistics offers a space to create sustainable service concepts and ideas with a broad perspective. There is increasing interest in how such extended integrations of drones in health care will interact with the social context of the human ecosystem of stakeholders and executing workforces [22,23].

Whether the implementation of drone transport will transform the laboratory services and logistics operations and health services [24], or if it is merely a technical implementation [25], is currently an unanswered question. However, implementing drones to create sustainability transitions in an extended perspective is a fascinating prospect that may require facilitating interactions between the realms of policy, economics, markets, culture, technology, and possibly public opinion.

Multiple theoretical models have been suggested for approaching such challenges, illustrating the multidimensional nature of sustainability transitions combined with the different aspects of structural change. Transitions for sustainability are goal-oriented with a specific purpose, and combinations of many "sustainable" solutions do not always offer obvious user benefits, as sustainability is a collective benefit. Challenges can present themselves during the process of both cultural and structural change: the existing, unsustainable systems may be fixed in place by various lock-in mechanisms related to the existing infrastructure, current competencies, and benefits for stakeholders, employees, and users, thus creating a dependence trajectory that makes it difficult to replace existing systems. Along with this broader definition of the problem comes a need for broader analytical perspectives [26].

Multiple models have been proposed as approaches to replacing and reconfiguring technological systems. For example, Hekkert et al. [27] described a technological innovation system approach from a multidimensional perspective but did not address structural change. The disruptive innovation approach of Christensen et al. [28] and the technological discontinuity model of Anderson and Tushman [29] are also helpful approaches but are mainly focused on the technology and market dimensions.

In contrast, Geels [30,31] and Geels and Schot [32] processed and refined the multilevel perspective (MLP) framework to explain how changes take place in socio-technical systems. As a follow-up related to drones, Haula et al. [33] suggested that "Interpreting drones through the lens of socio-technical theory, drones cannot be a standalone technological infrastructure but require an ecosystem to function optimally; humans to develop and manage them; regulations to protect the drones as well as people's freedoms from infringement; and perform the necessary responsibilities they were built for". This is an interesting hypothesis, although not further justified in their study. In addition, some of the previous statements on these topics may need modifications, for example, because future drones will mainly be autonomous, operated by remote systems with little need for personal attendance.

In our current context, we assume that drones are first implemented at the level of operational units (clinical units performing medical services, laboratory analysis, transport logistics) before generating innovations in a broader perspective at the institutional and health care system levels. The extent to which such use cases will require comprehensive organizational processes for implementation has not been extensively explored or understood. However, in designing the managerial policies related to drone implementations, it may be useful to profit from and build upon the experience of transformative policies in other areas of technology and innovation because drone solutions will combine several technological knowledge areas, such as software and hardware engineering, artificial intelligence, machine learning, internet of things (IoT) and logistical competence [34–36].

Based on a system design approach, we use the MLP [37] because it goes beyond studies of single technologies to focus on the various groups of stakeholders and their strategies, resources, beliefs, and interactions. We apply the MLP's micro (also called niche), meso (regime), and macro (landscape) perspectives and translate them into the health care context, where they parallel the levels of the clinic, hospital institutions, and health care systems in general (Figure 1).



Figure 1. The multi-level perspective on system innovations (adapted from Geels [38]).

In this study, we examine whether the principles of the MLP regarding system innovations as merging transitions from one socio-technical system to another may offer a framework that can be helpful for understanding system innovations related to drones in health care [33,39–42]. By combining the three MLP, i.e., dimensions speed of change, size of change, and period of change [43] and the four MLP phases [42,43], i.e., introduction of technology within the existing environment, exploration of functionalities and user preferences, putting change into practice in daily operations, and gradual replacement of existing solutions, we examine the current knowledge regarding the implementation of drones in health care [42]. As we assume drones to be implemented in a "bottom-up" process, we use the levels in the MLP to sort out the various stages of the implementation of drones in health care systems.

Emerging technologies are usually influenced by the existing institutional solutions (in our context, hospitals), which operate in a stable configuration until a new technology emerges and creates an interplay of multiple technologies [37]. Compliance with legacy systems, as well as ethical standards and standards of clinical and laboratory processes, are crucial in such processes [44].

We believe that drones may be used efficiently for a multitude of purposes in health services, and we intend to harvest knowledge from the literature on various organizational concepts and examine possible synergies between such theories. We built our study on a scoping literature review of topics associated with the MLP and socio-technical studies from differing arenas, supplemented with studies collected on a broader basis. Our particular interest is to examine the potential to create dynamic spin-offs related to the organizational transitions associated with the implementation of drones in health services. The study is based on the following research question:

What knowledge of socio-technical theories may support an extended focus associated with implementing drones into health care systems?

The article proceeds as follows: in the next section, we introduce our scoping review method. In Section 3, we present the results and identify the categories of our research focus. Next, we use the clinic (niche), institution (regime), and health care system (landscape) levels to organize the literature, using a procedure related to the MLP inspired by Prayag and Ozanne [45]. In Section 4, we discuss our findings. Finally, in Section 5, we offer our conclusions.

2. Materials and Methods

Scoping reviews can be used to provide an overview of a given topic [46]. Furthermore, in comparison, for example, with systematic reviews, the review question can have a broader "scope" which is in accordance with the research aim of this study. This scoping review follows the approach set out by Arksey and O'Malley [47]. The framework consists of five stages: (1) identifying research questions and search terms, (2) identifying relevant studies, (3) selecting studies, (4) charting and analyzing data, and (5) collecting, summarizing, and reporting results.

2.1. Research Question and Search Terms

Based on the question "What knowledge of socio-technical theories may support an extended focus associated with implementing drones into health care systems?" we created three search concepts: "Drones, UAV, UAS", "Health care, Health Systems, System Integration", and "Transportation, Logistics, Innovation".

These terms were included in the search strategy because they define the context. Furthermore, "system integration" was included to cover integration from technical, organizational, and social perspectives. The search strategy aimed to identify the relevant literature concerning:

- The integration of drones into existing systems;
- Potential drivers of and barriers to the integration of drones;
- Prerequisites for the integration of drones.

2.2. Identifying the Relevant Literature

The literature search was conducted in March 2021 using the PubMed and Scopus databases and then supplemented with an additional snowball strategy to retrieve other relevant articles. Three strings were created with the operator OR and a combined search with the operator AND for the structured search. The list below displays the search strategy:

- 1. Drones OR Unmanned Aerial Vehicles OR Unmanned Aerial System;
- 2. Healthcare OR Health Systems OR Systems Integration;
- 3. Transportation OR Logistics OR Innovations;
- 4. 1 AND 2 AND 3.

2.3. Exclusion Criteria for Literature

Based on our primary interest in the use of drones to support laboratory services with a potential to extend such solutions in a broad perspective, articles that described the use of drones to capture images or video footage, their use in humanitarian response, or that focused on algorithms, physical drone parts, energy consumption, or carbon emissions were excluded.

2.4. Analysis and Charting of the Data

The technical process of analyzing the textual content was conducted using coding [48]. The content analysis software Atlas.ti (Version 9.1.3) was used as a tool to organize the data digitally. A first coding scheme was jointly developed by the authors and gradually adapted to the different categories. An inductive process allowed the codes to emerge

naturally from the data [49]. Mayring [50] described this category development process as qualitative content analysis, in which categories emerge over incremental revisions and reductions of the categories while working through the text.

3. Results

The search of PubMed with the first, second, and third strings yielded 2101, 1,493,810, and 3,624,067 results, respectively. The combined concept search yielded 51 results from the years 2014 to 2021. After applying the exclusion criteria, a final sample of 29 articles was derived.

The search in Scopus with the three strings yielded 68,376, 504,787, and 1,418,049 results. The combined query yielded 67 search results from the years 2012 to 2021. After applying the exclusion criteria, a final sample of 49 articles was derived.

During the full-text review of the database results, 19 additional articles were identified for inclusion using a snowball strategy.

A total of 97 articles were included for full-text review. First, we scanned the text to identify articles that included the keywords "innovation", "socio", "culture", "leadership", and/or "integration". This approach resulted in a sample of 30 articles.

Second, the three MLP levels were chosen to organize the articles. The two authors categorized the papers blinded to each other and finalized the results by consensus when initial disagreement occurred (five papers). The remaining 25 articles are the subject of this results section.

Figure 2 below provides a detailed outline of the document inclusion process from the scoping review.

The earliest article was from 2016, but 16 of the articles were published between 2019 and 2021 (64%).

Table 1 below presents the author keywords from the review sample. The analysis of the 92 keywords shows that 22% of the articles used "drone", "drones", "UAV", "UAS", "drone integration", "drone communication" or "drone design". In addition, 16% used terms referring to technology, 10% to health care, and 8% to logistics and transportation. The remaining 43% of keywords used were only used once. No specific keyword regarding socio-technical systems was identified.

Count	Percent	Keyword
20	22%	Drones (including UAV, UAS, integration, communication, design)
16	17%	Technology (including AI, IoT, Machine Learning, Blockchain, 5G, automation, innovation, and disruption)
9	10%	Healthcare (including laboratory, microbiology, health systems, and services)
7	8%	Logistics (including transport, delivery, and supply chains)
40	43%	40 unique keywords: open science; biobanking; iTRANS; bystander CPR RPAS;
		Canada; prehospital care; throughput; cell phone data; intelligent transportation systems (ITS) platform; learning health care system; community engagement; mobile microbiology; consolidation; policy; Danish public healthcare; remote medicine; surveillance; disasters; battlefield medicine; emergencies; massive open online education; emerging infectious diseases; medium access control; EMS dispatcher; noncommunicable diseases; energy efficiency; out-of-hospital cardiac arrest public access defibrillation AED; epilepsy; portable instruments; ethical framework; public access defibrillation; global health precision medicine; health applications; sudden cardiac arrest; telemedicine; automatic external defibrillation; United States; value-sensitive design (VSD); user experience; values hierarchy
92	100%	

Table 1. Keyword analysis.

Starting with the three MLP levels, ten subcategories emerged from the analysis. Table 2 displays a list of the articles included in this study, together with the corresponding MLP levels and subcategories.



Figure 2. Document inclusion and exclusion process.

Table 2. Included articles.

Dimension	Category	Author Ferreras [51]; Gruson [24]; Vandenberg et al. [52] Eichleay et al. [25]; Mishra et al. [53]; Ferreras [51]; Khisa et al. [54]; Syed et al. [55] Mion [56]; Zegre-Hemsey et al. [57]; Poljak et al. [58] Van de Voorde et al. [59]; Shawn et al. [60] Balasingam [61]; Braun et al. [62]; Nentwich et al. [63] Vandenberg et al. [52]; Flahault et al. [64] Bhavnani et al. [65]; Mishra et al. [53]; Cawthorne et al. [66]; Mion [56]; Johannessen et al. [44] Ferreras [51]: Braun et al. [62]: Hiebert et al. [19]: Mion [56]:			Author						
	Digitalization	Ferreras [51]; Gruson [24]; Vandenberg et al. [52]									
Clinic	Integration of different technologies and services	Eichleay et al. [25]; Mishra et al. [53]; Ferreras [51]; Khisa et al. [54]; Syed et al. [55]									
Child	Public Acceptance	Mion [56]; Zegre-Hemsey et al. [57]; Poljak et al. [58] Van de Voorde et al. [59]; Shawn et al. [60]									
	Regulation/Legislation	Author Ferreras [51]; Gruson [24]; Vandenberg et al. [52] Eichleay et al. [25]; Mishra et al. [53]; Ferreras [51]; Khisa et al. [54]; Syed et al. [55] Mion [56]; Zegre-Hemsey et al. [57]; Poljak et al. [58] Van de Voorde et al. [59]; Shawn et al. [60] Balasingam [61]; Braun et al. [62]; Nentwich et al. [63] Vandenberg et al. [52]; Flahault et al. [64] Bhavnani et al. [65]; Mishra et al. [53]; Cawthorne et al. [66]; Mion [56]; Johannessen et al. [44] Ferreras [51]; Braun et al. [62]; Hiebert et al. [19]; Mion [56]; Truog et al. [67]									
	Integration challenges	Vandenberg et al. [52]; Flahault et al. [64]									
Institution	Facilitating innovation processes	Bhavnani et al. [65]; Mishra et al. [53]; Cawthorne et al. [66]; Mion [56]; Johannessen et al. [44]									
	Collaboration	Bhavnani et al. [65]; Mishra et al. [53]; Cawthorne et al. [66]; Mion [56]; Johannessen et al. [44] Ferreras [51]; Braun et al. [62]; Hiebert et al. [19]; Mion [56]; Truog et al. [67]									

Dimension	Category	Author						
	Adoption	Mion [56]; Hiebert et al. [19]; Johannessen et al. [44]						
Health care	Diffusion/Acceleration	Flahault et al. [64]; Mateen et al. [68]; Mion [56]						
	Change and transitions in relation to ethics	Faramondi et al. [69]; Cawthorne et al. [66]; Eichleay et al. [25] Carrillo-Larco et al. [70]; Mishra et al. [53]; Greaves et al. [71]						

Table 2. Cont.

3.1. Additional Documents

During our analysis of the documents and development of the discussion, we supplemented our literature review with further publications where appropriate, following a deductive identification of relevant articles. We included five additional recently published review articles [72–76], one essay [77], two online sources [78,79], and five research articles [80–84].

In addition, we identified eight MLP articles of relevance and included them in our discussion [26,30,40,85–89].

Even though the main arguments supporting the use of drones are their promise in terms of cost reductions, their ability to avoid ground traffic congestion, and their ability to maintain services in areas with poor transportation infrastructure and in hard-to-reach areas, only the review article by Hiebert et al. [19], among the articles identified in the scoping review, discussed these topics. We therefore specifically searched for papers dealing with costs and rural services to see if studies discussing perspectives related to our study subject were available and included four more articles [18,90–92] in addition to the study by Hiebert et al.

3.2. Descriptions of Findings

The descriptions of our literature findings are categorized according to whether they are concerned with the clinical, institutional, or health care system levels.

The Clinical Level (Niche)

The important topics discussed in these articles include digitalization as a driver of technological developments in general, the integration of drones related to their acceptance in public space and the role of regulation and legislation as enablers of future drone services.

3.3. Digitalization

Ferreras [51] discussed whether disruptive innovations involving robot vehicles supported by telecommunications or computers may be conducive to more efficient transportation and logistics solutions. From his perspective, drones represent an additional and new element in developing autonomous or semiautonomous vehicles, and there is substantial potential in their independence from heavy ground traffic and congestion.

Gruson [24] and Vandenberg et al. [52] discussed the digitalization and automation of laboratories. They provided an interesting perspective on the impact of technologies on laboratories. Gruson pointed out that integrating big data and real-time management, automation, blockchain, the internet of things (IoT), and the enhancement of the user experience are critical drivers for achieving "smart digital laboratories". Similar factors may also be relevant to perspectives related to drone solutions. As with Gruson, Vandenberg et al. supported the conclusion that new IT platforms that support integrated datasets across primary and secondary care and measures of outcomes and costs across patient pathways are needed.

3.4. Integration of Different Technologies and Services

Eichleay et al. [25] focused on the implementation of drones in either parallel or integrated setups into existing structures and systems. They pointed out the dilemma that

integration into existing structures may take too much time, whereas devising parallel technological systems may fracture health systems and cause informational and operational silos that may result in future inefficiencies. Furthermore, they also concluded that more information on what processes must be changed and how they impact workflows and health outcomes may contribute to the evaluation of sustainability and the actual cost of implementation.

Mishra et al. [53] considered various perspectives on integration synergies between 5G cellular systems and UAV technology. Their survey hypothesized that 5G technologies would enable seamless integration and UAV communication over mobile networks. Furthermore, B5G (so-called "6G") innovations may further enhance the performance and applicability for seamless integration of UAVs into mobile networks.

Ferreras suggested that the full potential of UAVs may be achieved with a combination of several technologies that need to be developed. For example, receiving constant location data provided by autonomous vehicles or autonomous drones in combination with powerful 3D computer visualizations could be used to build future computer interfaces for optimized transportation. In this transition, computer systems would move away from being human control-oriented toward being automated, and self-deciding systems [51].

The integration of UAVs with IoT networks was described as a new direction for research and industry by Khisa et al. [54]. IoT enables things to be connected anywhere, anytime, using any network, thus enabling almost any service. One of the most promising technologies for handling security-related issues in communication is the integration of blockchain with UAV-based IoT. Several studies have been carried out regarding the integration of machine learning mechanisms with IoT and UAV.

The versatility of drones and the potential to support different efficient solutions for smart transport applications were highlighted in the survey by Syed et al. [55]. For example, machine learning (ML) techniques, blockchain, or watermarking may alleviate security concerns regarding drones.

3.5. Public Acceptance

Mion [56] raised an essential question about public acceptance, pointing out that the general acceptance of new technological instruments is usually higher in the health care sector than in other domains, e.g., the application of drones in emergency situations such as the delivery of automated external defibrillators. The fact that drones may save minutes, which can save a life in critical clinical situations (Zegre-Hemsey et al. [57]), is observed as having high value for society and individuals, thus facilitating acceptance.

Based on case studies in Papua New Guinea, Asia, and Africa, a review by Poljak et al. [58] found that drones were generally well accepted. However, they concluded that more research may be needed to understand public acceptance in highly populated environments and high-density traffic airspaces. For example, Zegre-Hemsey et al. [57] suggested that as innovations in drone technology are entering emergency cardiac care [7,93], it may be necessary to explore ways to integrate drones into these environments.

The effect of the combination of purposes and actors involved on public acceptance was also discussed by Van de Voorde et al. [59]. Their article discussed how when police forces use drones for surveillance, regulations for safe and responsible handling of the drones and of the data they provide should not fall outside standard regulations. Relating this example to the transport of biological material, drones in this health care setting may have a higher chance of being accepted. However, at the same time, the proper and secure handling of patient-related data must be guaranteed.

The impact of increasingly demanding patients who expect the same level of innovation, service, and quality from health service providers that they see in other service sectors, such as online shopping, travel, and media, was discussed by Shawn et al. [60]. Consumers profit from being able to choose how their care offering is provided and can pick between different delivery models, including home health, concierge care, and online self-help. These new channels offer consumers qualities they look for from other service providers, such as convenience, thoughtfulness, timeliness, value, and price transparency [94]. Such qualities may contribute to the acceptance of drone services.

Public acceptance and social benefit may act as enablers for policymaking regarding drones [59]. It may, therefore, be essential to understand public acceptance when new environments are being explored.

3.6. Regulation/Legislation

Several authors have discussed the challenge of national legislation. Balasingam [61] argued that the lack of timely legislation often forces organizations to apply for exemptions, hampering the progress of technological innovation. According to Braun et al. [62], the pace of drone design innovation generated by new applications places considerable demands on governmental and local regulatory agencies, which are not always able to keep up with the pace of innovation [95].

Such regulatory lags due to slow adaptation to new technological options were also discussed by Nentwich et al. [63]. The consequences related to new technologies and rapid developments such as drones are that barriers to innovation are created, i.e., autonomous drones could not previously be licensed under existing regulations. As a remedy for this discrepancy in regulatory speed, Balasingam suggested that the stakeholders involved in the medical and drone industries, insurance companies, legislative authorities, and government bodies should work together and develop prompt legislative solutions to integrate drones seamlessly into our communities [61].

It is interesting that although they were published only a few years ago, the findings of several of these papers have fortunately become outdated because proper regulations are being put in place in both the US and the EU [78,79].

The Institutional Level (Regime)

Integrating existing services and collaborations between various stakeholders at the institutional level may be essential to analyze and understand how long-term improvements and value creation should be achieved.

3.7. Integration Challenges

A perspective discussed by Vandenberg et al. [52] is how the introduction of laboratory automation and the linkage of information systems for big(ger) data management, including artificial intelligence (AI), may also strengthen drone implementations. However, they pointed out that the initial optimism associated with these developments has entered a more reality-based phase of reflection on the significant challenges, complexities, and health care benefits posed by these innovations.

Parallels between the integration of different technology systems in general and the implementation of drones were pictured by Flahault et al. [64]. They described the impact of a future health system that connects personal, provider, and population-level health information. This would provide feedback loops on many levels, thus creating computer-supported mechanisms for learning and improving the quality of the overall health system. They describe drones as an exciting example of possible interactions and cooperation between technological sectors both within and outside the health system.

In their scenario, most of the building blocks for learning health care systems exist but are insufficiently connected because of many weak links:

- Integration requires unprecedented levels of interoperability * and standardization;
- Implementation faces many technical and organizational challenges and raises unsolved ethical, legal, and societal issues;
- Impact on health outcomes is difficult to measure and has been poorly addressed so far.

* Interoperability is concerned with the ability of different systems, devices, applications, or products to connect and communicate in a coordinated way without any effort from the end-user. Functions related to interoperability include data access, data transmission, and cross-organizational collaboration, regardless of developer or origin [54].

3.8. Facilitating Innovation Processes

Facilitating innovation processes may require leaders who know how to combine internal innovations with the capture of new ideas from outside their organizational boundaries. For example, Bhavnani et al. [65] argued that open-access and big data analytics are often developed outside of conventional medical and clinical arenas. To access this knowledge, fundamental changes to clinical teams' internal structures and composition may be necessary. For example, successful research teams should perhaps include clinicians and team members with expertise in big data analytics, bioinformatics, technology, engineering, health care administration, business and entrepreneurship, and health care policy. A cross-cultural, cross-competence process must, therefore, be developed.

Arenas for innovation may give the drone industry and others the potential to learn from each other, as a survey by Mishra et al. [53] suggested from a drone industry perspective. Their survey concluded that manufacturers in the drone industry are not exploring emerging technologies such as IoT, AI, and AR/VR sufficiently. Consequently, they do not fully appreciate which use cases would be interesting to roll out. Mishra et al. hypothesized that the skills needed for the drone industry will include sufficient domain training for equipment providers and technical users because it is necessary to extract users' specifications and requirements to maximize the benefits of the various use cases and generate drone applications.

The extent to which users are enabled to present their needs, specifications, and requirements in early innovation phases is another issue. Cawthorne et al. [66] proposed an ethically based design of drone prototypes to examine the public acceptance of health-related drones. Mion emphasized that organizations that intend to use drones for transport must be open to change and prepared to modify their operations [56].

Johannessen et al. [44] concluded that it would require extensive research to understand how best to engage clinical and laboratory leaders and managers in actively facilitating long-term improvement processes to optimize drone transport solutions. They highlighted the relevance of well-known specialized methods that focus on looking for waste in production processes to improve workflows and create more value with less effort: the LEAN method (originating from the Toyota car manufacturing system) and Six Sigma (originating from Motorola). Although the experience of implementing such organizational processes varies from successful improvements to processes to cases in which implementing LEAN in clinical cultures has sparked challenges that are more demanding than would have been the case with the implementation of the typical methods used in the industry, the generic concepts from the domain of industrial improvement techniques should be considered [96–104].

3.9. Collaborations

The recognition that successful integrations of technology generally require collaboration between professions that need to work together may also apply to drones, as suggested by several researchers [19,51,62,67]. They all recommended that authorities, leaders, and community representatives should be involved in this perspective and pointed out the importance of engaging a broad range of users to promote the development of health-related drone applications.

A review by Hiebert et al. [19] discussed how a range of health and digital proficiencies may support sustainable integration into health care services. Community engagement should include a broad spectrum of activities throughout the design, testing, development, integration, and evaluation stages of new technology program development. This was also supported by a report from a drone project transporting blood and pathology samples between two hospitals in Switzerland [56]; in addition to the public acceptance, regulatory framework, and risk management dimensions, it was concluded that involving the top managerial level is of crucial significance.

The Health Care System Level (Landscape)

On the overall health care level, important topics may include the adoption of technology, the broad diffusion and coordination of innovations, and acknowledgment of the characteristics of change and transition processes in such complex sectors that are subject to regulations, ethical standards, and responsibilities that are far beyond the traditional industrial perspectives.

3.10. Adoption

Regarding health care's implementation of technology in general, Mion [56] suggested that effective implication of new technologies requires careful consideration of whether the solutions improve patients' overall quality of care. They considered both the policy and operational frameworks to be highly significant for adopting new technologies and avoiding obstruction. Furthermore, public acceptance and other sociocultural factors may affect the adoption of most technological innovations.

In their review paper, Hiebert et al. [19] suggested that there may be significant negative implications for the development of health care drone applications if they are driven only by those engaged in computer and software engineering, computer science, and aviation without significant input from health care researchers or professionals. They suggested that representation from the various technical disciplines should be orchestrated in close alliances with nursing, public health, medicine, paramedicine, aviation, kinesiology, and pathology.

A perspective that conceptualizes the interplay within the technology-organizationalenvironment framework (TOE framework) was depicted by Tornatzky et al. [105]. They described the innovation process as either "developing" or "using", indicating that the innovation generation and adoption processes differ considerably. In some cultures, there may be an ambition to extend new technologies to a broader innovative culture. In contrast, other cultures appear to implement new technologies mainly by fitting them to existing solutions. Therefore, the TOE framework may offer factors affecting the adoption of technologies to improve health care logistics processes when applied in a health care logistics setting. The TOE dimension related to drone implementation was discussed in the context of a logistic drone model by Johannessen et al. [44].

3.11. Diffusion/Acceleration

The MLP argues that a wider diffusion of niche innovations may only be achieved when linked to the ongoing regime and landscape processes [88]. This was the focus of a paper by Flahault et al. [64] regarding precision medicine. They discussed how better targeting of public health interventions, even on a global scale, through innovation and technology should be targeted to maximize the effectiveness and relevance of multiple topics such as the use of remote sensing data to fight vector-borne diseases; the use of large databases of genomic sequences of foodborne pathogens to identify origins of outbreaks; the use of social networks and internet search engines to track communicable diseases; the use of cell phone data in humanitarian actions; and the use of drones to deliver health care services in remote and excluded areas. Moreover, they illustrated how technological solutions may often be spin-offs of inventions initiated from specific targets, showing the value of the expansion from the niche to the landscape level.

A paper by Mateen et al. [68] suggested that education and appropriate training of personnel, instructions to ensure proper maintenance, monitoring in-flight performance, and awareness campaigns are critical factors to consider ensuring that the full potential of drone delivery systems is reached.

This is also in line with an article by Mion [56] that described vital factors from the experience of integrating drones in Switzerland. They focused on the value of integrating

such new technologies with organizations' business models, making it the ordinary way of working in the future. However, the impact of drones in the Swiss case had only a limited impact on organizational routines.

Several studies mentioned above may not have considered that future drone solutions will be unmanned autonomous vehicles. This indeed implicates the complex integration of different technology systems, but whether extensive personnel training is needed may be less obvious. The extent to which personnel resources are needed in future extensive drone traffic remains to be studied, and this may influence whether drones as a service will represent a transportation solution that must be delivered as a complete service where the provider operates all parts.

3.12. Change and Transitions in Relation to Ethics

We have hypothesized that a socio-technical approach may be helpful in the process of implementation of technology, as it highlights the importance of using technological transformation as an opportunity to improve processes by combining organizational and technological needs.

Multiple learning points from technological implementations have been described by Faramondi et al. [69] in their discussion of the interplay between different cultures and professional competencies. They argued that the digital transformation of companies is only deemed innovative and useful for creating novel business opportunities and improving efficiency if the company also improves its financial and innovative performance in general, thus illustrating the value of extended implementations of specific solutions.

In laboratory services, methodological concepts, such as sample collection decision support systems, innovations for automated robotic phlebotomy, novel sampling technologies such as fully traceable automated blood tube dispensers, and algorithms that detect sample collection errors, have been proposed to enhance patients' quality of care and the efficiency of the health care system. Whether such solutions result in an improvement of the quality of specimens and minimization of harm is still to be documented [71,80–82].

Although one of the key strategies of the MLP is to stimulate learning processes, it does not explain how to facilitate these processes. Faramondi et al. [69] and Cawthorne et al. [66] proposed different frameworks. Faramondi suggested using a "value proposition canvas" (VPC), a concept initially developed by Osterwalder et al. [106], as a part of a business model canvas. VPC is conceptualized as a framework to ensure that there is a fit between a product and its customers. The VPC is used to highlight the beneficiaries (usually multiple) of a new product/service and the gains and pains provided by the new product/service to each beneficiary or others in general concerning alternative products (if any). By adopting this canvas model, the authors intended to assess the value of a proposed system to multiple beneficiaries, distinguishing between the economic, environmental, and social aspects of the system and their respective implications. Faramondi et al. provided two illustrative examples in the form of a scenario with and without drones where the drone scenario was analyzed using the VPC and summarized in a SWOT analysis.

Cawthorne et al. [66] suggested using an ethical framework for the emerging domain of drones in health care. They assumed that, ideally, drones will be beneficial (in terms of costs, health, jobs, and environmental sustainability), do no harm (in terms of safety and security, privacy, and jobs), enhance human autonomy (trust), be just (fairly distribute benefits and risks), and be easily understandable (explicable). Based on their framework, they proposed that drones conducting transports of biological material should be identifiable in parallel to ambulances and emergency services; for example, they could be marked with a red cross, indicating that they belong to the health services.

The suggestions by Faramondi et al. and Cawthorne et al. to look beyond technology are interesting because both authors propose approaches to learn from stakeholders. The approach by Faramondi et al. presents two well-known tools that can be used to map dimensions and learn from different stakeholders. The design suggested by Cawthorne et al. may be used to build a prototype that can be used to learn about the public acceptance of health-related drones. However, neither suggestion is based on a starting point of a deeper understanding of user and institutional needs.

A potential gap in the understanding of how drones can be integrated into the health system and how long-term sustainability should be achieved was discussed by Eichleay et al. [25]. They suggested five general steps of value in this respect: drones will need to operate within regulations, stakeholders must embrace the drone concept, financial resources need to be available, human resources must be in place, and operational procedures must be developed to work effectively within existing structures.

These steps, although based on interviews in Kenya—and thus possibly reflecting perspectives different from those of systems elsewhere—seem obvious for any system, although their argument that starting conversations about potential UAV integration and interoperability of systems in early phases is key to efficient system design and may have varying degrees of relevance in the drone context.

The topic of ethics has also been discussed by Carillo-Larco et al. [70]. They asked whether the innovative and constantly evolving nature of drone use may spark ethical challenges and suggested that ethics committees need to be aware of what permissions or regulations there are to operate drones in each area to ensure that all requirements are met. This was also discussed by Cawthorne et al. [66] in their reference to Van den Hoven et al. [107] that "technical innovation can entail moral progress ... (because) it enlarges the opportunity set by changing the world in such a way that we can live by all our values". However, "new options also bring new side-effects and risks", which must be managed.

Mishra et al. [53] also raised social concerns. The operation of UAVs must be adequately regulated to protect the privacy of business organizations and individuals. The existing regulations to protect privacy may not be sufficient due to the rapid evolution of UAV technology and its increasing capabilities; therefore, there is a need to formulate further legislation to protect privacy.

Drones for Remote and Rural Services and Cost Perspectives

Among the rich literature on the cost of drone services and the use of drones for remote and last-mile delivery services, the review paper by Hiebert et al. [19] identified 20 articles that discuss the ability of drones to improve the response time of emergency services due to their ability to fly above roadways, water, and forested areas and 12 reports that discuss how drones may be used to improve access to health services in difficult-to-reach areas. These studies described four overarching topics: health applications; the benefits and costs of drones; the factors influencing use and performance in sociocultural contexts; and community engagement and sociocultural contexts as key factors in tailoring new use of drone technologies to health systems. These studies noted the importance of working with diverse stakeholders (e.g., medical and drone industries, insurance companies, pharmacies, retail outlets, entrepreneurs, legislative authorities, and other policymakers) to successfully integrate drones into health systems and pointed out that there is limited literature on how drone applications may influence patient groups and communities, raising the question of for whom drone applications in health care are being developed.

In a recent review, Nyababa et al. [91] summarized several studies of the sociotechnical debate on the drone delivery of medical supplies in Africa. They concluded that in the African context, the implementation of medical drones will revolutionize health care delivery systems, particularly for rural communities that are hard to reach during health emergencies due to poor road infrastructure, thus improving access to health care. Interestingly, they pointed out that the adoption of drones in the medical sector in Africa may be an example of what is known as "leapfrogging", that is, when developing countries skip the gradual process of technological evolution and adoption seen in developed countries and leapfrog over these gradual steps to the rapid adoption of novel devices and systems. However, they noted that the initial cost of implementation of medical drones is usually high, thus implying that the adoption of medical drones in Africa is likely to be a long-term process, as it will take a considerable time for the benefits of drone implementation to make up for the implementation costs, depending on the size of the population that will be served.

Müller et al. [18] did not study health services but discussed the MLP in relation to the mutual interdependence of transport systems during their development and the significance of the interaction of the transport systems' evolution with the socio-economic landscape. They concluded that the landscape comprises deep structural trends such as economic development and social paradigms, where several trends put pressure on the regime, slowly or spontaneously, resulting in the need for the dominant technology regime to adapt to the pressure. This techno-economic pressure is related to disruptive innovations, implying new paradigms that change labor skill profiles, demand patterns, the competitive base for companies' products, and production methods. They concluded that the landscape may generate remarkable pressure to modify the regime's innovation pathway, resulting in the inclusion of techno-economic pressures in the industry regime's incremental innovation pathway.

Zailani et al. [92] compared the costs of drones versus ambulances for transporting blood products to treat maternal obstetric hemorrhages in situations with challenging terrain and traffic flow. Their economic evaluation concluded that although drone transportation of blood products costs more compared to ambulance transport, the significantly reduced travel time offsets the cost. From an economic viewpoint, they concluded that drones are a more cost-effective and viable mode of blood product transportation, particularly during emergencies. The findings of this study add to the body of knowledge pertaining to the cost-effectiveness of drones as a vehicle for health care service delivery where delivery time is of crucial importance.

The review paper by Ling et al. [90] discussed the use of aerial drones for various purposes, such as blood delivery, medical device delivery (automated external defibrillators), and medication delivery, and concluded that aerial drones promise improved health care delivery by providing faster response times, reduced transportation costs, and improved access to medical products/services in remote and/or underserved environments. These are all factors that may be conducive to public acceptance of drones.

4. Discussion

The development of drones in health care will migrate from technological and medical environments into institutional applications for various purposes and may thus have an impact on the entire health care sector. The broad perspective of applications is illustrated by the span ranging from drone delivery of medications [74], emergency transports of blood products in maternal health care [76], and not least the extensive use of drone services to combat the COVID pandemic [75,84]. The scientific articles in our study indicate that it is necessary to build a cross-cultural approach and establish a discourse across several stakeholders to develop a sustainable future health care system. Based on our analysis of the literature, we believe that drone implementations may benefit from considering various organizational and societal topics. Although the extent of the implementation and the value created may differ across various health care systems, geographic localizations, and organizational environments, and the methods for successful implementation may differ from one context to another, we propose that some basic topics should be studied and properly understood regarding the added value that may be achieved if drones are appropriately integrated in an extended perspective.

There seems to be a solid consensus in the literature on the importance of public acceptance. Several researchers have focused on the value of communication between stakeholders across and within systems and have looked at this issue from a socio-technologic perspective, arguing that a holistic understanding of how the adoption of various drone applications may impact a community is necessary. This is a significant concept for the MLP method, which is aimed at building a universal understanding of and implementation approach to disruptive technologies. Hiebert et al. [19] found the following arguments in favor of community engagement: community understanding and buy-in, ensuring relevance, and increasing sustainability through local control. Furthermore, information about community needs can guide policymakers and decision-makers to properly integrate drones. Agency, understood as the ability of actors to intervene in and change system innovations, is another important concept in understanding the MLP framework [40].

The large diversity of opinions in the public perception of drones is illustrated by the distance between a report by Truog et al. [67] from Malawi, which described how community leaders were afraid that autonomous drones could be perceived as something unnatural and possessed, and our own research at Oslo University Hospital, where we have found that the use of drones in health care is seen positively across professional groups, ages, and locations [83].

Because community perceptions are heavily influenced by culture and prior experience, the response to the use of drones in each local system will reflect the local beliefs, practices, and attitudes of that area. Such responses should not be considered generalizable within or across countries. Governments, drone companies, and implementing partners should assess community perceptions of drones when proposing activities in new areas where attitudes and preferences are poorly understood.

The consideration of ethical questions is essential in this context and has been addressed by several authors. A review by Wang et al. [73] concerning the use of drones for humanitarian services identified several critical areas of concern, with minimizing the risks of harm and protecting privacy appearing to be the most critical points. However, they concluded that ethical conflicts may emerge in the shift from humanitarian drone programs to commercially available drone delivery services. Furthermore, such transitions may reshape issues related to data management and security, control, and responsibility.

4.1. Future Actions: From the Clinical to Institutional Level—The Proof of Concept?

Clinics are one of the most crucial loci for technological innovation. Clinicians and others with innovative mindsets try continuously to solve problems by experimenting, combining, and testing new approaches and tools. In parallel to this, we assume that the process of drone solutions begins with someone launching the idea as a way of transporting biological material. Drones promise a substantial opportunity to interact with other environments, both within and outside the health system [64]. An active innovation policy may enhance such developments, where new knowledge is explored and exploited in a positive feedback loop.

How Does the MLP Concept Apply to This Process?

As we assume drones to be implemented in a "bottom-up" process, we find the three MLP dimensions speed of change, size of the change and the four MLP phases of implementation MLP interesting to guide the process. The MLP argues that landscape pressures create "windows of opportunity" for niche technologies to emerge. However, the starting point of pressure may often come from below, from the niche of "clinical" environments that create innovations in daily practice originating from dynamics in ordinary activities and challenges, such as producing, promoting, adopting, and aligning technologies; cultivating novelties within existing regimes; enlisting users and making them available for integration into practices [89]. Nevertheless, the MLP theory may also be used to understand such processes as the creation of pressures and windows of opportunity at the landscape level.

According to the MLP, innovations are strongly influenced by existing regimes and landscapes. New technologies fit into existing regimes. However, over time, new functionalities, forms, and design options are explored. The co-evolution of form and function may lead to a wider diffusion, adaptation, and new socio-technical regime [30]. However, this process may take time and depend on previous experience with integrating technological knowledge [86]. The MLP suggests that experimental projects help work toward long-term change, fostering learning processes and institutional change. Furthermore,

while policymakers can facilitate learning and network building, niche actors may decide the process [87].

An essential element in the MLP is that technologies are introduced into social and economic systems and thus embedded in an "extended universe" where understanding the effects of innovations and changes in systems is necessary. The challenge for innovation not only rests on economic potential but also on the societal changes induced by innovative activity and environmental and social sustainability consequences. Along with this broad framing of the problem comes a need for open-minded analytical perspectives [26].

The institutional level refers to the existing system, which is a guarantor for stability with established structures, rules, and groups of actors. Such systems develop over time, which can make them resistant to change. The MLP refers to such resistance as lock-in mechanisms related to needed changes to established work processes or to the replacement of technical tools or approaches and suggests that technological innovations may divide the audience into enthusiasts and skeptics, early adopters and the fearful, and the curious and the indifferent [56]. However, from a socio-technical perspective, the challenge may lie not in people supporting or opposing the new technology but in a lack of concordance between the requirements of the technologies and the practices that people perform and have developed over time [89].

MPL stresses that the success of an innovation depends substantially on how it becomes integrated into existing systems and technologies and on the extent to which it is approved and fits within regulatory frameworks. Proper integrations developed by deep involvement enforce acceptance by users and society at large [77]. Furthermore, public acceptance and the social benefit of innovations can enable health policies.

Innovations are created through the interactions between technological, social, economic, cultural, and political aspects [77], and technological innovations may be unpredictable for many reasons. Therefore, if and to what extent drones impact structures in laboratories, whether drones in combination with other technologies have the power to transform the health care system, or to what extent in-depth organizational processes are needed to implement drones, is currently unknown and only understood in hypothetical terms.

The aforementioned interactions should also include evaluation of the acceptance of drone integration because assuming such acceptance by default may lead to a later rejection, which in the long run may impact the broader integration process and invalidate all the previous regulatory work done to ensure smooth integration [84].

The requirements created by innovations at the clinical level may spark responses at the regime level, i.e., adaptations or regulations are needed to support further developments. On the one hand, prompt up-front regulation could enhance innovations and actively drive new developments. On the other hand, a holistic perspective at the regime or landscape level may need supplementary or broader information than the specific clinical level to ensure proper interests for an extended perspective. The institution may need to engage interdisciplinary teams crucial for giving feedback on consequences beyond the initial perspective. From such a perspective, the MLP theorizes that the integration of drones and drone-related applications can be strengthened with a process across different users and levels in the system [19].

Based on the MLP principles, leaders become essential in providing arenas for collaboration, both from within and outside health systems [64,65]. Creating and maintaining linkages and networks between clinics, institutions, and health systems may be conducive to a wider diffusion of drones.

In their review of logistics research, Rejeb et al. [72] discussed how IoT in the context of logistics requires the organization to continuously adapt, engage with, and implement new technologies to enhance the performance of logistic systems. With respect to drones, they recommended that managers inspect the technical features to enable them to identify use cases that may integrate economic, social, and environmental considerations into their strategic decision-making processes.

4.2. The Health Care System Level

Health care systems at the overarching level include broader societal processes, such as political, economic, and demographic trends.

The MLP landscape topic assumes that policy forms an integral part of an innovation, either because of directly triggered changes or because policy changes may be a prerequisite for the innovation. The MLP theory assumes that continuous interaction and communication between the MLP levels are crucial to creating the forces that allow the various transitions not to end until changes in the social and technical elements are embedded in the complete system, i.e., at the user level, the production level, and the institutional and system levels. Therefore, it is helpful to understand the potential policy effects at the earliest stage to take the necessary precautions with respect to unforeseen political consequences or to prepare an argument for any policy changes that should occur. In addition, this may help identify specific issues that influence how stakeholders should be addressed to keep their expectations as realistic as possible. Furthermore, understanding the degree to which organizations need to change is essential for developing a realistic concept and ideas of the potentials and limitations of a given domain. Finally, this is an essential platform for identifying where uncertainties remain and which issues need more attention.

The MLP theory points out that policymakers have an essential role in recognizing relevant niche innovations at the earliest possible occasion to identify lock-ins and help remove them. Creating an environment for trust may contribute an additional nutrient for the diffusion of innovation. Finally, by defining common goals and timely expectations, all stakeholders may work for a shared vision for a future system.

Limitations

Our scoping review has several limitations. First, some relevant papers may not have been identified on account of a search concept that may have been too narrow. However, the search results generated in literature studies that use a structured search approach are often limited.

Furthermore, choosing socio-technical theory as our theoretical lens determined our preferences for conducting our search. We conducted searches of the academic literature in the PubMed and Scopus databases. PubMed was chosen as a medical database, Scopus, because of its broad coverage in diverse fields and its quality [72]. Google Scholar was used to snowball relevant articles, which may be a further limitation. Google Scholar's search logic is based on full-text searches and thus generates a high volume of search results. Our search strategy was based on a structured search, which may have needed to be adapted for use with Google Scholar.

Furthermore, articles that were published after we conducted the literature search in March 2021 were not included. It is also possible that more extensive results could have been obtained from the gray literature, but we chose to restrict our search to peer-reviewed scholarly journal articles.

Additionally, the analysis of the literature revealed categories that may have yielded significantly more studies. For example, a search for "public acceptance" may have generated additional relevant studies for inclusion in this review—however, this scoping review intended to provide a more general overview. Thus, although its conclusions cannot be generalized, this review may spark a discussion about the socio-technical considerations for integrating drones.

Our objective in this study was to look for the value of taking a socio-technical approach. However, our search strategy did not identify any studies that used the keyword "socio-technical". This may be because socio-technical thinking is not very widespread. The MLP can be used to address this research gap.

5. Conclusions

This paper is based on our research question regarding how knowledge of sociotechnical theory may support an extended focus on implementing drones into health care systems.

Based on a scoping review and supplementing literature, we found evidence in the literature that building a cross-cultural approach and establishing a discourse across several stakeholders is of importance to developing a sustainable future health system. Furthermore, from a socio-technological perspective, ethical considerations and public acceptance are topics highly relevant to the implementation of drones.

Several authors have addressed ethical topics. In the transition from humanitarian drone projects to commercially available service offerings, ethical issues may be seen concerning data management and security, control, and responsibility.

In addition, the topic of public acceptance has been raised by several authors. Community engagement is helpful for buy-in and ensuring the relevance of the innovation. Because acceptance is deeply influenced by culture and prior experience, the acceptance of drones should be evaluated.

Furthermore, addressing community engagement can ensure an agency in terms of the ability of actors to intervene in system change. Experimental projects can cultivate learning processes and institutional change. Policymakers can facilitate learning and network building. An active innovation policy can prompt up-front regulation, enhance innovations, and actively drive new developments.

Systemic knowledge about privacy is needed to obtain a social consensus on the capabilities of drones, thereby also guiding policymakers and decision-makers. Niche actors may decide on the process.

A holistic perspective and the regime or landscape level may need supplementary or broader information than the specific clinical level to ensure that the proper interests are considered in an extended perspective. The institution may need to engage interdisciplinary teams, which are crucial for giving feedback about consequences beyond the initial perspective.

In this process of transition, leaders are essential for creating and maintaining linkages and networks between clinics, institutions, and health systems. The MLP shows that continuous interaction and communication between the multiple levels create forces that secure the various transitions until changes in the social and technical elements are embedded in the complete system. Furthermore, the MLP is helpful for understanding potential policy effects at the earliest stage to take the necessary precautions against unforeseen political consequences and prepare arguments regarding the policy changes that should occur.

Author Contributions: H.E.C.: conceptualization, methodology, validation, analysis, writing original draft preparation, visualization; K.-A.J.: conceptualization, analysis, writing—review and editing, supervision, project administration. All authors have read and agreed to the published version of the manuscript.

Funding: The current work has been funded by the Norwegian Research Council under Grant 282207/2018.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Acknowledgments: We thank Martina Keitsch and Ole Andreas Alsos from the Norwegian University of Technology and Science, and Erik Fosse from the Intervention Centre at the National Hospital for helpful comments.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Casalino, L.; Gillies, R.R.; Shortell, S.M.; Schmittdiel, J.A.; Bodenheimer, T.; Robinson, J.C.; Rundall, T.; Oswald, N.; Schauffler, H.; Wang, M.C. External incentives, information technology, and organized processes to improve health care quality for patients with chronic diseases. *JAMA J. Am. Med. Assoc.* 2003, 289, 434–441. [CrossRef] [PubMed]
- Sturesson, L.; Groth, K. Effects of the digital transformation: Qualitative study on the disturbances and limitations of using video visits in outpatient care. J. Med. Internet Res. 2018, 20, e221. [CrossRef] [PubMed]
- Vogt, C.; Gersch, M.; Spies, C.; Bengler, K. Digital Transformation in Healthcare: How the Potential of Digital Health Is Tackled to Transform the Care Process of Intensive Care Patients Across All Healthcare Sectors. In *Digitalization Cases*; Springer: Cham, Switzerland, 2018. [CrossRef]
- Susanto, H.; Chen, C.K. Information and Communication Emerging Technology: Making Sense of Healthcare Innovation. In Internet of Things and Big Data Technologies for Next Generation Healthcare; Studies in Big Data; Springer International Publishing: Cham, Switzerland, 2017; pp. 229–250. [CrossRef]
- 5. Yoon, S.N.; Lee, D. Artificial intelligence and robots in healthcare: What are the success factors for technology-based service encounters? *Int. J. Healthc. Manag.* 2019, *12*, 218–225. [CrossRef]
- Amukele, T.K.; Sokoll, L.J.; Pepper, D.; Howard, D.P.; Street, J. Can unmanned aerial systems (drones) be used for the routine transport of chemistry, hematology, and coagulation laboratory specimens? *PLoS ONE* 2015, *10*, e0134020. [CrossRef] [PubMed]
- Claesson, A.; Fredman, D.; Svensson, L.; Ringh, M.; Hollenberg, J.; Nordberg, P.; Rosenqvist, M.; Djarv, T.; Osterberg, S.; Lennartsson, J.; et al. Unmanned aerial vehicles (drones) in out-of-hospital-cardiac-arrest. *Scand. J. Trauma Resusc. Emerg. Med.* 2016, 24, 124. [CrossRef]
- 8. Haidari, L.A.; Brown, S.T.; Ferguson, M.; Bancroft, E.; Spiker, M.; Wilcox, A.; Ambikapathi, R.; Sampath, V.; Connor, D.L.; Lee, B.Y. The economic and operational value of using drones to transport vaccines. *Vaccine* **2016**, *34*, 4062–4067. [CrossRef]
- 9. Amukele, T.K.; Hernandez, J.; Snozek, C.L.H.; Wyatt, R.G.; Douglas, M.; Amini, R.; Street, J. Drone Transport of Chemistry and Hematology Samples over Long Distances. *Am. J. Clin. Pathol.* **2017**, *148*, 427–435. [CrossRef]
- 10. Hassanalian, M.; Abdelkefi, A. Classifications, applications, and design challenges of drones: A review. *Prog. Aerosp. Sci.* 2017, 91, 99–131. [CrossRef]
- Kim, S.J.; Lim, G.J.; Cho, J.; Côté, M.J. Drone-Aided Healthcare Services for Patients with Chronic Diseases in Rural Areas. J. Intell. Robot. Syst. 2017, 88, 163–180. [CrossRef]
- 12. Rosser, J.C.; Vignesh, V.; Terwilliger, B.A.; Parker, B.C. Surgical and medical applications of drones: A comprehensive review. JSLS J. Soc. Laparoendosc. Surg. 2018, 22, e2018.00018. [CrossRef]
- Scalea, J.R.; Restaino, S.; Scassero, M.; Blankenship, G.; Bartlett, S.T.; Wereley, N. An Initial Investigation of Unmanned Aircraft Systems (UAS) and Real-Time Organ Status Measurement for Transporting Human Organs. *IEEE J. Transl. Eng. Health Med.* 2018, 6, 1–7. [CrossRef] [PubMed]
- Johannessen, K.A.; Wear, N.K.S.; Toska, K.; Hansbo, M.; Berg, J.P.; Fosse, E. Pathologic Blood Samples Tolerate Exposure to Vibration and High Turbulence in Simulated Drone Flights, but Plasma Samples Should be Centrifuged after Flight. *IEEE J. Transl. Eng. Health Med.* 2021, 9, 1–10. [CrossRef] [PubMed]
- 15. Perlee, D.; van der Steege, K.H.; den Besten, G. The effect of drone transport on the stability of biochemical, coagulation and hematological parameters in healthy individuals. *Clin. Chem. Lab. Med.* **2021**, *59*, 1772–1776. [CrossRef] [PubMed]
- Amukele, T.K.; Street, J.; Carroll, K.; Miller, H.; Zhang, S.X. Drone transport of microbes in blood and sputum laboratory specimens. J. Clin. Microbiol. 2016, 54, 2622–2625. [CrossRef]
- Ochieng, W.O.; Ye, T.; Scheel, C.; Lor, A.; Saindon, J.; Yee, S.L.; Meltzer, M.I.; Kapil, V.; Karem, K. Uncrewed aircraft systems versus motorcycles to deliver laboratory samples in west Africa: A comparative economic study. *Lancet Glob. Health* 2020, *8*, e143–e151. [CrossRef]
- Müller, S.; Rudolph, C.; Janke, C. Drones for Last Mile Logistics: Baloney or Part of the Solution? Elsevier B.V.: Amsterdam, The Netherlands, 2019; Volume 41, pp. 73–87. [CrossRef]
- Hiebert, B.; Nouvet, E.; Jeyabalan, V.; Donelle, L. The Application of Drones in Healthcare and Health-Related Services in North America: A Scoping Review. Drones 2020, 4, 30. [CrossRef]
- Homier, V.; Brouard, D.; Nolan, M.; Roy, M.A.; Pelletier, P.; McDonald, M.; de Champlain, F.; Khalil, E.; Grou-Boileau, F.; Fleet, R. Drone versus ground delivery of simulated blood products to an urban trauma center: The Montreal Medi-Drone pilot study. J. Trauma Acute Care Surg. 2021, 90, 515–521. [CrossRef]
- Osakwe, C.N.; Hudik, M.; Říha, D.; Stros, M.; Ramayah, T. Critical factors characterizing consumers' intentions to use drones for last-mile delivery: Does delivery risk matter? J. Retail. Consum. Serv. 2021, 65, 102865. [CrossRef]
- Tatham, P.; Stadler, F.; Murray, A.; Shaban, R.Z. Flying maggots: A smart logistic solution to an enduring medical challenge. J. Humanit. Logist. Supply Chain. Manag. 2017, 7, 172–193. [CrossRef]
- 23. Moshref-Javadi, M.; Winkenbach, M. Applications and Research avenues for drone-based models in logistics: A classification and review. *Expert Syst. Appl.* 2021, 177, 114854. [CrossRef]
- 24. Gruson, D. New Solutions for the Sample Transport and Results Delivery: A Digital Lab. eJIFCC 2018, 29, 210–214. [PubMed]
- Eichleay, M.; Evens, E.; Stankevitz, K.; Parker, C. Using the Unmanned Aerial Vehicle Delivery Decision Tool to Consider Transporting Medical Supplies via Drone. *Glob. Health Sci. Pract.* 2019, 7, 500. [CrossRef] [PubMed]

- Smith, A.; Voß, J.P.; Grin, J. Innovation studies and sustainability transitions: The allure of the multi-level perspective and its challenges. *Res. Policy* 2010, 39, 435–448. [CrossRef]
- Hekkert, M.P.; Suurs, R.A.A.; Negro, S.O.; Kuhlmann, S.; Smits, R.E.H.M. Functions of innovation systems: A new approach for analysing technological change. *Technol. Forecast. Soc. Chang.* 2007, 74, 413–432. [CrossRef]
- Christensen, C.M. The Innovator's Dilemma: When New Technologies Cause Great Firms to Fail; The Management of Innovation and Change Series; Harvard Business School Press: Boston, MA, USA, 1997.
- Anderson, P.; Tushman, M.L. Technological Discontinuities And Dominant Designs: A Cyclical Model of Technological Change. Adm. Sci. Q. 1990, 35, 604–633. [CrossRef]
- Geels, F.W. Technological Transitions and System Innovations: A Co-Evolutionary and Socio-Technical Analysis; Edward Elgar Publishing: Cheltenham, UK, 2005.
- 31. Geels, F.W. Processes and patterns in transitions and system innovations: Refining the co-evolutionary multi-level perspective. *Technol. Forecast. Soc. Chang.* 2005, 72, 681–696. [CrossRef]
- 32. Geels, F.W.; Schot, J. Typology of sociotechnical transition pathways. Res. Policy 2007, 36, 399-417. [CrossRef]
- Haula, K.; Agbozo, E. A systematic review on unmanned aerial vehicles in Sub-Saharan Africa: A socio-technical perspective. *Technol. Soc.* 2020, 63, 101357. [CrossRef]
- Ayamga, M.; Akaba, S.; Nyaaba, A.A. Multifaceted applicability of drones: A review. *Technol. Forecast. Soc. Chang.* 2021, 167, 120677. [CrossRef]
- Ghelichi, Z.; Gentili, M.; Mirchandani, P.B. Logistics for a fleet of drones for medical item delivery: A case study for Louisville, KY. Comput. Oper. Res. 2021, 135, 105443. [CrossRef]
- Kuru, K. Planning the future of smart cities with swarms of fully autonomous unmanned aerial vehicles using a novel framework. IEEE Access 2021, 9, 6571–6595. [CrossRef]
- Geels, F.W. Major system change through stepwise reconfiguration: A multi-level analysis of the transformation of American factory production (1850–1930). *Technol. Soc.* 2006, 28, 445–476. [CrossRef]
- Geels, F.W.; Sovacool, B.K.; Schwanen, T.; Sorrell, S. Sociotechnical transitions for deep decarbonization. *Science* 2017, 357, 1242–1244. [CrossRef] [PubMed]
- Geels, F.; Raven, R. Non-linearity and Expectations in Niche-Development Trajectories: Ups and Downs in Dutch Biogas Development (1973–2003). *Technol. Anal. Strateg. Manag.* 2006, 18, 375–392. [CrossRef]
- Gaziulusoy, A.İ. System Innovation for Sustainability: A Scenario Method and a Workshop Process for Product Development Teams. Ph.D. Thesis (Monograph), University of Auckland, Auckland, New Zealand, 2010.
- 41. Tuama, D.Ó. Ripples through the city: Understanding the processes set in motion through embedding a public bike sharing scheme in a city. *Res. Transp. Bus. Manag.* 2015, *15*, 15–27. [CrossRef]
- 42. Geels, F.W. Socio-technical transitions to sustainability: A review of criticisms and elaborations of the Multi-Level Perspective. *Curr. Opin. Environ. Sustain.* 2019, 39, 187–201. [CrossRef]
- Rotmans, J.; Kemp, R.P.M.; van Asselt, M. More Evolution than Revolution: Transition Management in Public Policy. Foresight Lond. J. Insur. Risk Manag. 2001, 3, 15–31. [CrossRef]
- Johannessen, K.A.; Comtet, H.; Fosse, E. A Drone Logistic Model for Transporting the Complete Analytic Volume of a Large-Scale University Laboratory. Int. J. Environ. Res. Public Health 2021, 18, 4580. [CrossRef]
- Prayag, G.; Ozanne, L.K. A systematic review of peer-to-peer (P2P) accommodation sharing research from 2010 to 2016: Progress and prospects from the multi-level perspective. J. Hosp. Mark. Manag. 2018, 27, 649–678. [CrossRef]
- Munn, Z.; Peters, M.D.J.; Stern, C.; Tufanaru, C.; McArthur, A.; Aromataris, E. Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach. *BMC Med. Res. Methodol.* 2018, 18, 143. [CrossRef]
- Arksey, H.; O'Malley, L. Scoping studies: Towards a methodological framework. Int. J. Soc. Res. Methodol. 2005, 8, 19–32. [CrossRef]
- 48. Lazar, J. Research Methods in Human-Computer Interaction, 2nd ed.; Morgan Kaufmann: Cambridge, MA, USA, 2017.
- Hsieh, H.F.; Shannon, S.E. Three Approaches to Qualitative Content Analysis. *Qual. Health Res.* 2005, 15, 1277–1288. [CrossRef] [PubMed]
- 50. Mayring, P. Qualitative content analysis: Demarcation, varieties, developments. Forum Qual. Soc. Res. 2019, 20, 16. [CrossRef]
- Ferreras, L.E. iTRANS: Proactive ITS Based on Drone Technology to Solve Urban Transportation Challenge. In Disrupting Mobility: Impacts of Sharing Economy and Innovative Transportation on Cities; Meyer, G., Shaheen, S., Eds.; Springer International Publishing: Cham, Switzerland, 2017; pp. 323–333. [CrossRef]
- Vandenberg, O.; Durand, G.; Hallin, M.; Diefenbach, A.; Gant, V.; Murray, P.; Kozlakidis, Z.; van Belkum, A.; Vandenberg, O. Consolidation of Clinical Microbiology Laboratories and Introduction of Transformative Technologies. *Clin. Microbiol. Rev.* 2020, 33, e00057-19. [CrossRef]
- Mishra, D.; Natalizio, E. A survey on cellular-connected UAVs: Design challenges, enabling 5G/B5G innovations, and experimental advancements. *Comput. Netw.* 2020, 182, 107451. [CrossRef]
- Khisa, S.; Moh, S. Medium Access Control Protocols for the Internet of Things Based on Unmanned Aerial Vehicles: A Comparative Survey. Sensors 2020, 20, 5586. [CrossRef]
- 55. Syed, F.; Gupta, S.K.; Alsamhi, S.; Rashid, M.; Liu, X. A survey on recent optimal techniques for securing unmanned aerial vehicles applications. *Trans. Emerg. Telecommun. Technol.* **2020**, *32*, e4133. [CrossRef]

- 56. Mion, F.U. Flying drones to exchange lab samples service innovation by the Swiss multisite hospital EOC. In *Service Design and Service Thinking in Healthcare and Hospital Management*; Springer: Cham, Switzerland, 2019. [CrossRef]
- Zègre-Hemsey, J.; Bogle, B.; Cunningham, C.; Snyder, K.; Rosamond, W. Delivery of Automated External Defibrillators (AED) by Drones: Implications for Emergency Cardiac Care. *Curr. Cardiovasc. Risk Rep.* 2018, 12, 1–5. [CrossRef]
- Poljak, M.; Šterbenc, A. Use of drones in clinical microbiology and infectious diseases: Current status, challenges and barriers. Clin. Microbiol. Infect. 2020, 26, 425–430. [CrossRef]
- 59. Van de Voorde, P.; Gautama, S.; Momont, A.; Ionescu, C.M.; De Paepe, P.; Fraeyman, N. The drone ambulance [A-UAS]: Golden bullet or just a blank? *Resuscitation* 2017, 116, 46–48. [CrossRef]
- 60. Shaw, B.; Chisholm, O. Creeping Through the Backdoor: Disruption in Medicine and Health. *Front. Pharmacol.* **2020**, *11*, 818. [CrossRef] [PubMed]
- 61. Balasingam, M. Drones in medicine—The rise of the machines. Int. J. Clin. Pract. 2017, 71, e12989. [CrossRef] [PubMed]
- Braun, J.; Gertz, S.D.; Furer, A.; Bader, T.; Frenkel, H.; Chen, J.; Glassberg, E.; Nachman, D. The promising future of drones in prehospital medical care and its application to battlefield medicine. *J. Trauma Acute Care Surg.* 2019, 87, S28–S34. [CrossRef] [PubMed]
- Nentwich, M.; Hórvath, D.M. Delivery Drones from a Technology Assessment Perspective; Overview Report; Institute for Technology Assessment of the Austrian Academy of Sciences (ITA): Vienna, Austria, 2018.
- Flahault, A.; Geissbuhler, A.; Guessous, I.; Guérin, P.J.; Bolon, I.; Salathé, M.; Escher, G. Precision global health in the digital age. Swiss Med. Wkly. 2017, 147, w14423. [CrossRef]
- Bhavnani, S.P.; Munoz, D.; Bagai, A. Data Science in Healthcare: Implications for Early Career Investigators. Circ. Cardiovasc. Qual. Outcomes 2016, 9, 683–687. [CrossRef]
- Cawthorne, D.; Wynsberghe, A.R. From HealthDrone to FrugalDrone: Value-Sensitive Design of a Blood Sample Transportation Drone. In Proceedings of the 2019 IEEE International Symposium on Technology and Society (ISTAS), Medford, Boston, MA, USA, 15–16 November 2019; pp. 1–7. [CrossRef]
- Truog, S.; Maxim, L.; Matemba, C.; Blauvelt, C.; Ngwira, H.; Makaya, A.; Moreira, S.; Lawrence, E.; Ailstock, G.; Weitz, A.; et al. Insights Before Flights: How Community Perceptions Can Make or Break Medical Drone Deliveries. *Drones* 2020, *4*, 51. [CrossRef]
- Mateen, F.J.; Leung, K.H.B.; Vogel, A.C.; Cissé, A.F.; Chan, T.C.Y. A drone delivery network for antiepileptic drugs: A framework and modelling case study in a low-income country. *Trans. R. Soc. Trop. Med. Hyg.* 2020, 114, 308–314. [CrossRef]
- Faramondi, L.; Oliva, G.; Ardito, L.; Crescenzi, A.; Caricato, M.; Tesei, M.; Muda, A.O.; Setola, R. Use of drone to improve healthcare efficiency and sustainability. In Proceedings of the 2020 43rd International Convention on Information, Communication and Electronic Technology (MIPRO), Opatija, Croatia, 28 September–2 October 2020; pp. 1783–1788.
- Carrillo-Larco, R.M.; Moscoso-Porras, M.; Taype-Rondan, A.; Ruiz-Alejos, A.; Bernabe-Ortiz, A. The use of unmanned aerial vehicles for health purposes: A systematic review of experimental studies. *Glob. Health Epidemiol. Genom.* 2018, 3, e13. [CrossRef]
- Greaves, R.F.; Bernardini, S.; Ferrari, M.; Fortina, P.; Gouget, B.; Gruson, D.; Lang, T.; Loh, T.P.; Morris, H.A.; Park, J.Y.; et al. Key questions about the future of laboratory medicine in the next decade of the 21st century: A report from the IFCC-Emerging Technologies Division. *Clin. Chim. Acta* 2019, 495, 570–589. [CrossRef]
- 72. Rejeb, A.; Rejeb, K.; Simske, S.J.; Treiblmaier, H. Drones for supply chain management and logistics: A review and research agenda. *Int. J. Logist.* 2021, 1–24. [CrossRef]
- Wang, N.; Christen, M.; Hunt, M. Ethical Considerations Associated with "Humanitarian Drones": A Scoping Literature Review. Sci. Eng. Ethics 2021, 27, 51. [CrossRef] [PubMed]
- Lin, C.A.; Shah, K.; Mauntel, L.C.C.; Shah, S.A. Drone delivery of Medications: Review of the landscape and legal considerations. *Am. J. Health-Syst. Pharm.* 2018, 75, 153–158. [CrossRef] [PubMed]
- 75. Oigbochie, A.E.; Odigie, E.; Adejumo, B.I.G. Importance of Drones in Healthcare Delivery Amid a Pandemic: Current and Generation Next Application. *Open J. Med. Res.* **2021**, *2*, 1–13. [CrossRef]
- Zailani, M.A.H.; Sabudin, R.Z.A.R.; Rahman, R.A.; Saiboon, I.M.; Ismail, A.; Mahdy, Z.A. Drone for medical products transportation in maternal healthcare: A systematic review and framework for future research. *Medicine* 2020, 99, e21967. [CrossRef]
- Maclaine Pont, P.; van Est, R.; Deuten, J. Shaping Socio-Technical Innovation through Policy: Essay Commissioned by the Department of Knowledge, Innovation and Strategy of the Dutch Ministry of Infrastructure and the Environment; Rathenau Institute: The Hague, The Netherlands, 2016.
- Federal Aviation Administration. U.S. Transportation Secretary Elaine L. Chao Announces FAA Certification of Commercial Package Delivery. 2019. Available online: https://www.faa.gov/newsroom/us-transportation-secretary-elaine-l-chao-announcesfaa-certification-commercial-package (accessed on 27 November 2021).
- Lavallée, C. The EU Policy for Civil Drones: The Challenge of Governing Emerging Technologies. 2019. Available online: https: //brussels-school.be/publications/policy-briefs/eu-policy-civil-drones-challenge-governing-emerging-technologies (accessed on 27 November 2021).
- Loh, T.P.; Saw, S.; Chai, V.; Sethi, S.K. Impact of phlebotomy decision support application on sample collection errors and laboratory efficiency. *Clin. Chim. Acta* 2011, 412, 393–395. [CrossRef] [PubMed]
- Piva, E.; Tosato, F.; Plebani, M. Pre-analytical phase: The automated ProTube device supports quality assurance in the phlebotomy process. *Clin. Chim. Acta* 2015, 451, 287–291. [CrossRef]

- Balter, M.L.; Chen, A.I.; Fromholtz, A.; Gorshkov, A.; Maguire, T.J.; Yarmush, M.L. System design and development of a robotic device for automated venipuncture and diagnostic blood cell analysis. In Proceedings of the 2016 IEEE/RSJ International Conference on Intelligent Robots and Systems (IROS), Daejeon, Korea, 9–14 October 2016; Volume 2016, pp. 514–520. [CrossRef]
- Comtet, H.E.; Johannessen, K.A. The Moderating Role of Pro-Innovative Leadership and Gender as an Enabler for Future Drone Transports in Healthcare Systems. Int. J. Environ. Res. Public Health 2021, 18, 2637. [CrossRef]
- Martins, B.O.; Lavallée, C.; Silkoset, A. Drone Use for COVID-19 Related Problems: Techno-solutionism and its Societal Implications. *Glob. Policy* 2021, 12, 603–612. [CrossRef]
- Öztekin, E.E.; Gaziulusoy, A.İ. Designing Transitions Bottom-up: The agency of design in formation and proliferation of niche practices. *Des. J.* 2019, 22, 1659–1674. [CrossRef]
- Cohen, W.M.; Levinthal, D.A. Absorptive Capacity: A New Perspective on Learning and Innovation. Adm. Sci. Q. 1990, 35, 128–152. [CrossRef]
- Geels, F.W.; Eames, M.; Steward, F.; Monaghan, A. The Feasibility of Systems Thinking in Sustainable Consumption and Production Policy: A Report to the Department for Environment, Food and Rural Affairs; Report; Defra: London, UK, 2008.
- Schot, J.; Geels, F.W. Strategic niche management and sustainable innovation journeys: Theory, findings, research agenda, and policy. *Technol. Anal. Strateg. Manag.* 2008, 20, 537–554. [CrossRef]
- Laakso, S.; Aro, R.; Heiskanen, E.; Kaljonen, M. Reconfigurations in sustainability transitions: A systematic and critical review. Sustain. Sci. Pract. Policy 2021, 17, 15–31. [CrossRef]
- 90. Ling, G.; Draghic, N. Aerial drones for blood delivery. Transfusion 2019, 59, 1608–1611. [CrossRef] [PubMed]
- 91. Nyaaba, A.A.; Ayamga, M. Intricacies of medical drones in healthcare delivery: Implications for Africa. *Technol. Soc.* 2021, 66, 101624. [CrossRef]
- Zailani, M.A.; Azma, R.Z.; Aniza, I.; Rahana, A.R.; Ismail, M.S.; Shahnaz, I.S.; Chan, K.S.; Jamaludin, M.; Mahdy, Z.A. Drone versus ambulance for blood products transportation: An economic evaluation study. *BMC Health Serv. Res.* 2021, 21, 1308. [CrossRef]
- Schierbeck, S.; Hollenberg, J.; Nord, A.; Svensson, L.; Nordberg, P.; Ringh, M.; Forsberg, S.; Lundgren, P.; Axelsson, C.; Claesson, A. Automated external defibrillators delivered by drones to patients with suspected out-of-hospital cardiac arrest. *Eur. Heart J.* 2021, 42, ehab724.0656. [CrossRef]
- Murphy, K.; Jain, N. Riding The Disruption Wave In Healthcare. 2018. Available online: https://www.forbes.com/sites/ baininsights/2018/05/01/riding-the-disruption-wave-in-healthcare/ (accessed on 22 November 2021).
- 95. Floreano, D.; Wood, R.J. Science, technology and the future of small autonomous drones. Nature 2015, 521, 460–466. [CrossRef]
- 96. Radnor, Z.; Walley, P. Learning to walk before we try to run: Adapting lean for the public sector. *Public Money Manag.* 2008, 28, 13–20.
- Casey, J.T.; Brinton, T.S.; Gonzalez, C.M. Utilization of lean management principles in the ambulatory clinic setting. Nat. Clin. Pract. Urol. 2009, 6, 146. [CrossRef]
- Eller, A. Rapid assessment and disposition: Applying LEAN in the emergency department. J. Healthc. Qual. 2009, 31, 17–22. [CrossRef]
- Waldhausen, J.H.; Avansino, J.R.; Libby, A.; Sawin, R.S. Application of lean methods improves surgical clinic experience. J. Pediatr. Surg. 2010, 45, 1420–1425. [CrossRef] [PubMed]
- Chiarini, A.; Bracci, E. Implementing lean six sigma in healthcare: Issues from Italy. Public Money Manag. 2013, 33, 361–368. [CrossRef]
- Bhat, S.; Gijo, E.V.; Jnanesh, N.A. Application of Lean Six Sigma methodology in the registration process of a hospital. Int. J. Product. Perform. Manag. 2014, 63, 613–643. [CrossRef]
- Costa, L.B.M.; Godinho Filho, M. Lean healthcare: Review, classification and analysis of literature. Prod. Plan. Control 2016, 27, 823–836. [CrossRef]
- Moraros, J.L.; Nwankwo, C.M. Lean interventions in healthcare: Do they actually work? A systematic literature review. Int. J. Qual. Health Care 2016, 28, 150–165. [CrossRef] [PubMed]
- 104. Yadav, G.; Desai, T.N. Lean Six Sigma: A categorized review of the literature. Int. J. Lean Six Sigma 2016, 7, 2–24. [CrossRef]
- Tornatzky, L.G.; Fleischer, M.; Chakrabarti, A.K. The Processes of Technological Innovation; Issues in Organization and Management Series; Lexington Books: Lexington, MA, USA, 1990.
- 106. Osterwalder, A.; Osterwalder, A.; Pigneur, Y.; Bernarda, G.; Smith, A.; Papadakos, T. Value Proposition Design: How to Create Products and Services Customers Want; Strategyzer Series; WILEY: New York, NY, USA, 2015.
- den Hoven, J.V.; Lokhorst, G.J.; de Poel, I.V. Engineering and the Problem of Moral Overload. Sci. Eng. Ethics 2011, 18, 143–155.
 [CrossRef]

Paper 2

H. E. Comtet and K.-A. Johannessen, "The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems," eng, *International Journal of Environmental Research and Public Health*, vol. 18, no. 5, pp. 1–15, 2021, ISSN: 1661-7827





Article The Moderating Role of Pro-Innovative Leadership and Gender as an Enabler for Future Drone Transports in Healthcare Systems

Hans E. Comtet ^{1,2,*} and Karl-Arne Johannessen ¹

- ¹ The Intervention Centre, Oslo University Hospital, 0372 Oslo, Norway; karajoha@online.no
- ² The Department of Design, Norwegian University of Science and Technology, 7491 Trondheim, Norway
- * Correspondence: hanscomtet@gmail.com

Abstract: Drones have been suggested as an emerging technology that has the potential to disrupt and improve healthcare. The attitude among healthcare workers towards the use of drones is important for its successful implementation. Our aim was to examine individual and institutional variables among employees that may be relevant for the successful implementation of drones. This study used a self-administered questionnaire to investigate the expectations and beliefs among 400 employees at three Norwegian healthcare organizations regarding the future role of drones in the provision of healthcare. The results showed that the use of drones in healthcare is positively perceived across professional groups, age, and location. Working in an innovative environment, having experienced previous technological change in one's working environment, and having leadership that supports new ideas were identified as drivers of individual beliefs regarding the use of drones as an innovative solution in future healthcare services. Men had significantly higher scores than women, and this was associated with reporting innovative leadership. This may indicate that a future implementation of drone usage should focus on local system environments and may depend on the presence of innovative leadership. Our results are harvested from a developed health care system and should be applicable for similar technologically advanced systems where the full potential of drone solutions may benefit from the integration of drones into the overall socio-technical system.

Keywords: drones; unmanned aerial vehicle (UAV); healthcare; transport; socio-technical system; innovation

1. Introduction

Technological developments are enabling new solutions and challenging how we deliver existing services to an extent that was unimaginable only a few years ago [1]. Healthcare is an area that is facing large changes with respect to new treatment modalities, automatization, robotization, artificial intelligence, and innovations in existing methods aiming to individually tailor services and treatment. Technology has been suggested as one remedy that may alleviate the surging costs of healthcare. Demographic change, increased multimorbidity, and an increasing prevalence of lifestyle-related diseases in developed healthcare systems are all cost-driving factors [2]. It is also expected that technology may enable new solutions for the provision of services in less developed healthcare systems [3,4].

Unmanned aerial vehicles (hereafter referred to as drones) are one of several emerging technologies [5,6] that may contribute to this perspective. In healthcare, possible applications of drones have been investigated and include search-and-rescue operations in natural disasters, the delivery of vaccines and medicines to rural areas and healthcare technology devices in emergency situations, and the rapid transport of blood samples and organs [7–14].

Despite the challenges related to meteorological conditions, transport safety and security, and the short flight range, many researchers foresee drones as a transportation solution that could be viable within a few years [14–18]. The European Union expects that a fleet of 100,000 drones will be used for transport purposes by 2050 [19].



Citation: Comtet, H.E.; Johannessen, K.-A. The Moderating Role of Pro-Innovative Leadership and Gender as an Enabler for Future Drone Transports in Healthcare Systems. Int. J. Environ. Res. Public Health 2021, 18, 2637. https://doi.org/10.3390/ijerph18052637

Academic Editor: Rodney P Jones

Received: 22 January 2021 Accepted: 23 February 2021 Published: 5 March 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The institutional and clinical context in which drones may be a useful medium for the transport of biological material has not been extensively investigated, and whether the introduction of drones as a transport solution in healthcare will be a significant gain, remains unknown. Extreme demands for safety and uptime should be anticipated, and drones are still in an early innovation phase [20]. The speed of implementation as a transport solution may depend on differing sectors and purposes. Although multiple studies have demonstrated the proof of concept with respect to drone applications for multiple health care purposes [7–14], varying levels of momentum in the implementation of drones for high-volume and time-intensive transport in healthcare may be expected [8,21–23].

Furthermore, whether the implementation of drones for transport in healthcare systems will mainly be based on technical drone requirements alone or whether it will require specific organizational or system-related modifications may need further exploration [24]. As yet, the main focus of drones has been on their potential as airborne vehicles that can bypass ground-based traffic congestion [18] and provide accessibility for services in rural and remote locations [25,26]. It remains unclear whether such benefits and innovations can also be achieved for different transport options related to healthcare. This may concern organizational culture, logistics, and system improvements.

Although the implementation of drone solutions must be initiated and achieved at the local clinical or institutional level, an overall systemic perspective may contribute to understanding the scope of the necessary changes [3,27]. The impact of future regulations and the testing and use of drones [28], as well as interests from various stakeholders [29,30], may be relevant. The successful integration of drones may, therefore, benefit from a holistic socio-technical perspective that is adapted to healthcare cultures and systems [31,32].

The concept of "socio-technical systems" has been used to convey that the success of introducing new technologies does not depend only on the technical product itself, but also on overall policies, user experiences, infrastructure, and existing organizational structures [33]. The relevance of organizational structures and expectations related specifically to the use of drones in hospital systems has been studied by Krey et al. [34] and others [4,35–37]. These authors agree that this is an important issue requiring further investigation.

In particular, there is limited study on whether drone solutions can be achieved in existing logistics and organizational patterns that are found in specific settings, such as high-volume transport across large healthcare institutions or time-critical transport that requires an extremely high and regular uptime. Optimal service design often starts by including stakeholders to shape the best environment for new solutions to strengthen the innovation process [38–40]. In this context, the attitudes and culture among employees in hospitals that are planning to implement drones as a future transport option need to be accepting of this new technology.

This study focused on three Norwegian healthcare organizations that are building new hospitals by 2030 and are considering drone transport as a major part of their future transport logistics.

The largest hospital is Oslo University Hospital (OUS), consisting of four hospitals located throughout the Oslo area: The National Hospital (Rikshospitalet with local, regional, and national services), Ullevål University Hospital (local, regional, and national services), Radiumhospitalet (a specialized cancer hospital), and Aker University Hospital (a local, central hospital). In 2019, OUS had a total patient activity of 94,000 hospitalizations, 45,000 day care treatments, and 874,000 outpatient consultations. The hospital has 24,000 employees, and patient treatment takes place at more than 40 locations within a distance of 20 km. OUS is a large hospital by European standards, providing services that span from local hospital treatment to advanced specialized services and transplantations. The hospital has a structure that is well suited for drone transport across institutional locations.

The Central Hospital of Innlandet services a 52,072 km² area (an area slightly larger than Denmark) and covers 42 local communities with a total of 368,000 inhabitants. The hospital has 8700 employees, with patient treatment taking place at 40 primary locations and

25 emergency locations. This organization is planning to have a centralized hospital structure by 2030, with one main central institution and three additional institutional locations. The hospital is currently exploring the use of drone services for conducting primary logistics across its institutional locations and between 80 local healthcare centers (i.e., general practitioners).

The Central Hospital of Vestre Viken is located west of Oslo and provides health services to 26 communities with 500,000 inhabitants. This hospital services a nearly 20,000 km² area with a maximum distance of 180 km. The hospital has 9500 employees in total and institutions at five different locations. This organization is planning a new main hospital in the city of Drammen and is considering drone service as a future transport solution.

Our main goal in this study was to examine individual and institutional variables among employees that may be related to the expectations of and positivity towards future drone transport to identify the possible relevance of these variables for the successful implementation of drones.

2. Approach and Research Questions

We applied aspects from the multilevel perspective (MLP) developed by Geels et al. [41,42] related to socio-technical systems. This perspective investigates interlinkages between multiple aspects of an organizational system rather than as isolated phenomena in a local microsystem [43].

MLP distinguishes three levels: micro (niche), meso (regime), and macro (landscape), and these terms can parallel the institutional levels of clinic, hospital, and healthcare, respectively. MLP has the potential to increase our understanding of the dynamics that occur in the innovation phases of technological change in such systems [33,44].

MLP defines the technology itself as the core of the technologic change approach [42,45]. It conceptualizes four phases of technological change [44,46]: (1) Technology is introduced within the existing environment, (2) functionalities and user preferences are explored, (3) the change is put into practice in daily operations, and (4) gradual replacement of existing solutions.

These four phases describe the innovation journey that is characterized by various activities and dynamics. Niche innovations become gradually more specific and stable. The process of implementing niche innovations may benefit from positive feedback mechanisms that add new resources to the iterative development. However, if expectations are not met, then faith in the new technology might diminish and resources for it might be withdrawn [47].

Our current drone study was related to the first two phases. We used a structured questionnaire to obtain information related to organizational and technological topics among 400 employees from the three above-mentioned healthcare organizations.

3. Methods

A self-administered questionnaire containing 36 structured questions (Appendix A) was conveyed by mail, published on the organization's intranet, or presented to randomly contacted employees during workdays (in the case of OUS and Vestre Viken). We received a total of 415 responses: 39 by mail, 234 by intranet, and 142 by random contact. The responders were categorized into professional groups as presented in Table 1.

We chose random sampling for a study group of approximately 400 subjects. Although a representative sample should contain, for example, 80 physicians (based upon the personnel structure in the hospitals), selection of 80 physicians across 45 differing specialties in the hospitals might represent a selection bias using representative sampling, as such a method would need a weighted sampling across small and large specialties. Furthermore, whereas representative sampling minimizes bias from known causes, random samples minimize bias from unknown causes [48]. We had no control of the respondents on the intranet questionnaires, and therefore chose a random sampling method for all questionnaires. This resulted in a fairly representative study group, as the percent of the differing personnel categories in our sample versus relative number of workforce were administration 15%/14%, bioengineer 14%/8%, nurse 31%/39%, other 12%/12%, other patient 14%/10%, and physician 19%/17%.

We had three targets of information in our survey. We aimed at capturing "the overall picture" among healthcare employees regarding the individual digital interest and competence, knowledge, and expectations of future drones, some indicators of the local cultures. Multiple variables might be used as we had no prior knowledge of the status. We also intended to limit the time to perform the response, anticipating that much time for filling in the questionnaire could be negative. The 36 questions consisted of 16 yes–no questions, 13 multiple-choice questions, and a Likert scale (rating of 1–5) questions. We tested the questionnaires before the study using three physicians, three nurses, and two secretaries to assess the time needed to fulfill the questionnaire and solve ambiguous questions.

Information of our study was presented on the intranet of the three hospitals. The survey questions alternated between yes/no, multiple-choice, and Likert-scale response options, intending to make it more interesting and appealing to answer [49]. For the survey that was distributed physically to employees at the hospitals, we addressed the employees by inviting them to participate for an important health-related research, i.e., we asked as insiders. Of course, avoiding some respondents that intentionally reply with wrong answers might be difficult to control for in survey-based studies.

The questions collected demographic and background data of the responders, their knowledge of drones, experience with technological change, and attitudes/expectations for the future. The demographic characteristics included profession, gender, age, self-assessed digital competence, currently working under innovative leaders, and working in a culture resistant to change. The technological perspective of the responders was inquired by asking about their knowledge of drones in general and in healthcare, personal beliefs regarding drones in the future, experience of radical technological change (not specified), and anticipated digital implications for their career. Attitudes/expectations for the future were gathered by asking about their expectations of the new hospital, expectations of upcoming technologies for improving healthcare, and perceived threats to their own career development posed by technological change.

We excluded 15 responses because of missing data. The questionnaire data were transferred into Excel for descriptive analysis. The statistical analyses were performed with SPSS version 27. Differences between groups were examined by two-way analysis of variance, and significant Pearson correlations between single variables were used to perform multivariate analyses. To assess whether the employees believed that drones will be implemented in future healthcare services (yes/no), we used stepwise logistic regression with this dichotomous variable against multiple variables. We used dummy variables for personnel groups and the institution. A grouping variable of gender (man = 0, woman = 1) was applied to assess the differences between men and women. A significance level of p < 0.05 was considered to be statistically significant.

4. Results

A summary of the core variable data is presented according to profession in Table 1. There was a majority of women responders (64%), but the physician group had a significantly higher percentage of men (57%) than women. The mean age was lowest in the nurse group (p < 0.01 compared with the total sample).

Administrative staff, nurses, and physicians had the highest belief that in the future drones would be used in healthcare (p < 0.05); however, there was no such trend related to future drone transport at their own hospital. The personnel groups that are closest to patients, i.e., nurses, other patient-related personnel, and physicians, had the lowest expectations of a new, future hospital with respect to improvements in the patient and employee perspective. Administrative personnel scored the lowest on having experienced radical change, while bioengineers scored significantly higher on this topic (p < 0.01).

Int. J. Environ. Res. Public Health 2021, 18, 2637

Table 1. Summary of data by personnel group.

Profession		Administration (n = 59)		Bioengineer (n = 54)		Nurse (<i>n</i> = 110)		Other (<i>n</i> = 47)		Other Patient Related (n = 55)		Physician (n = 75)		Total Population	
Category/Variable	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	
Background															
Mean Age (years)	41.2	11.5	39.8	12.4	32.9	12.1	39.1	13.1	39.0	11.4	35.2	12.9	37.30	14.3	
Years Worked in Hospital (years)	13.5	9.5	14.8	12.4	9.3	10.2	10.1	10.7	9.3	9.9	7.2	8.3	10.6	9.6	
Self Assessed Digital Competence (score 1-5)	3.8	0.7	3.9	0.8	3.7	0.7	3.8	0.9	3.8	0.9	3.6	0.9	3.8	0.64	
Positive Culture for Change (% yes)	88%	33%	74%	44%	90%	30%	78%	42%	73%	45%	83%	38%	82%	38%	
Innovative Leadership (% yes)	83%	38%	74%	44%	84%	37%	70%	47%	71%	46%	69%	46%	76%	43%	
Arena for Innovation (yes)	59%	50%	59%	50%	62%	49%	65%	48%	49%	50%	55%	50%	57%	50%	
Knowledge of Drones															
Knowledge of Drones in Health Care (% yes)	73%	45%	76%	43%	56%	70%	72%	62%	55%	63%	49%	62%	62%	63%	
General Knowledge of Drones (score 1–5)	2.93	0.92	2.75	0.97	2.32	0.82	3.0	1.20	2.55	1.12	2.64	1.13	2.7	1.0	
Believe Drones in Future Health Care (% yes)	81%	39%	67%	48%	73%	45%	54%	50%	60%	49%	77%	42%	70%	45%	
Believe Drones in Own Hospital in Future (% yes)	66%	48%	63%	49%	67%	47%	74%	44%	53%	50%	64%	48%	65%	48%	
Technological Experience and Expectations															
Experienced Radical Technological Changes (% yes)	32%	47%	59%	50%	39%	49%	33%	47%	47%	50%	49%	50%	43%	49%	
Believe New Hospital Positive for Employees (% yes)	68%	47%	65%	48%	66%	55%	70%	47%	64%	73%	53%	55%	57%	49%	
Believe New Hospital Positive for Patients (% yes)	75%	44%	69%	47%	69%	52%	76%	48%	58%	66%	56%	55%	61%	48%	
Believe Digitalization may improve Health Care (% yes)		18%	98%	14%	98%	19%	98%	15%	98%	13%	99%	12%	98%	15%	
Hospital need Change to adapt to Technol. Development (% yes)		28%	87%	34%	83%	38%	65%	48%	80%	40%	75%	44%	82%	38%	
Worried Own Work May be removed by Future Technology (%)		39%	9%	32%	29%	39%	19%	46%	12%	37%	16%	36%	17%	37%	
Positive Expectations Technology Improve Own Work (scale 1-5)		0.90	3.87	0.99	3.93	0.82	3.87	1.26	3.62	1.03	3.67	0.93	3.83	0.81	

There were some score differences across gender (Table 2). Significantly more men had knowledge of drones in general and an awareness of drones in healthcare; however, the belief that drones would be used in future healthcare services was similar between men and women. The scores of women reflected more skepticism regarding the future risk of general technological development in healthcare.

Several variables had interesting covariations. Having innovative leadership increased one's belief that drones would be used in future healthcare services. This was primarily seen in men, as 82% of men who worked in innovative cultures believed in the use of drones compared with 66% of women (p < 0.01). Similarly, for those who experienced radical technological change in one's own work, a significantly higher percentage of men (69%) than women (48%) (p = 0.015) believed that drones would be used in healthcare (Figure 1a–d).



(a) Percentage of men and women with an innovative leader who believe in the use of drones in healthcare.



(c) Percentage of men and women who have experienced radical change and believe in the use of drones in healthcare.



(b) Similar to (a) but for those who do not have an innovative leader.



(d) Similar to (c) but for those who have not reported experiencing radical change.

Figure 1. Interesting covariations that show (**a**) percentage of men and women with an innovative leader who believe in the use of drones in healthcare, (**b**) percentage of men and women without an innovative leader who believe in the use of drones in healthcare, (**c**) percentage of men and women who have experienced radical change and believe in the use of drones in healthcare, (**d**) percentage of men and women who have not experienced radical change and believe in the use of drones in healthcare.

A similar but weaker effect was observed in relation to having an arena for innovative communication; in this setting, 80% of men believed that drones would be used in future healthcare compared with 67% of women (p < 0.05).
Int. J. Environ. Res. Public Health 2021, 18, 2637

 Table 2. Summary of data by gender.

 Males (n = 138)

 Females (n = 262)

	Males	n = 138)	Females $(n = 262)$					
Category/Variable	Mean	SD	Mean	SD	Mean Male/Female	Sig. (2-Tailed)	Mean Difference	Std. Error Difference
Background								
Mean Age	40.07	11.68	35.46	12.70	13%	0.01	3.25	1.87
Years Worked in Hospital	10.06	9.72	10.54	10.73	-5%	ns	-0.99	ns
Self Assessed Digital Competence	3.92	0.84	3.67	0.78	7%	0.01	0.25	ns
Positive Culture for Change	82%	39%	83%	38%	-1%	0.05	0.06	< 0.05
Innovative Leadership	75%	44%	77%	42%	-3%	ns	-0.02	< 0.05
Arena for Innovation	56%	50%	60%	49%	-7%	ns	-0.03	< 0.05
Knowledge of Drones								
General Knowledge of Drone	3.08	0.97	2.40	0.99	28%	0.001	0.69	< 0.01
Knowledge of Drones in Health Care	67%	51%	59%	65%	15%	0.00	0.20	< 0.05
Believe Drones in Future Health Care	78%	42%	66%	47%	17%	0.01	0.02	< 0.05
Technological Experience and Expectations								
Experienced Radical Technological Changes	43%	50%	43%	50%	1%	ns	-0.00	ns
Believe New Hospital Positive for Employees	64%	55%	64%	55%	1%	ns	0.04	ns
Believe New Hospital Positive for Patients	67%	52%	67%	54%	-1%	ns	0.03	ns
Believe Digitalization may improve Healthcare	95%	22%	99%	9%	-4%	ns	-0.04	0.02
Hospital need to Change to adapt to Technological Change	1.28	0.36	1.41	0.41	-9%	0.04	-0.18	ns
Worried Own Work May be removed by Future Technology	14%	34%	19%	39%	-26%	0.01	0.01	< 0.05
Positive Expectations for own Work	3.80	0.87	3.82	1.02	-1%	ns	-0.08	ns

ns = not significant.

Variables that had significant Pearson correlations were included in a multivariate logistic regression analysis using the belief in drone usage in future healthcare as a dichotomous dependent variable (1: believe, 0: do not believe). The results of the final step in a backward analysis are presented in Table 3.

Table 3. Multivariate logistic regression with a belief in drone usage in future healthcare as the dichotomous dependent variable.

Variable	В	S.E.	Wald	Sig.	Exp(B)
General Knowledge of Drones	0.36	0.14	6.46	0.001	1.44
Knowledge of Drones in Healthcare	0.57	0.22	7.08	0.008	1.773
Experienced Radical Technological Changes	0.66	0.26	6.61	0.010	1.929
Innovative Leadership	0.63	0.31	4.18	0.041	1.872
Believe Digitalization may improve Healthcare	0.60	0.14	18.32	0.000	1.825
Nurse	0.67	0.31	4.69	0.030	1.95
Physician	0.89	0.36	5.97	0.015	2.43
Administration	1.00	0.40	6.32	0.012	2.73
Age	-0.02	0.01	4.27	0.039	0.98
Constant	-1.10	0.98	1.27	0.260	0.33

B = Coefficient (intercept); S.E. = Standard error; Wald = Wald chi-square test; Sig. = p-value; Exp(B) = Odds ratio.

Regarding profession, physicians and administrative personnel showed the largest odds ratio for the belief in drone usage in future healthcare, and experience of radical technological change and having an innovative leader were the strongest environmental predictors. Interestingly, increasing age was associated with a lower optimism for the use of drones in healthcare.

5. Discussion

Our results showed that an average of 70% of the participants envisioned drones as a realistic transport solution in future healthcare. This finding was similar across multiple personnel groups, although administrative employees, physicians, and nurses had the highest belief in drone usage. Interestingly, the idea that their own hospital would implement drone solutions was somewhat lower.

The high trust in drones as a future transport solution in health services may be interesting in several aspects. For such subarctic climatic conditions as in Norway, with temperatures varying from -30 to +30 degrees Celsius and wide ranges in wind and precipitation, it has not been proven that the current drone technology may provide near 100% uptime with acceptable quality of biologic specimens. Although a recent study reported that both normal and pathologic blood samples may tolerate exposure to vibration and turbulence [23], multiple challenges, such as icing of drone propellers and wings, and drone range and turbulence around buildings and infrastructures, remain to be studied. We do not believe that our study group has an informed envision based on such topics for their positive assumptions of future drone solutions.

On the other hand, we may interpret our findings as our health care workers have high fidelity in upcoming technologies, that they have heard about and observed drones in general, and therefore have an enthusiastic expectation to drones in the future.

With a socio-technical approach based on individual reported profiles, our study intended to identify possible topics related to technology and local variables in existing systems that may be associated with drone implementation.

Our results demonstrate positive attitudes toward drones among healthcare workers across multiple professions. These workers reported a mean self-assessed digital competence score varying from 3.6 to 3.9 on a 1–5 Likert scale. They had a reasonably high knowledge of drone usage in general and in healthcare, high confidence that technological solutions may improve health services in the future, and high expectations for technology to improve their own work. The employees' concern for their jobs related to future technological change was low across all groups. These observations may, in part, reflect the positive attitude towards technology displayed by the Norwegian population [50].

Experiencing innovation in the local environment, having an innovative leader, and working in an arena that simulated innovation were all positively correlated to the belief in using drone solutions in future healthcare. A similar effect was observed for experiencing radical technology changes in the course of work. However, we cannot conclude whether such individual responses represent local working environments or whether they merely represent individual attitudes. Nevertheless, the significance of such factors with respect to the implementation of drone transport in future health systems should not be ignored, as previous reports have suggested that these topics may be relevant if implementing specific technologies can generate a broader acceptance of general technological innovation [1,3,24]. Our results are in accordance with previous studies that have suggested that greater experience leads to increased familiarity and the ability to learn about technology in general [51] and that the experience and knowledge of early adopters/believers can positively contribute to further technology implementation [52].

The full impact of the gender difference in the current study on the introduction of drones in healthcare is not known. Notably, the effect of working in an innovative culture was a strong factor for men but less for women (Table 2). In contrast, the response profile was quite similar for the two genders in the group that did not report innovative leadership. Whether this may indicate that men are more receptive to innovative cultures than women would need more research. Gender differences related to drones was discussed in a previous report of food drone delivery by Hwang et al. [53]. They found that women who perceived a technology to be innovative were more likely to use the technology than those who did not perceive it to be innovative.

In Norwegian hospitals, 65% of employees are female. With a managerial lens, any gender difference may be of importance related to organizational and innovative developments. If drones are emerging as realistic solutions for the purpose that our hospitals are aiming, understanding possible differences across personnel groups and genders may be useful for the implementation of drones in an extended framework. If we assume that women have a less receptive attitude to innovative cultures than their male colleagues, this should be understood and taken into account.

In our study, administrative personnel, physicians, and nurses showed the largest odds ratios for believing that drones would be used in future healthcare, although they did not stand out with significant higher scores on either digital competence or knowledge of drones. One hypothesis may be that having a higher academic education may be associated with more engagement in future technology. The high score in the administrative group may also be associated with open minded attitude related to managerial positions, although we have no detailed information on their background.

As nurses and physicians are professions close to the clinical activities, they may have more positive experience with the continuous changes in technology and patient treatment over time, although not all changes are observed as radical. As they represent influence on the provision of clinical services, they potentially may be advocates for the introduction of drone solutions, if such solutions are concluded realistic in the future. Further research is needed to fully understand these findings. In a similar study of the Swiss system, Krey et al. [34] reported different attitudes to drones between employees and patients and between healthcare-related personnel and other employee groups. Our results are different from those of Krey et al. because the administrative group had the highest score in our study. This indicates that such factors may vary across different systems and cultures.

The results in Table 3 illustrate several interesting odds ratios Exp(B). The odds ratio for believing in future drone solutions had the largest value in administrative personnel and physicians (2.73 and 2.43, respectively). This is 89.5% and 68% higher than the odds ratio of the general knowledge of drones, and 52.5% and 37% higher than the odds ratio of knowledge of drones in health care. In comparison, the odds ratio of physicians deviated only 25% from that of nurses. These numbers may spark several research questions regarding possible causes as academic background, professional attitudes and cultures, experience from technology, and more.

People's knowledge of emerging technologies may be incomplete and based on their personal values and knowledge constructs [54]. A lack of involvement and knowledge might generate skepticism, whereas positive experiences of radical technological change may build attitudes that are more accepting of change. This idea is consistent with our finding that previous experience with radical change supported a positive attitude toward the use of drones in future healthcare.

Both Mion [36] and Knoblauch et al. [4] discussed the importance of having dedicated project management to support the implementation of drones through the various innovation phases. Change is not driven by the technological solution alone but in combination with the adoption by users, which helps structure the solutions [33,55]. Incentives for experimentation, learning, network building, and vision building have been suggested as crucial elements in the push for change [44,52,56]. These ideas are universal principles, and they correspond to our findings that innovative and supportive management strengthens the belief in future drone solutions.

Negative experiences with a new technology may create acceptance problems whereby people become resistant to subsequent changes [44,57,58]. By contrast, positive experiences and anecdotes may enhance the cultural appeal and social acceptance at the local level [44,52,56]. Users may gradually learn through practical interactions with technology and explore new ways to organize workflows [45].

We found that our study group was receptive to new solutions; however, differing institutional and individual attitudes should be considered by the local leadership and management in future drone implementations. A socio-technical approach might be helpful in this implementation process, as it highlights the importance of using technological transformation as an opportunity to improve processes by combining organizational and technological needs [59,60].

The ability to change and improve existing solutions by implementing technological innovations—in contrast to modifying new solutions to fit existing routines—may become a critical factor for the success of future healthcare organizations.

Drone transport may have the potential to reduce costs by merger of large laboratory services that are traditionally performed at multiple locations with the duplication of infrastructure and 24/7 services. If drones represent a transport system that offers close to 100% uptime with sufficient quality, they may contribute to the centralization of timecritical laboratory services, reducing both operational costs and costs of infrastructure investment. Additionally, drones may also represent decentralization by increasing the pick-up services in rural regions by more frequent and timely transport schedules than current road transport. This may offer both reduced costs for the health care and improved quality for patients.

The full potential of implementing such solutions into existing health care systems may depend on logistics and cooperation across institutional and district systems, thus asking for an extended, socio-technical framework [24,59]. Exploring and understanding the readiness of an organization may therefore be a crucial factor for the implementation of drones. Mion [36] emphasizes that organizations intending to use drones for transport need to be open for change and prepared to modify their operations. Although focusing on the microlevel may be crucial in the initial implementation phases, a systemic perspective may facilitate integrating drones into the healthcare system by considering the entire process, encompassing both social and technical aspects. This would help to improve current work practices and realize the full potential of drones in the long run. Socio-technical systems do not work on their own, but in an interplay between technology and the user environment [42]. Innovative leadership and culture may strengthen the acceptance and implementation of using drone transport in healthcare. Furthermore, it may shape the organizational structure that may be necessary to adapt to technological change.

If drones are emerging as realistic solutions for the involved hospitals, their implementation should attempt to obtain improvements of processes and cultures beyond the primary transport topic. The practical implications of our results may therefore be mul-

11 of 15

tiple. The conceptual framework of attitude and knowledge among health care workers defines the framework for decision-makers of how management and leadership should be performed to attain a disrupting technology adoption in an extended healthcare logistic process. Observing that there may be differing attitudes across the professions and possibly across genders, may be of significant importance.

Our data are harvested from hospitals that have declared drones as a future transport strategy. This may imply differing attitudes compared to other systems. Our findings may therefore be influenced by such specific factors. However, we observe hospitals in several health care systems are developing such strategies and give some generality for our results.

No doubt, implementations of drone solutions in less developed systems may have quite differing challenges as those in more developed systems. In our context, all three hospitals intend improve costs by merging large laboratories, but in combination with improvements in decentralized services, as drones may increase service level in rural and remote locations. The latter parallels demand in many lesser health care systems as well. Nevertheless, observing differing attitudes and expectations across employee groups will have general validity. In particular, in a socio-technical analysis, we conclude that gender, innovative leadership, and innovative arenas may be relevant elements to study when implementing any technological innovation.

6. Strengths and Limitations

Our study is subject to some limitations. The findings are limited to a Norwegian setting. Norway is a high-expender, ranked among Organization for Economic Co-operation and Development countries (OECD) from second to fourth in terms of total health expenditure per capita [61]. This may influence attitudes related to economic expectations for new solutions. Furthermore, Norway, along with other Scandinavian countries, is a leader regarding equity policies. Gender differences may diverse from other health care systems.

We did not define the term "drone" but assumed that most Norwegians associate the term drone with a standard multicopter drone which is regularly shown in the media and is owned by half a million Norwegians.

Although this study included more individuals than in similar previous studies and our results may be valid at the individual level, a limitation of our study is related to survivorship bias [62]. Recruiting respondents via the organization's intranet may have biased the responders towards those who have a special interest in the topic, and they may not be representative of the overall employee population. This may have caused some selection bias in our study group, and we cannot make conclusions on a local clinic or institutional level. Nevertheless, the study group provided interesting perspectives at the individual level across fairly broad personnel groups.

Using a self-administered questionnaire may have introduced errors related to different understandings and interpretations of the questions. Survey questionnaires do not allow for follow-up of the answers and are not suited for a deeper investigation of the responses [63]. Furthermore, our study was carried out in a technologically advanced healthcare system, and the findings may not be as relevant to less developed systems [34]. However, the association between an innovative culture and the successful innovation and implementation of technology has been suggested by others [64] and is supported by our findings.

7. Conclusions

This study suggests that drones are perceived positively across a variety of professions, ages, and locations. A major finding was that working in an innovative environment, having a leader that supports new ideas, and having previous experiences with technological change are drivers for the belief that drones will be used in healthcare and the positive perception of future technology transitions. Our findings require further research related to the impact of the investigated factors on the establishment of drone solutions. However, we believe that to realize the optimal and extensive benefits of implementing drone solutions, the managerial approach should be open minded, stimulate innovation, encourage experimentation, and support the risk-taking that will be necessary.

Author Contributions: H.E.C.: Conceptualization, methodology, formal analysis, and writing original draft; K.-A.J.: Supervision, formal analysis, and writing—review and editing. All authors have read and agreed to the published version of the manuscript.

Funding: The current work has been funded by the Norwegian Research Council.

Informed Consent Statement: The privacy representative office of the hospital stated not necessary.

Acknowledgments: We thank Erik Fosse from the Intervention Centre and Martina Keitsch from the Norwegian University of Technology and Science for helpful comments.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Questionnaire

- Your profession? *Physician, Nurse, Bioengineer/laboratory employee, Other patient-related* work, Administration, Other
- How do you rate your digital competence? *Manage—Superuser*
- Do you think digital solutions contribute to more efficient health services? Yes/No
- Do you think a new hospital will be beneficial for employees? Yes/No
- Do you think a new hospital will be beneficial for patients? Yes/No
- What do you know about drones in general? Very little—Very much
- Have you heard about drones in healthcare? Yes/No
- Do you think a future drone-based transportation of biological material (blood samples, biopsies, other material) is realistic? Yes/No/Don't know
- In which area do you think a drone-based transportation can have a positive effect? *Time, Quality, Digital, Collaboration and Communication, Don't know*
- In which situations do you think a drone-based transportation can be an advantage over current transportation methods? Long distances, Predictability, Possibility for immediate transport, Consolidations of laboratories, Don't know
- Do you think your hospital will use drones in the future? Yes/No/Don't know
- How long do you think it takes to fly a blood sample taken at Rikshospitalet with a drone to Ullevål? 0–15, 15–30, 30–45, 45–60, 60–90 min, Don't know
- What do you believe are the biggest risks when it comes to drone flights for medical purposes? Flight safety, Data safety, Biological quality, Sample safety
- Does your leader support innovative ideas? Yes/No
- How does your leader react to innovative ideas? With rejecting/With doubt/Open/With interest/With support
- Do you have an arena to discuss and/or test innovative ideas in your unit? Yes/No
- Are you leaders active in planning for future change? Yes/No
- Are you involved in the planning for future change? Yes/No
- Do you think that technological development poses a risk for your hospital? Yes/No/ Don't know
- From your perspective, does technological development require that the hospital needs to change? *Yes/No/Don't know*
- Are you optimistic that technology can improve your work in the future? *Pessimistic—Optimistic*
- Are you concerned that your work might disappear or that you might lose work as a result of technological development? *Yes/No*
- Have you experienced significant medical-technical change in your area the last 5–10 years? Yes, radical/Yes, to some degree/Some, but less significant/No
- If you answered yes to the previous question, did technological development lead to logistical/operational change? *Not relevant/Yes/No*
- How satisfied are you with your work today? Not satisfied—Very satisfied

- What gives you most work satisfaction? Interesting work/Work with patients/Work
 autonomy/No work-related stress/Nice colleagues/Exciting change projects/Busy and the day
 goes fast/Good at my work/New challenges
- Do you experience good collaboration and communication with other professions in your work? *Disagree—Agree*
- Do you have enough time during the day to accomplish your work in a satisfactory way? Yes/No
- Will your work be less or more interesting in the future? Less interesting— More interesting
- How important is it for you to know what you need to do during the day? Not important—Very important
- How often do you have to do unpleasant work? Never/Rarely/Sometimes/Often
- What are the reasons for that feeling? Always behind/Poorly handled organisational change/Delayed work/Too much to do/Poor communication/No support/Inconsistent expectations/Too many patients/Concerned for patients
- Do you think the hospital needs other then medical competence in the future? Yes/No
- What are the biggest challenges for the healthcare system from your perspective? Shortage of resources/Increasing costs/Increasing unnecessary treatments/Competence/Poor communication/Specialisation
- Your age? 0–19/20–29/30–39/40–49/50–59/60 years or older
- Your gender? *Female/Male*

References

- 1. Bitner, M.J. Service and technology: Opportunities and paradoxes. Manag. Serv. Qual. Int. J. 2001, 11, 375–379. [CrossRef]
- 2. Coiera, E.; Hovenga, E.J.S. Building a Sustainable Health System. Yearb. Med. Inf. 2007, 16, 11–18.
- 3. Haula, K.; Agbozo, E. A systematic review on unmanned aerial vehicles in Sub-Saharan Africa: A socio-technical perspective. *Technol. Soc.* 2020, 63, 101357. [CrossRef]
- Knoblauch, A.M.; de la Rosa, S.; Sherman, J.; Blauvelt, C.; Matemba, C.; Maxim, L.; Defawe, O.D.; Gueye, A.; Robertson, J.; McKinney, J.; et al. Bi-directional drones to strengthen healthcare provision: Experiences and lessons from Madagascar, Malawi and Senegal. *BMJ Glob. Health* 2019, 4, e001541. [CrossRef]
- 5. Ullah, F.; Sepasgozar, S.; Wang, C. A Systematic Review of Smart Real Estate Technology: Drivers of, and Barriers to, the Use of Digital Disruptive Technologies and Online Platforms. *Sustainability* **2018**, *10*, 3142. [CrossRef]
- Bai, C.; Dallasega, P.; Orzes, G.; Sarkis, J. Industry 4.0 technologies assessment: A sustainability perspective. Int. J. Prod. Econ. 2020, 229, 107776. [CrossRef]
- Amukele, T.K.; Sokoll, L.J.; Pepper, D.; Howard, D.P.; Street, J. Can unmanned aerial systems (drones) be used for the routine transport of chemistry, hematology, and coagulation laboratory specimens? *PLoS ONE* 2015, *10*, e0134020. [CrossRef] [PubMed]
- Rosser, J.C., Jr.; Vignesh, V.; Terwilliger, B.A.; Parker, B.C. Surgical and Medical Applications of Drones: A Comprehensive Review. JSLS J. Soc. Laparoendosc. Surg. 2018, 22, e2018.00018. [CrossRef] [PubMed]
- Scalea, J.R.; Restaino, S.; Scassero, M.; Blankenship, G.; Bartlett, S.T.; Wereley, N. An Initial Investigation of Unmanned Aircraft Systems (UAS) and Real-Time Organ Status Measurement for Transporting Human Organs. *IEEE J. Transl. Eng. Health Med.* 2018, 6, 4000107. [CrossRef] [PubMed]
- Claesson, A.; Fredman, D.; Svensson, L.; Ringh, M.; Hollenberg, J.; Nordberg, P.; Rosenqvist, M.; Djarv, T.; Osterberg, S.; Lennartsson, J.; et al. Unmanned aerial vehicles (drones) in out-of-hospital-cardiac-arrest. *Scand. J. Trauma Resusc. Emerg. Med.* 2016, 24, 124. [CrossRef]
- 11. Haidari, L.A.; Brown, S.T.; Ferguson, M.; Bancroft, E.; Spiker, M.; Wilcox, A.; Ambikapathi, R.; Sampath, V.; Connor, D.L.; Lee, B.Y. The economic and operational value of using drones to transport vaccines. *Vaccine* **2016**, *34*, 4062–4067. [CrossRef] [PubMed]
- 12. Hassanalian, M.; Abdelkefi, A. Classifications, applications, and design challenges of drones: A review. *Prog. Aerosp. Sci.* 2017, 91, 99–131. [CrossRef]
- 13. Amukele, T.K.; Hernandez, J.; Snozek, C.L.H.; Wyatt, R.G.; Douglas, M.; Amini, R.; Street, J. Drone Transport of Chemistry and Hematology Samples Over Long Distances. *Am. J. Clin. Pathol.* **2017**, *148*, 427–435. [CrossRef] [PubMed]
- Kim, S.J.; Lim, G.J.; Cho, J.; Côté, M.J. Drone-Aided Healthcare Services for Patients with Chronic Diseases in Rural Areas. J. Intell. Robot. Syst. 2017, 88, 163–180. [CrossRef]
- Otto, A.; Agatz, N.; Campbell, J.; Golden, B.; Pesch, E. Optimization approaches for civil applications of unmanned aerial vehicles (UAVs) or aerial drones: A survey. *Networks* 2018, 72, 411–458. [CrossRef]
- Nentwich, M.; Hórvath, D.M. Delivery Drones from a Technology Assessment Perspective; Overview Report, No.2018-01; ITA: Vienna, Austria, 2018. Available online: http://www.austriaca.at/0xc1aa5576_0x00377231.pdf (accessed on 8 December 2020).

- 17. Khan, R.; Tausif, S.; Javed Malik, A. Consumer acceptance of delivery drones in urban areas. *Int. J. Consum. Stud.* 2018, 43, 87–101. [CrossRef]
- Kellermann, R.; Biehle, T.; Fischer, L. Drones for parcel and passenger transportation: A literature review. Transp. Res. Interdiscip. Perspect. 2020, 4, 100088. [CrossRef]
- 19. SESAR Joint Undertaking (EU Body or Agency). *European Drones Outlook Study*; Publication Office of the EU: Luxembourg, 2017. Available online: https://op.europa.eu/s/oHuM (accessed on 8 December 2020) [CrossRef]
- 20. Müller, S.; Rudolph, C.; Janke, C. Drones for last mile logistics: Baloney or part of the solution? *Transp. Res. Procedia* 2019, 41, 73–87. [CrossRef]
- Wulfovich, S.; Rivas, H.; Matabuena, P. Drones in Healthcare. In *Digital Health*; Springer International Publishing: Cham, Switzerland, 2018; pp. 159–168. [CrossRef]
- Rosamond, W.D.; Johnson, A.M.; Bogle, B.M.; Arnold, E.; Cunningham, C.J.; Picinich, M.; Williams, B.M.; Zègre-Hemsey, J.K. Drone Delivery of an Automated External Defibrillator. N. Engl. J. Med. 2020, 383, 1186–1188. [CrossRef] [PubMed]
- Johannessen, K.A.; Wear, N.K.S.; Toska, K.; Hansbo, M.; Berg, J.P.; Fosse, E. Pathologic Blood Samples Tolerate Exposure to Vibration and High Turbulence in Simulated Drone Flights, but Plasma Samples Should be Centrifuged After Flight. *IEEE J. Transl. Eng. Health Med.* 2021, 9, 4000110. [CrossRef]
- 24. Eichleay, M.; Evens, E.; Stankevitz, K.; Parker, C. Using the Unmanned Aerial Vehicle Delivery Decision Tool to Consider Transporting Medical Supplies via Drone. *Glob. Health Sci. Pract.* **2019**, *7*, 500. [CrossRef] [PubMed]
- Rabta, B.; Wankmüller, C.; Reiner, G. A drone fleet model for last-mile distribution in disaster relief operations. *Int. J. Disaster Risk Reduct.* 2018, 28, 107–112. [CrossRef]
- Carrillo-Larco, R.M.; Moscoso-Porras, M.; Taype-Rondan, A.; Ruiz-Alejos, A.; Bernabe-Ortiz, A. The use of unmanned aerial vehicles for health purposes: A systematic review of experimental studies. *Glob. Health Epidemiol. Genom.* 2018, 3, e13. [CrossRef] [PubMed]
- Kling, R.; Lamb, R. IT and organizational change in digital economies: A socio-technical approach. SIGCAS Comput. Soc. 1999, 29, 17–25. [CrossRef]
- Konert, A.; Smereka, J.; Szarpak, L. The Use of Drones in Emergency Medicine: Practical and Legal Aspects. *Emerg. Med. Int.* 2019, 2019, 3589792. [CrossRef]
- 29. Lapointe, L.; Mignerat, M.; Vedel, I. The IT productivity paradox in health: A stakeholder's perspective. *Int. J. Med. Inform.* 2011, 80, 102–115. [CrossRef]
- Jones, R.W.; Despotou, G. Unmanned Aerial Systems and Healthcare: Possibilities and Challenges. In Proceedings of the 2019 14th IEEE Conference on Industrial Electronics and Applications (ICIEA), Xi'an, China, 19–21 June 2019; pp. 189–194. [CrossRef]
- 31. Jones, P.; Kijima, K. Systemic Design: Theory, Methods, and Practice; Springer: Tokyo, Japan, 2019; Volume 8.
- Behymer, K.J.; Flach, J.M. From Autonomous Systems to Sociotechnical Systems: Designing Effective Collaborations. She Ji J. Des. Econ. Innov. 2016, 2, 105–114. [CrossRef]
- Geels, F.W. System innovations and transitions to sustainability: Challenges for innovation theory. In Proceedings of the SPRU 40th Anniversary Conference, Brighton, UK, 7–9 September 2006; Volume 14.
- 34. Krey, M.; Seiler, R. Usage and Acceptance of Drone Technology in Healthcare—Exploring Patients and Physicians Perspective. In Proceedings of the 52nd Hawaii International Conference on System Sciences, Maui, HI, USA, 8–11 January 2019.
- Laksham, K. Unmanned aerial vehicle (drones) in public health: A SWOT analysis.(Review Article). J. Fam. Med. Prim. Care 2019, 8, 342._413_18. [CrossRef] [PubMed]
- Mion, F.U. Flying Drones to Exchange Lab Samples: Service Innovation by the Swiss Multisite Hospital EOC. In Service Design and Service Thinking in Healthcare and Hospital Management: Theory, Concepts, Practice; Pfannstiel, M.A., Rasche, C., Eds.; Springer International Publishing: Cham, Switzerland, 2019; pp. 463–479. [CrossRef]
- 37. Hiebert, B.; Nouvet, E.; Jeyabalan, V.; Donelle, L. The Application of Drones in Healthcare and Health-Related Services in North America: A Scoping Review. *Drones (Basel)* 2020, *4*, 30. [CrossRef]
- Seyfang, G.; Smith, A. Grassroots innovations for sustainable development: Towards a new research and policy agenda. *Environ. Politics* 2007, 16, 584–603. [CrossRef]
- Smith, A. Translating Sustainabilities between Green Niches and Socio-Technical Regimes. *Technol. Anal. Strateg. Manag.* 2007, 19, 427–450. [CrossRef]
- 40. Boons, F.; McMeekin, A.; Edward Elgar, P. Handbook of Sustainable Innovation; Edward Elgar Publishing: Cheltenham, UK, 2019
- Geels, F.W. Technological transitions as evolutionary reconfiguration processes: A multi-level perspective and a case-study. *Res. Policy* 2002, 31, 1257–1274. [CrossRef]
- 42. Geels, F.W. From sectoral systems of innovation to socio-technical systems: Insights about dynamics and change from sociology and institutional theory. *Res. Policy* 2004, *33*, 897–920. [CrossRef]
- 43. van der Geest, S.; Speckmann, J.D.; Streefland, P.H. Primary health care in a multi-level perspective: Towards a research agenda. *Soc. Sci. Med.* **1990**, *30*, 1025–1034. [CrossRef]
- Geels, F. Socio-technical transitions to sustainability: A review of criticisms and elaborations of the Multi-Level Perspective. Curr. Opin. Environ. Sustain. 2019, 39, 187–201. [CrossRef]
- 45. Geels, F.W. Technological Transitions and System Innovations: A Co-Evolutionary and Socio-Technical Analysis; Edward Elgar Publishing: Cheltenham, UK, 2005

- Rotmans, J.; Kemp, R.P.M.; van Asselt, M. More Evolution than Revolution: Transition Management in Public Policy. Foresight J. Insur. Risk Manag. 2001, 3, 15–31. [CrossRef]
- Geels, F.; Raven, R. Non-linearity and Expectations in Niche-Development Trajectories: Ups and Downs in Dutch Biogas Development (1973–2003). Technol. Anal. Strateg. Manag. 2006, 18, 375–392. [CrossRef]
- Schield, M. Random Sampling versus Representative Samples. In ASA 1994 Proceedings of the Section on Statistical Education; American Statistical Association: Boston, MA, USA, 1994; pp. 107–110.
- 49. Salkind, N. Encyclopedia of Research Design; SAGE: Los Angeles, CA, USA, 2010.
- Berridge, C.; Furseth, P.I.; Cuthbertson, R.; Demello, S. Technology-Based Innovation for Independent Living: Policy and Innovation in the United Kingdom, Scandinavia, and the United States. J. Aging Soc. Policy 2014, 26, 213–228. [CrossRef] [PubMed]
- 51. Alba, J.W.; Hutchinson, J.W. Dimensions of Consumer Expertise. J. Consum. Res. 1987, 13, 411–454. [CrossRef]
- Rosenbloom, D.; Berton, H.; Meadowcroft, J. Framing the sun: A discursive approach to understanding multi-dimensional interactions within socio-technical transitions through the case of solar electricity in Ontario, Canada. *Res. Policy* 2016, 45, 1275– 1290. [CrossRef]
- 53. Hwang, J.; Lee, J.S.; Kim, H. Perceived innovativeness of drone food delivery services and its impacts on attitude and behavioral intentions: The moderating role of gender and age. *Int. J. Hosp. Manag.* **2019**, *81*, 94–103. [CrossRef]
- Druckman, J.N.; Bolsen, T. Framing, Motivated Reasoning, and Opinions About Emergent Technologies. J. Commun. 2011, 61, 659–688. [CrossRef]
- Cohen, T.; Jones, P. Technological advances relevant to transport—Understanding what drives them. Transp. Res. Part A Policy Pract. 2020, 135, 80–95. [CrossRef]
- Roberts, C.; Geels, F. Public Storylines in the British Transition from Rail to Road Transport (1896–2000): Discursive Struggles in the Multi-Level Perspective. Sci. Cult. 2018, 27, 513–542. [CrossRef]
- Truog, S.; Maxim, L.; Matemba, C.; Blauvelt, C.; Ngwira, H.; Makaya, A.; Moreira, S.; Lawrence, E.; Ailstock, G.; Weitz, A.; et al. Insights Before Flights: How Community Perceptions Can Make or Break Medical Drone Deliveries. *Drones (Basel)* 2020, *4*, 51. [CrossRef]
- 58. Clothier, R.A.; Greer, D.A.; Greer, D.G.; Mehta, A.M. Risk Perception and the Public Acceptance of Drones. *Risk Anal* 2015, 35, 1167–1183. [CrossRef] [PubMed]
- 59. Berg, M. Patient care information systems and health care work: A sociotechnical approach. *Int. J. Med. Inform.* **1999**, *55*, 87–101. [CrossRef]
- 60. Burns, L.R.; Bradley, E.H.; Weiner, B.J. Shortell and Kaluzny's Health Care Management: Organization, Design, and Behavior, 7th ed.; Cengage: Boston, MA, USA, 2020.
- 61. OECD. Health Expenditure-OECD. Available online: https://www.oecd.org/els/health-systems/health-expenditure.htm#:~: text=LatestOECDfiguresonhealth,2020%2CupdatedinNovember2020 (accessed on 8 February 2021).
- 62. Wallis, W.A. The Statistical Research Group, 1942–1945. J. Am. Stat. Assoc. 1980, 75, 334–335. [CrossRef]
- 63. Lazar, J. Research Methods in Human-Computer Interaction; John Wiley: Chichester, UK, 2010.
- Amabile, T.M.; Conti, R.; Coon, H.; Lazenby, J.; Herron, M. Assessing the Work Environment for Creativity. Acad. Manag. J. 1996, 39, 1154–1184. [CrossRef]

Paper 3

H. E. Comtet, M. Keitsch, and K.-A. Johannessen, "Realities of using drones to transport laboratory samples: Insights from attended routes in a mixed-methods study.," eng, *Journal of multidisciplinary health-care*, vol. 15, pp. 1871–1885, 2022

Journal of Multidisciplinary Healthcare

8 Open Access Full Text Article

ORIGINAL RESEARCH

Realities of Using Drones to Transport Laboratory Samples: Insights from Attended Routes in a Mixed-Methods Study

Hans E Comtet ^{1,2}, Martina Keitsch², Karl-Arne Johannessen^{1,3}

¹The Intervention Centre, Oslo University Hospital, Oslo, 0424, Norway; ²Department of Design, Norwegian University of Science and Technology (NTNU), Trondheim, 7491, Norway; ³Department of Health Management and Health Economics, Faculty of Medicine, University of Oslo, Oslo, 0318, Norway

Correspondence: Hans E Comtet, The Intervention Centre, Oslo University Hospital, Postboks 4950, Oslo, 0424, Norway, Email hanscomtet@gmail.com

Purpose: Drones are an emerging technology with the potential to improve laboratory logistics. This study is based on the hypothesis that the implementation of drones will benefit from an understanding of the current system and that existing transport solutions should be optimized before drone solutions are considered. It focuses on carriers transporting biological material today by car along a traditional circular route. It aims to explore how the current transport service is organized, identify areas for improvement, and investigate carriers' perceptions of how drones could be integrated into or substituted for their services.

Methods: A mixed-methods approach was applied, combining a questionnaire, shadowing of nine transport trips, semi-structured interviews, and time assessments.

Results: The carriers shared insights into how to optimize existing transport solutions in terms of structure, culture, attitudes, and overall functionality. Most carriers expressed in the questionnaire that they were little involved in innovation work. The time assessment revealed that not the driving times, but the loading times had the highest deviations from plans and thus represented the area with the greatest potential for simple improvements. Questions about the use and organizational impact of drones are generated, and their prospects are discussed from a broad sociotechnical perspective.

Conclusion: Our concept was to broaden our understanding of implementing drones into existing systems in a fairly simple setting. Although improved logistics may take place without complex processes, future research opportunities such as the impact of drones on organizational processes and social dynamics in the adoption of drones may be needed if more complex systems are involved. The paper proposes experimenting with, and learning from, transport with "road vehicles" and drones in combination and suggests that improvements should be made to existing transport solutions before drones are implemented.

Keywords: logistics, transportation of laboratory samples, reconfiguration, sociotechnical systems, participation

Introduction

The healthcare sector is influenced by an increasing stream of new technological solutions that aim to improve medical treatment and care offerings.^{1,2} These new technologies promise a variety of possibilities for healthcare but also come with certain technical, functional, and operational challenges.

Such emerging technologies may be enthusiastically welcomed by health professionals. Doctors are eagerly implementing new robots for microsurgeries,³ radiologists embrace the use of new and improved in-hospital three-dimensional printing,⁴ and surgeons implement new features and functionalities in mixed-reality applications.⁵

One area that has a long history of applying automation and technical solutions is clinical laboratories. There are developments in point-of-care analyses (POC) that may revolutionize the sector, both by simplifying hospital analyses and providing access to important analyses in a home care setting for patient follow-ups. An emerging technology that, it has been suggested, could disrupt laboratory services is unmanned aerial vehicles (UAVs, drones).^{6–9} Oslo University Hospital plans to construct new hospital buildings by 2030, and considers the use of drones an element of its future

© 1022 Contet et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms. work you hereby accept the Terms. Non-commercial custof the work are permitted works are provided the work is properly accept the Terms. Non-commercial uses of the work are permitted works are per transport logistics.^{10,11} Although the realistic potential and value of drone transport remain unknown, it is well established that both infrastructure for landing and take-off, drone surveillance as well as structures for loading will be needed for efficient and robust drone transport. Few studies have reported on the potential of drone transport to influence hospital logistics from an extended perspective.

The value of the drone services offered by companies such as Zipline¹² and other providers in Africa, where drone transport may provide services to remote and hard-to-reach areas, is easy to appreciate.¹³ Even with open roads, delivery by drone transport may be faster and cheaper than ground transport in these areas. Although drone technology is developing quickly, drones will show their full potential only when they are able to fly beyond the line of sight (BVLOS)¹⁴ and fully autonomously for long distances. Nentwich et al¹⁵ predicted that autonomous drones would need to be programmed to reach their targets, to have self-reliant route planning, and to be self-sufficient in averting obstacles. A combination of different technologies may be necessary to reach this level of automation.¹⁶ Although Mion¹⁷ reported in a Swiss case study that the implementation of drones had only minor effects on organizational processes, he also argued that a broader perspective should be taken when integrating drones.

A widespread adoption of telehealth solutions¹⁸ or the acceptance of machine learning applications for clinical laboratories¹⁹ may require taking a broader sociotechnical perspective to fully understand how these technologies can be integrated into existing health care systems. However, whether traditional and existing clinical processes, logistics, and patient care can be adapted to drone solutions taking a similar approach, and if and how drones contribute to sustainable healthcare systems, is currently not completely understood.^{20,21}

New solutions and approaches may challenge the status quo, namely, established work processes and routines. Existing systems may survive for long periods if they are locked in by traditional ways of operation. In an early phase, new technologies may develop without consensus being reached on their overall benefits if, for example, it is challenging to comprehend how the new tool can be integrated into existing workflows in a way that takes advantage of the technology to improve existing processes.

Awad et al²² and Bongomin et al²³ illustrate a future vision of a connected healthcare system, and in their scenarios, drones may play an important role. Although it is not obvious that the future scenarios they sketch out will take place, the combination of multiple future technologies will certainly generate new solutions that we were not able to foresee only a few years ago. In Norway, political guidelines for future patient treatment aim to increase the use of digital patient contacts, enabling digital home treatment for 30% of patients by 2030.²⁴ This ambition was sparked after the boost of such patient contact during the COVID-19 pandemic. Whether the extensive transitions visioned by Awad et al and Bongomin et al may be enabled by single technologies or will need transformational changes in multiple technologies and between multiple actors,^{25–27} remains to be seen. Focusing on the interplay of multiple singular disruptions, with greater attention paid to complete concepts, may be helpful in observing add-ons that may trigger further innovations.^{24,27}

One approach that has demonstrated its usefulness in approaches to extensive transformations is the multi-level perspective (MLP). This may be considered a part of transition studies that is useful for providing a holistic perspective on system changes, as the MLP takes into consideration both the involvement of different actors and multiple interactions in a system and dynamic alteration at different systemic levels.

According to Geels, a restricted focus on innovations often ignores in-depth interests in system innovation.²⁶ In comparison, system innovation has a broader analytical view that may include aligning with more overall ongoing processes or multiple innovations. Together and over time, these innovations may lead to system innovations. This study aims to mitigate this gap by focusing on drones and a reconfiguration pathway of system innovation.

Sociotechnical approaches emphasize understanding the work practices in which technology will be used.²⁸ Following this argument, this article focuses on the carriers at OUS today who transport biological material by "road vehicles" along a traditional circular route.

The hospital transport services' carriers are employees who should be considered potential stakeholders and resources in implementing drone solutions. Automated transport of drones may influence the carriers' future job positions, and they may represent knowledge of how new solutions should be designed if drones are to be implemented as hybrid solutions that combine existing practices with a new transport modality.

This study is based on the hypothesis that the implementation of drones will benefit from an understanding of the current system, and that existing transport solutions should be examined and possibly optimized before drone solutions are considered.

Our research ambitions are as follows:

- 1. To explore how the current transport service is organized and identify areas for potential improvement; and
- To study what the carriers perceive as important factors related to how drones could be integrated into or substituted for their service offerings.

Background

Institutions

Oslo University Hospital (OUS) comprises four hospitals located within Oslo: The National Hospital (providing local, regional, and national services), Ulleval Hospital (providing local, regional, and national services), Radium Hospital (a specialized cancer hospital), and Aker Hospital (a local and central hospital). The current ground transport provides transport of both biological test samples, blood, and other items across differing locations.

Route Organization

Figure 1 shows a map of the institutions. The routes today are mostly organized in a circular design with multi-stop routings except for some on-to-one trips. The different lines indicate the distance between the locations, together with the number of trips.



Figure I The National Hospital, Ulleval Hospital, Radium Hospital, Aker Hospital, and the Blood Bank in the Red Cross house.

Four routes from the National Hospital to Radium Hospital are served by a bus today that transports patients but also carries payloads of various types. This part of the OUS transport service was not considered in our study.

The article proceeds as follows. The following section introduces the MLP framework and reconfiguration pathways. The next section describes the methods, followed by presenting the results and a discussion. Then, continuing with the limitations of our study, we conclude with our findings.

Is a Reconfiguration Approach Relevant for Drone Implementation?

This paper assumes that drone solutions are initiated by innovators and technologically oriented enthusiasts at the clinical level and that their initiatives will influence the institutional levels of clinics, departments, and institutions. This process may create upstream driving forces in the institution, which in turn may create pressure from above to regulate, adapt, and form the potential implementations and consequences of the innovation. Such processes parallel the hypothesis and sociotechnical approach of the MLP and transition management theory.^{29–32}

The potential value of a reconfiguration pathway, as depicted in the MLP, is generated by the method's hypothesis that transitions are not caused by the breakthrough of one technology but by a combination of multiple component innovations.³³ These new components are initially developed in what MLP denotes as "niches", which are equivalent to the clinical units in our system. In the system discussed in this paper, the MLP approach would identify clinics as a protected space for radical innovations that need protection because of their strict regulations regarding medical soundness and documentation.³⁴ Different forms of subsidies may provide protection. Simultaneously, the more a technology is developed and used, the more is learned about it, and the more it improves.³⁵ Clinics are also the level that provides space for experimentation and learning processes. Thus, they may also be significant arenas for building socio-technological networks to develop organizational cultures that support innovation.³⁶

This requires adequate adoption of the innovations at the clinical department level to solve the challenges caused by unfamiliar innovations at the institutional level, namely, the hospital (equivalent to the regime level in the MLP system).^{33,37} To obtain a gradual implementation of innovations at the healthcare level (parallel to the MLP landscape level), deeper structural trends must be triggered to create opportunities and pressure for further change. The expectations and ambitions of political guidelines are examples of such structural trends.

Multiple technologies, including drones, are being continuously developed in combination with each other. Many stepwise changes may add up to reconfigurations that, over time, may create major system changes. Figure 2 illustrates the reconfiguration pathway.

Methods

A mixed-methods approach was applied, combining a questionnaire, shadowing of nine transport trips, semi-structured interviews, and time assessments with carriers from the transport department.

The findings in this article are based on 16 questionnaires, 9 reports from shadowing, 8 interviews, and multiple time measures from the transports. By combining both quantitative and qualitative data, the objective was to picture the current services and the carrier's work.^{38,39} Self-administered questionnaires were given directly to the participants.

Ethical Considerations

The study was approved by the hospital's Institutional Review Board and complied with required ethical and legal standards relating to participant anonymity and confidentiality. The data material does not contain any identifying characteristics. Furthermore, data aggregation is only on group-level information. Participation in the survey and interviews was voluntary. All interviewees gave informed consent orally. The actual participation, and the information provided to the participants, imply consent based on conclusive behavior. As the study is conducted anonymously, it is outside the Regional Committees for Medical and Health Research Ethics (REK) mandate. Therefore, ethical approval from REK was not considered necessary.



Figure 2 The reconfiguration pathway of the MLP theory translated into a healthcare concept, adapted from Geels FW, Schot J. Typology of sociotechnical transition pathways. Res Policy. 2007;36(3):399–417 Copyright 2007, with permission from Elsevier.⁵⁵

Attended Routes for Observation

Three internal routes were attended three times. Information about arrival and departure times, stops, delays, and disturbances, as well as a general assessment of the transport flow, was gathered by interviewing the driver as well as measuring actual time intervals during the trips.

To ensure that identical questions were presented to all drivers, an interview guide was used to funnel the conversation with the carriers into the questions while accompanying them.

We asked about their opinions of (1) their workday, (2) factors that influence time, (3) changes in routes, (4) barriers and opportunities regarding their work, and (5) their thoughts about the future.

Questionnaire

A self-administered questionnaire containing 36 structured questions was distributed amongst the carriers in the transport department to understand the transport department better. Subsequently, participants were selected using purposive sampling,⁴⁰ with the main interest of exploring the range of opinions.⁴¹ As a result, we received a total of 16 responses.

Transport Department, Routes, and Work Process

The transportation department at OUS is organized as part of the OUS Hospital Service Division (OSS). The transportation department is located at Ulleval Hospital, from where carriers drive across the different institutions. The transport department has 20 carriers who operate different routes using "road vehicles", covering the transport of various biological materials, supplies, clothes, and food. The carpool ranges from normal-sized delivery vans used the most (eg, Renault Kangoo) to trucks (eg, Mercedes 3.5 ton).

The routes cover both internal and external clients. Our study focuses on the internal routes between the National Hospital, Ulleval University Hospital, Radium Hospital, and Aker Hospital.

The carriers are based at Ulleval Hospital, where they start their daily routes for trips across the institutions. The loading is done manually by the carriers, who do both the pick-up and delivery at the different laboratories. The pick-up

is organized differently in different locations, and the payload is also stored differently. As we followed the routes, we observed that three major factors influenced the transport time; delays in the traffic, delays when the cars had to wait at the loading site, causing varying and prolonged loading times, and the transfer of delays for the next trip. This sparked the hypothesis that the multi-stop routing with layovers was not always efficient. Furthermore, such multiple routings would not be relevant for future drone services because such serial transports would include legs with multiple drone takeoffs and landings, consuming significant energy for hovering and transport times.¹⁴ We, therefore, intended to consider direct routings, ie, round trip tours between two locations. We measured all time intervals during the shadowing of the tours and also interviewed the drivers about the delays as they occurred. Based on the time measurements, we could experiment with different alternative direct routings, always ensuring that current transport routings were covered to the same extent as in existing routings. Time assessments were analyzed by a simple model using Microsoft EXCEL 2016. Mean, maximum and minimum driving and loading times were calculated.

The current transport system involves a daily driving distance of approximately 290 km. We followed each route three times on weekdays from Monday to Thursday in June 2021. Table 1 below illustrate the current work processes.

Analysis

The analyses followed a convergent design, in which the quantitative and qualitative data were analyzed separately and then merged to combine the results.⁴⁰ This design allowed us to link different findings and gain a complete understanding of the transport system.

The interviews were noted, transcribed, and entered into Atlas.ti (<u>www.atlasti.com</u>), which is a qualitative data analysis software package. However, Atlas.ti was not used to analyze the data but primarily to organize the data.

The transcripts were analyzed, and the data was structured into relevant categories.^{42,43} The categories emerged from the review and were selected for their relevance to drones.

The round-trip transport times across the locations were assessed, except for the Ulleval–Aker route, which did not have a return trip but was a one-way transport. The measured time intervals in the current routes include driving times (the time from departure from one location to arrive at the next location) and loading times (assessed as the time from arrival to departure for the next leg in the route). The loading time thus includes both the waiting time and the time for delivering/ collecting the payloads at each stop. These time intervals were used to assess total transport times in the existing trips across the different locations and to estimate theoretically optimized transport times based on minimal time intervals.

Results

Observations

Six categories emerged from the review of the transcripts, together with the findings and consequences (Table 2).

Table I Route Description

Routes	Collected Payload	Approx. Total Length/Day
Ulleval University Hospital, the National Hospital, and Aker University Hospital	Biological material and other goods between 07:30 and 14:30. Transported in bags, Styrofoam boxes, or envelopes. Envelopes are also used for mail between the hospitals.	80 km
Ulleval University Hospital, the National Hospital, and Red Cross	Five trips to the National Hospital. Blood is transported in suitcases, transport bags, or Styrofoam boxes. Mail between the hospitals is transported in a laptop bag.	86 km
Radium Hospital, Aker University Hospital, Ulleval University Hospital, and the National Hospital	Pathologic samples between 08:00 and 16:00. Samples are transported in gray boxes. Mail for the other hospitals is in pink boxes.	I24 km

	Tab	le 2 Categorie	s That Emerged from the Transcripts
	#	Category	Description
ıltidisciplinary H	I	Change and adjustments of routes	This theme describes changes to the rou and the degree to which the carriers bel they have the power to change and adjus routes.
	2	Customer perspective	The carriers have a service function with respect to the laboratories. This theme describes how the carriers perceive and represent their roles.
		Tab # 1	Table 2 Categorie # Category I Change and adjustments of routes 2 Customer perspective

This theme describes changes to the routes

	adjustments of routes	and the degree to which the carriers believe they have the power to change and adjust the routes.	with only minor adjustments. Change is typically initiated top- down. The hospital divisions and laboratories are the gatekeepers of change.	accomplish.
2	Customer perspective	The carriers have a service function with respect to the laboratories. This theme describes how the carriers perceive and represent their roles.	The service offering depends on the carriers' attitude and familiarity with the locations. In addition to the transport of samples, the carriers help to place samples, and supply empty racks and containers. They help transport mail, fabrics, garbage, medical equipment, and medicine.	Lack of incentives and influential force to change the system may lead the carriers to become rather passive service workers, ie, not actively trying to change the system.
3	Service Structure	The service structure looked at conditions of how the service is organized today.	Constant delays when picking up samples, because the carriers are asked to wait a few minutes until the samples are ready for transport. The carriers follow a schedule with fixed times, which includes additional time for waiting.	In relation to drones, the delays may be problematic because drones fly according to scheduled times. One carrier pointed out, "following the plan [routing] slavishly also means not being able to wait."
4	Management of change	This theme describes who may be responsible for change and if carriers have areas where they can discuss and exchange ideas.	The carriers suggested that they should be involved in changing routes. Furthermore, the carriers wanted to have leaders who appreciate their meaning, as well as space to experiment and try out new things.	The carriers indicated that greater involvement would result in their being more motivated and satisfied at work. Furthermore, the carriers may have no space for exchanging their concerns or ideas about how to improve the current system.
5	Threats and support by drones	This theme deals with anxiety about possible job loss and how carriers think drones could support their work.	The carriers assessed the use of drones for transport purposes as a threat, because for them drones had been talked about as a replacement for their service offering. However, the carriers also argued that drones could support their work.	The carriers are afraid of being sacrificed for efficiency goals or private companies that overtake the transport. The carriers are convinced that although technology may replace some work, there will always be a need for a backup, Furthermore, they do not believe that drones can replace them in meeting the transport needs to which they cater with "road vehicles" today. Different sizes, weights, and established routines that are difficult to change were mentioned as reasons. However, the carriers are open to using drone support. One carrier argued, "Carriers plus drones can complement each other", while another added, "All of the carriers have additional routes they could give to a drone to make it easier for themselves." Another suggested, "Drones may replace 'road vehicles' for all the taxi trips that the laboratory orders."
6	Miscellaneous	This theme describes the miscellaneous observations that were made while	Transport boxes are not standardized and varying in size and material. Furthermore, carriage issues, such as leaked formol	Transport solutions are not suited for transportation by drone.

The routes have existed almost unchanged over a long time

Findings

Consequences

We thus observe that making changes to routes is difficult to

Ouestionnaire

Sample Characteristics

The mean age of the 16 participants was 48.34 years (SD 2.88). The mean number of years worked at OUS was 11.75 years (SD 3.18).

A summary of the most relevant questions in the questionnaire is presented below in Table 3.

Time Measures of the Routings

The routes that we joined were driven daily between 07:30 and 16:30. Table 4 shows the mean, maximum, and minimum driving times and loading times and the estimated total transport times. In the multi-leg routings, the loading times are the sum of all loading times at each site.

Table 4 shows that the loading time represented 33–64% of the total transport times, and the high loading times were often caused by one delay at one of the stations. In our assessments for improvements of the transport routings, we used the mean driving time in direct routes and the shortest loading time, assuming that the latter represented a realistic improvement if better logistics are implemented.

Times from the current and suggested new routings are compared in the upper part of Table 5, the direct routings in the lower part. The most profound gain was observed by transforming the multi-leg routing to direct routings. This is illustrated in Figure 3 for the two most complex routings; the Ulleval-Aker transport with four legs and the Radium-Aker with three legs.

A simple illustration is shown in Figure 3.

Using such modifications to the existing routings, it was theoretically possible to use one less car compared with the current system.

As a summary of our findings, we observed that the carriers had substantial knowledge of how the current transports could be improved, they were interested in how new solutions should be developed, and not least, the timelines of the current ground transport could be substantially improved by simple changes.

Discussion

This study examined basic topics related to logistics, service levels, and personnel involved in an existing ground transport system that it has been suggested could be substituted for by drone transport. Our findings showed that insights

Category	Scale	Variable
Threats		
Are you concerned that your work might disappear or that you might lose work because of technological development?	Yes/No	5/11 (31/69%)
Are you optimistic that technology can improve your work in the future? Pessimistic—Optimistic	Scale 1–5	3.00 (SD 1.36)
Initiating		
Do you think a future drone-based transportation of biological material (blood samples, biopsies, other material) is realistic?	Yes/No/Do not know	9/2/5 (56/13/31%)
Do you think your hospital will use drones in the future?	Yes/No/Do not know	9/1/6 (56/6/38%)
Involvement		
Does your leader support innovative ideas?	Yes/No	1/15 (6/94%)
Do you have an arena to discuss and/or test innovative ideas in your unit?	Yes/No	2/14 (13/87%)
Are your leaders active in planning for future change?	Yes/No	0/16 (0/100%)
Are you involved in the planning for future change?	Yes/No	1/15 (6/94%)

Table 3 Summar	of the Most	Relevant Qu	lestions in	the Questionnaire
----------------	-------------	-------------	-------------	-------------------

1878

Existing Routes	Average Driving Time	Average Loading Time	Maximum Loading Time	Minimum Loading Time	Mean Total Transport Time	Loading Time as % of Total Transport Time
UL-AK-RH-UL	40,2	33,6	48,0	25,0	73,8	46%
UL-RA-RH-AK-UL	44,4	43,7	83,0	21,0	88, I	50%
UL-RA-RH-RA-UL	63,0	30,7	43,0	23,0	93,7	33%
UL-RA-RH-UL	39,0	68,0	82,0	55,0	107,0	64%
UL-RH-AK-UL	22,3	24,8	42,0	7,0	47,0	53%
UL-RH-RK-UL	39,0	39,3	45,0	35,0	78,3	50%
UL-RH-UL	19,3	17,5	46,0	8,0	36,7	48%
UL-RK-UL	27,3	28,4	44,0	16,0	55,7	51%
Average All	30,0	27,8	83,0	7,0	57,9	48%

Table 4 Driving, Loading, and Optimized Times

Notes: All times in minutes

Abbreviations: UL, Ulleval Hospital; AK, Aker Hospital; RH, the National Hospital; RA, Radium Hospital.

	Route	Ulleval- Aker (4 Legs)	Radium- Aker (3–4 Legs)
Current Route (Multiple Legs)	Mean driving time	44.3	43.0
	Maximum driving time	59.0	52.0
	Minimum driving time	35.0	35.0
	Mean loading time	23.9	40.2
	Maximum loading time	67.0	83.0
	Minimum loading time	10.0	16.0
	Mean total transport time	68.2	83.2
	Maximum total transport time	126.0	135.0
	Minimum total transport time	45.0	51.0
Direct Route	Mean Driving Time	14.0	28.0
	Maximum driving time	16.0	40.0
	Minimum driving time	11.0	18.0
	Minimum loading time	10.0	8.0
	Mean Total Transport Time	24.0	36.0
	Time Saved	44.2	47.2

 Table 5 Current and Direct Route

Notes: All times in minutes.

from the carriers who perform the current ground transport contributed knowledge helpful in optimizing the existing transport solution concerning the overall functionality. More general, our generated insights on structure, culture, and attitudes may be relevant to obtaining a more innovative transport system. This, in turn, provides information about how drones should be implemented, either as a substitution for or as an integrated part of current ground transport.

The carriers had a somewhat passive attitude toward the system performance, which seemed to be associated with their perception that they had little influence on how the existing system is developed and improved. The current solutions have mostly remained unchanged for 20 years, although the carriers stated that they had presented several ideas for improvement. Their impression was that although the external framework for transport had changed in many ways



Total Transport Times in Excisting and Direct Routes

Figure 3 Total transport times in existing and direct routes.

over the period, their proposals for improvements for example HUBs were rarely discussed. The carriers said that they were open-minded regarding involvement in solving problems and improving quality, but that they perceived the institution as having a deficit in open-minded management.

From our interviews, we concluded that the carriers have suggestions regarding how to reduce transport times and improve regularity and how drone transport should be integrated with current solutions in the future. They are motivated to contribute to the implementation of drone services with a positive attitude, but their expectations of being involved are low because they feel that their workplace does not involve in the innovative processes. Although this perception was strong among the carriers, we suspect that this is not representing the actual state of the department and assume that the carriers' statements that none of the leaders are active in planning for future change are considerably biased. Furthermore, the fact that only 31% of the carriers had some worries about their future jobs if drones were implemented is not equivocally telling they are scarcely involved in system development and organization. Nevertheless, we conclude that involvement should be considered because carriers do more than just transport goods; they also significantly influence how this transport works concerning how their customers are acting and the service quality perception of the customers.

We observe that the current ground transport system is not primarily focused on transport times, but rather on maintaining flexibility for the laboratories that are their customers. If the laboratories ask for the transport to wait and postpone a given departure, this is accommodated, despite that this will cause a delay. This flexibility may be part of the explanation for some of the long loading times that we observed. Whereas such flexibility may seem valuable for the actual site requesting elasticity of the transport at an actual time, there is scarce knowledge of whether this reduces the total service when all transport legs are considered as a whole.

This was also reported by the carriers. They had multiple examples of how the flexible departures were often caused by insufficient planning and processes in the laboratories to prepare for the pick-up. This indicates that, whereas the routings are organized as regular trips, the laboratories may act as if the transports are tailored to the laboratories' insufficient preparations. Accordingly, a consequence of their efforts to satisfy their customers is that there are often delays, with consequences for subsequent trips.

We, therefore, conclude that the current transport system is not considered time-critical; that is, the total transport times are not being monitored or processed for standardization or optimization. This, in turn, sparks the question of whether the "timely" demand of the current transport system makes it a good candidate for drone solutions. With the short distances between transport sites, the time savings from drones will be limited. The optimized times in Table 5 show an improvement of 17–38%, calculated in minutes saved; this indicates an improvement between 5.4 and 16.3

minutes. It may be argued that any minutes saved by a drone carrying a cardiac defibrillator could save a life. However, it may be doubted that this is equally crucial to transporting routine blood samples. Some minutes saved may have marginal clinical significance in specific healthcare systems where patients may wait for months for their first contact with healthcare provisions.¹⁴ Furthermore, to what extent faster transport will influence the efficiency of healthcare to any significant degree may be an open research question.

If the variations in the transport times and loading times that we observed are acceptable, this may indicate a limited willingness to pay for drone solutions. The potential competitiveness of drones will therefore be related to whether they have lower costs than the current ground transport system for comparable service times or result in real-time savings.

If large cities have frequent future drone trips and routings for multiple sectors and with many operators, drone transport should be expected to be strictly regulated by time slots and trajectories of unmanned BVLOS flights. This may, in turn, demand strict and timely routings with minimal deviations, and the "flexible departures" in the current ground transport system at OUS may violate such regulatory criteria. This may have a significant impact on how drone solutions are organized in urban locations.

The potential of drones to avoid ground traffic congestion has also been used as an argument for the time savings achieved by drone implementations.^{13,44–47} Furthermore, in this context, stricter organization of the routings may be needed. In our system, it was the loading times and not the driving times that caused the highest deviations from plans and may thus represent the area with the greatest need for improvement. The minimum loading times as benchmark times promise considerable potential for time savings. However, whether such improvements, with only minutes, gained per leg, have any clinical value at all remains to be evaluated.

Theoretically, building a revised model with optimized transport times could change the existing route organization. The optimized total transport times show that hourly routings may be a realistic alternative to the current model and that one round-trip leg with one car could handle transport needs with shorter transport times than the existing ones. In our theoretical redesign, one car and carrier could be made redundant, thereby having some gains. Such improvements should be implemented before comparison between drone and ground transport is performed. However, our theoretical model would demand strict loading and departure times; otherwise, substantial delays may occur. The value of strictly regular departures is that missing a specific departure, and thus waiting for the next one, will cause minimal delays.

Will Drone Implementation Need Specific Organizational Processes for Implementation?

The current transport routes are not complex, and they are not focused primarily on transport times. Improvements should be realistic either with or without drones. Drones may therefore represent a logistical solution with relatively little impact on organizational processes and may have little effect on other clinical services apart from the transportation service. This suggests that the implementation of drones may be rather uncomplicated concerning the pure transport topic. Therefore, we conclude that the institutionalization of drones, ie, the process of innovation,⁴⁸ may be achieved without many complex processes. This is in accordance with our findings, as only five of 16 carriers expressed concern in the questionnaire.

Our study showed that the current system focuses on flexibility, but although the current solution has not been changed extensively in the last 20 years, there should be a regular assessment of which topics are the most valuable for the laboratories.

An alternative to a simplistic approach to drone implementations in simple models like ours, is the hypothesis that in more complex systems, such solutions should be considered from a broader organizational perspective. The combination of micro-behaviors as individuals adopt an innovation, ongoing regime developments and landscape pressures may, over time, result in more significant system changes. One perspective could be that the transport department rebrands itself as an innovation hub for logistics with an open culture for experiments and change. As a carrier put it, "The old structure needs to be cracked to be more innovative."

The theory, however, is that socio-technologic transitions require the interplay of technological and social processes that reinforce each other.²⁵ In our setting, our insights into the technological process are scarce because drone technology solutions are developed externally and not from a hospital perspective. Diffusion across technology and medical processes related to drones may therefore be rather limited. Although drones may seem promising, there is no guarantee for drones to break through and disrupt laboratory transports.

Getting support from drones was discussed as a valuable add-on to the service offering of the transport department. According to the carriers, drones may be used for specific routes, ie, as a replacement for taxi trips that today are used for emergency analysis or outside of regular working hours. However, economic, or technological considerations may strongly influence whether this is realistic, and there are not many scientific data to enlighten this topic.

However, the carriers seemed to have the essential knowledge about logistics and the continuous transport of biological material and other goods and they believed that a combination of drones and the current system may be beneficial. Including them in the introduction of drones may support the acceptance of drones, a topic that has gained much interest in the literature.^{49–54}

Limitations of the Study

The small volume of data in the current study may of course be observed as a significant limitation. Small volumes may both challenge the validity of scientific conclusions as well as the soundness for generalization across differing systems. However, our conclusions are not depending on complex scientific deductions but logical conclusions from simple observations and we intended to elaborate on possible ideas for simple approaches. We hypothesize that the generalizability of such findings across differing systems may depend more on systemic differences related to service need and structure, than on the specific findings in each setting.

Although Oslo University Hospital is a large hospital, compared to European standards, the distances between the various locations are short. Furthermore, traffic congestion is probably less than in many larger cities. Therefore, some of our findings may not apply to other locations. Large hospitals in other cities may have quite different structures and transport needs. Nevertheless, involving the workforce in providing transport solutions may reveal important insights.

Our finding that current ground transport could easily be improved is therefore specific to a rather plain model. The interviews were conducted at the OUS transport division and are ultimately not representative of other hospitals or transport departments. Furthermore, the existing ground system with "road vehicles" at Oslo University Hospital is not based on transport times but flexibility for the laboratories. Although these factors may limit the generalization of our actual findings, it may be of value for other transport departments and hospitals to conduct similar research to understand how carriers value drones. Their narratives provided insights and underscore the necessity of open-minded and innovative leadership to facilitate change and acceptance.

This study focused on topics related to current transport solutions and showed that a comparison of drones with optimized ground transport shows close competition. The argument about avoiding ground traffic congestion and the value of time efficiency has been put into perspective. Of course, if drone transport shall be implemented in future transport solutions, multiple infrastructures are needed such as landing sites and automatic solutions for loading and unloading, drone surveillance and drone maintenance. Unfortunately, the study of such topics is currently challenging to perform today because of the immaturity of such solutions. It should also be expected that various types of drones and drone sizes will be used for differing purposes, demanding differing solutions for the mentioned structures. No doubt, such perspectives will be necessary to define in future studies. Nevertheless, the advantages of drones in other settings may be more evident. For example, drone transports in rural districts and on-demand services in emergencies have proven valuable and may have more significant clinical value, resulting in greater willingness to pay, than short-distance transports.

Although a sociotechnical perspective with the MLP and transitions research has a powerful analytical force, transitions do happen without it. The purpose of our conceptualization was to broaden our understanding of how to implement drones into existing systems. We believe that several future research opportunities, such as the impact of drones on organizational processes and social dynamics in the adoption of drones, may extend the understanding of these questions and nudge transitions in the desired directions, and have significant relevance in more complex settings.

Conclusion

The carriers involved in this study had useful knowledge of how future drone solutions should be implemented and were positive about the use of drones as part of their future services. They showed that existing transport solutions should be analyzed carefully to realize possible improvements before drones are implemented. The lesson learned from their input is that drone implementations should be evaluated in context with current ground solutions, either as integrations into or partial substitutions for current services, and that involved personnel should be involved in developing such solutions.

We conclude that in straightforward transport systems such as the one we examined, improvements to current solutions may be obtained without complex processes. This may of course vary between different locations, service demands, and the complexity of the services. Whether an extended organizational perspective of drone implementations is needed in more complex settings needs further research. We also conclude that although improvements may be obtained by unsophisticated changes, the potential for added organizational and innovative processes should always be considered, in particular, if multiple new technologies are included.

Informed Consent Statement

Informed consent was obtained from all subjects involved in the study.

Acknowledgments

We are grateful to the transport department at Ulleval Hospital for their cooperation and participation in this research project. Furthermore, we thank Ole Andreas Alsos from NTNU for helpful comments regarding the visualization and David Mulrooney, Ph.D., from Edanz for editing a draft of this manuscript.

Funding

The current work has been funded by the Norwegian Research Council under Grant 282207/2018.

Disclosure

The authors report no conflicts of interest in this work.

References

- 1. Ćwiklicki M, Duplaga M, Klich J. The digital transformation of healthcare: health 4. 0. Milton: Taylor & Francis Group; 2021.
- Greaves RF, Bernardini S, Ferrari M, et al. Key questions about the future of laboratory medicine in the next decade of the 21st century: a report from the IFCC-Emerging technologies division. *Clin Chim Acta*. 2019;495:570–589. doi:10.1016/j.cca.2019.05.021
- 3. Arezzo A, Gagliardi G. Transanal endoscopic microsurgery: is robotics the way to go? Tech Coloproctol. 2021;25(11):1179–1182. doi:10.1007/s10151-021-02514-8
- Mitsouras D, Liacouras PC, Wake N, Rybicki FJ. RadioGraphics update: medical 3D printing for the radiologist. Radiographics. 2020;40(4):E21– E23. doi:10.1148/rg.2020190217
- Kumar RP, Pelanis E, Bugge R, et al. Use of mixed reality for surgery planning: assessment and development workflow. J Biomed Inform. 2020;112S:100077. doi:10.1016/j.yjbinx.2020.100077
- Otto A, Agatz N, Campbell J, Golden B, Pesch E. Optimization approaches for civil applications of unmanned aerial vehicles (UAVs) or aerial drones: a survey. *Networks*. 2018;72(4):411–458. doi:10.1002/net.21818
- 7. Sanfridsson J, Sparrevik J, Hollenberg J, et al. Drone delivery of an automated external defibrillator A mixed method simulation study of bystander experience. Scand J Trauma Resusc Emerg Med. 2019;27(1):40. doi:10.1186/s13049-019-0622-6
- 8. Wulfovich S, Rivas H, Matabuena P. Drones in Healthcare. In: Digital Health. Springer; 2018:159-168.
- Scalea JR, Restaino S, Scassero M, Blankenship G, Bartlett ST, Wereley N. An Initial Investigation of Unmanned Aircraft Systems (UAS) and real-time organ status measurement for transporting human organs. *IEEE J Transl Eng Health Med.* 2018;6:4000107. doi:10.1109/ JTEHM.2018.2875704
- Johannessen K-A, Comtet H, Fosse E. A drone logistic model for transporting the complete analytic volume of a large-scale university laboratory. Int J Environ Res Public Health. 2021;18(9):4580. doi:10.3390/ijerph18094580
- Services MoHaC. National health and hospital plan 2020–2023. The Norwegian Government; 2019. Available from: https://www.regjeringen.no/en/ dokumenter/meld-st-7-20192020/id2678667/. Accessed June 29, 2022.
- Ackerman E, Koziol M. The blood is here: zipline's medical delivery drones are changing the game in Rwanda. *IEEE Spectr.* 2019;56(5):24–31. doi:10.1109/MSPEC.2019.8701196
- 13. Mateen FJ, Leung KHB, Vogel AC, Cisse AF, Chan TCY. A drone delivery network for antiepileptic drugs: a framework and modelling case study in a low-income country. *Trans R Soc Trop Med Hyg.* 2020;114(4):308–314. doi:10.1093/trstmh/trz131
- Johannessen KA. A conceptual approach to time savings and cost competitiveness assessments for drone transport of biologic samples with unmanned aerial systems (Drones). Drones. 2022;6(3):62. doi:10.3390/drones6030062

- 15. Nentwich M, Hórvath DM Delivery drones from a technology assessment perspective. Overview report; 2018.
- Ferreras LE. iTRANS: proactive ITS based on drone technology to solve urban transportation challenge. In: Meyer G, Shaheen S, editors. Disrupting Mobility. Cham: Springer International Publishing; 2017:323–333.
- 17. Mion FU Flying drones to exchange lab samples service innovation by the Swiss multisite hospital EOC. 2019.
- Taylor J, Coates E, Wessels B, Mountain G, Hawley MS. Implementing solutions to improve and expand telehealth adoption: participatory action research in four community healthcare settings. BMC Health Serv Res. 2015;15(1):529. doi:10.1186/s12913-015-1195-3
- De Bruyne S, Speeckaert MM, Van Biesen W, Delanghe JR. Recent evolutions of machine learning applications in clinical laboratory medicine. Crit Rev Clin Lab Sci. 2021;58(2):131–152. doi:10.1080/10408363.2020.1828811
- Eichleay M, Evens E, Stankevitz K, Parker C. Using the unmanned aerial vehicle delivery decision tool to consider transporting medical supplies via drone. *Glob Health Sci Pract.* 2019;7(4):500–506. doi:10.9745/GHSP-D-19-00119
- 21. Barbero S, Pallaro A. Systemic design for sustainable healthcare. *Design J.* 2017;20(sup1):S2473–S2485. doi:10.1080/14606925.2017.1352762
- 22. Awad A, Trenfield SJ, Pollard TD, et al. Connected healthcare: improving patient care using digital health technologies. Adv Drug Deliv Rev. 2021;178:113958. doi:10.1016/j.addr.2021.113958
- Bongomin O, Yemane A, Kembabazi B, et al. Industry 4.0 disruption and its neologisms in major industrial sectors: a state of the art. J Eng. 2020;2020:1–45. doi:10.1155/2020/8090521
- Authority S-ENRH. Regional utviklingsplan 2035 [Regional development plan 2035 New ways of working and better use of technology]. South-Eastern Norway Regional Health Authority; 2021.https://helse-sorost.no/regional-utviklingsplan-2035#nye-arbeidsformer-og-bedre-bruk-avteknologi. Accessed March 3, 2022.
- 25. J-P V, Bauknecht D, Kemp R. Reflexive Governance for Sustainable Development. Cheltenham: Edward Elgar; 2006.
- Geels FW. Disruption and low-carbon system transformation: progress and new challenges in socio-technical transitions research and the multi-level perspective. *Energy Res Soc Sci.* 2018;37:224–231. doi:10.1016/j.erss.2017.10.010
- Geels FW, McMeekin A, Mylan J, Southerton D. A critical appraisal of sustainable consumption and production research: the reformist, revolutionary and reconfiguration positions. *Glob Environ Change*. 2015;34:1–12. doi:10.1016/j.gloenvcha.2015.04.013
- Berg M. Patient care information systems and health care work: a sociotechnical approach. Int J Med Inform. 1999;55(2):87–101. doi:10.1016/ S1386-5056(99)00011-8
- Geels FW. Socio-technical transitions to sustainability: a review of criticisms and elaborations of the multi-level perspective. Curr Opin Environ Sustain. 2019;39:187–201. doi:10.1016/j.cosust.2019.06.009
- Loorbach D, Schwanen T, Doody BJ, Arnfalk P, Langeland O, Farstad E. Transition governance for just, sustainable urban mobility: an experimental approach from Rotterdam, the Netherlands. J Urban Mobil. 2021;1:100009. doi:10.1016/j.urbmob.2021.100009
- Whitmarsh L. How useful is the multi-level perspective for transport and sustainability research? J Transp Geogr. 2012;24:483–487. doi:10.1016/j.jtrangeo.2012.01.022
- 32. Geels FW. Technological Transitions and System Innovations: A Co-Evolutionary and Socio-Technical Analysis. Edward Elgar Publishing; 2005.
- Geels FW. Major system change through stepwise reconfiguration: a multi-level analysis of the transformation of American factory production (1850–1930). Technol Soc. 2006;28(4):445–476. doi:10.1016/j.techsoc.2006.09.006
- Leväsluoto J, Määttä H, Hyytinen K. Challenges of customer oriented health care service models: perspectives from the renewal of a primary health care and in; 2014.
- Geels FW. From sectoral systems of innovation to socio-technical systems: insights about dynamics and change from sociology and institutional theory. Res Policy. 2004;33(6):897–920. doi:10.1016/j.respol.2004.01.015
- Geels FW, Kemp R. Dynamics in socio-technical systems: typology of change processes and contrasting case studies. *Technol Soc.* 2007;29 (4):441–455. doi:10.1016/j.techsoc.2007.08.009
- Loorbach D, Frantzeskaki N, Avelino F. Sustainability transitions research: transforming science and practice for societal change. Annu Rev Environ Resour. 2017;42(1):599–626. doi:10.1146/annurev-environ-102014-021340
- Greene JC, Caracelli VJ, Graham WF. Toward a conceptual framework for mixed-method evaluation designs. *Educ Eval Policy Anal.* 1989;11 (3):255–274. doi:10.3102/01623737011003255
- 39. Andrew S, Halcomb EJ. Mixed Methods Research for Nursing and the Health Sciences. Chichester: Wiley-Blackwell; 2009.
- 40. Creswell JW, Plano Clark VL. Designing and Conducting Mixed Methods Research. 3rd ed. Los Angeles: Sage; 2018.
- Gaskell G. dividual and Group Interviewing. In: Bauer MW, Gaskell G, editors. Qualitative Researching with Text, Image and Sound. London: SAGE Publications Ltd; 2000.
- 42. Dey I. Qualitative Data Analysis: A User-Friendly Guide for Social Scientists. London: Routledge; 1993.
- 43. Miles MB, Huberman AM, Saldaña J. Qualitative Data Analysis: A Methods Sourcebook. 4th ed. Los Angeles: SAGE; 2019.
- 44. Ling G, Draghic N. Aerial drones for blood delivery. Transfusion. 2019;59(S2):1608–1611. doi:10.1111/trf.15195
- 45. Hwang J, Lee JS, Kim H. Perceived innovativeness of drone food delivery services and its impacts on attitude and behavioral intentions: the moderating role of gender and age. Int J Hosp Manag. 2019;81:94–103. doi:10.1016/j.ijhm.2019.03.002
- 46. Park J, Kim S, Suh K. A Comparative analysis of the environmental benefits of drone-based delivery services in urban and rural areas. Sustainability. 2018;10(3):5.
- Rejeb A, Rejeb K, Simske SJ, Treiblmaier H. Drones for supply chain management and logistics: a review and research agenda. Int J Logist Res Appl. 2021;1–24. doi:10.1080/13675567.2021.1981273
- Vargo SL, Akaka MA, Wieland H. Rethinking the process of diffusion in innovation: a service-ecosystems and institutional perspective. J Bus Res. 2020;116:526–534. doi:10.1016/j.jbusres.2020.01.038
- Tan LKL, Lim BC, Park G, Low KH, Yeo VCS. Public acceptance of drone applications in a highly urbanized environment. *Technol Soc.* 2021;64:101462. doi:10.1016/j.techsoc.2020.101462
- Johnson AM, Cunningham CJ, Arnold E, Rosamond WD, Zegre-Hemsey JK. Impact of using drones in emergency medicine: what does the future hold? Open Access Emerg Med. 2021;13:487–498. doi:10.2147/OAEM.S247020
- Komasová S. Possible inspiration: drone-related literature and its potential for public perception research. J Intell Robot Syst. 2021;103(3). doi:10.1007/s10846-021-01498-9

- Smith A, Dickinson JE, Marsden G, Cherrett T, Oakey A, Grote M. Public acceptance of the use of drones for logistics: the state of play and moving towards more informed debate. *Technol Soc.* 2022;68:101883. doi:10.1016/j.techsoc.2022.101883
- Comtet HE, Johannessen KA. The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems. Int J Environ Res Public Health. 2021;18(5):2637. doi:10.3390/ijerph18052637
- 54. Comtet HE, Johannessen KA. A socio-analytical approach to the integration of drones into health care systems. *Information*. 2022;13(2):62. doi:10.3390/info13020062
- 55. Geels FW, Schot J. Typology of sociotechnical transition pathways. Res Policy. 2007;36(3):399-417. doi:10.1016/j.respol.2007.01.003

Journal of Multidisciplinary Healthcare

Dovepress

Publish your work in this journal

The Journal of Multidisciplinary Healthcare is an international, peer-reviewed open-access journal that aims to represent and publish research in healthcare areas delivered by practitioners of different disciplines. This includes studies and reviews conducted by multidisciplinary teams as well as research which evaluates the results or conduct of such teams or healthcare processes in general. The journal covers a very wide range of areas and welcomes submissions from practitioners at all levels, from all over the world. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/journal-of-inflammation-research-journal

f 🔰 in 🕨 DovePress 1885

Appendix A

Statements of co-authorship

Statements of co-authorship from:

- 1. Karl-Arne Johannessen
- 2. Martina Keitsch

To whom it may concern

Statement of co-authorship on joint publications to be used in the PhD-thesis of Hans E. Comtet

As co-author on the following joint publications in the PhD-thesis "Drones in Healthcare Systems: Insights from a Multi-Level Perspective" by Hans E. Comtet

Declaration of co-authorship on the following articles:

(1) H. E. Comtet and K. A. Johannessen, "*A Socio-Analytical Approach to the Integration of Drones into Health Care Systems*," Inf., vol. 13, no. 2, p. 62, 2022, doi: 10.3390/info13020062.

The paper was written by Comtet and Johannessen. Comtet created the first draft. The authors jointly created a first coding scheme and followed an inductive process to develop categories.

(2) H. E. Comtet and K.-A. Johannessen, "*The moderating role of pro-innovative leadership and gender as an enabler for future drone transports in healthcare systems,*" Int. J. Environ. Res.Public Health, vol. 18, no. 5, pp. 1–15, 2021, doi: 10.3390/ijerph18052637.

The paper was written by both Comtet and Johannessen. Comtet created the first draft, both authors created the questionnaire, Comtet distributed it. Both authors analyzed the data, Johannessen performed the statistical analyses.

(3) Comtet HE, Keitsch M, Johannessen KA. "*Realities of Using Drones to Transport Laboratory Samples: Insights from Attended Routes in a Mixed-Methods Study,*" J Multidiscip Healthc. 2022;15:1871-1885, https://doi.org/10.2147/JMDH.S371957

Comtet wrote a first draft of the paper. Comtet and Johannessen planned the data collection. Comtet collected the data. Johannessen conducted the data analysis and contributed to the writing process. Keitsch provided feedback and comments throughout the writing process.

I declare that his contribution to the papers is correctly identified, and I agree that this work is to be used as part of the thesis.

Oslo 17.12.22 Place, date

Marlan Johanno

Karl-Arne Johannessen

To whom it may concern

Statement of co-authorship on joint publications to be used in the PhD-thesis of Hans E. Comtet

As co-author on the following joint publications in the PhD-thesis "Drones in Healthcare Systems: Insights from a Multi-Level Perspective" by Hans E. Comtet

Declaration of co-authorship on the following articles:

(1) Comtet HE, Keitsch M, Johannessen KA. "Realities of Using Drones to Transport Laboratory Samples: Insights from Attended Routes in a Mixed-Methods Study," J Multidiscip Healthc. 2022;15:1871-1885, https://doi.org/10.2147/JMDH.S371957

Comtet wrote a first draft of the paper. Johannessen and Comtet planned the data collection. Comtet collected the data. Johannessen contributed to the writing process. Keitsch provided feedback and comments throughout the writing process.

I declare that his contribution to the papers is correctly identified, and I agree that this work is to be used as part of the thesis.

Trondheim 20.12.2022

Place, date

Martina Keitsch



ISBN 978-82-326-7400-8 (printed ver.) ISBN 978-82-326-7399-5 (electronic ver.) ISSN 1503-8181 (printed ver.) ISSN 2703-8084 (online ver.)

